Regulation of Food Intake in Adults with and without Obesity: The Role of the Gastrointestinal Tract and Gut-Brain Axis

Sarah Althubeati, BSc, MSc

Thesis submitted to the University of Nottingham for the

degree of Doctor of Philosophy

November 2023

Division of Food, Nutrition & Dietetics School of Biosciences University of Nottingham

Abstract

Obesity is a complex global health issue affecting a significant portion of the population. In the UK, it is estimated that approximately 1 in 4 adults and 1 in 5 children aged 10 to 11 years are living with obesity. Impairment in food intake regulation, including hunger and satiety sensations, are key factors contributing to overeating and weight gain, particularly in individuals with obesity. While various mechanisms may explain these alterations, such as altered appetite and satiety regulators, accelerated gastric emptying (GE), and heightened brain responses to food cues and reward, it remains inconclusive whether these mechanisms are altered in people living with obesity compared to normal-weight (NW) adults.

There are various approaches to studying food regulation, each offering unique insights into the complex mechanisms that control appetite, satiety, and food intake. Non-invasive imaging, particularly magnetic resonance imaging (MRI), provides a powerful tool for investigating the physiological mechanisms underlaying the regulation of food intake. The work in this PhD thesis aims to combine physiological measurements obtained by using MRI with behavioural assessments (i.e., subjective satiety rating), to provide a more comprehensive understanding of appetite control in NW adults and alterations associated with obesity. The work in this thesis included a functional neuroimaging meta-analysis, and three eating behaviour intervention studies, two of which used MRI techniques.

A functional neuroimaging meta-analysis was performed to identify brain areas associated with changes in appetite and satiety regulators in NW and Obese adults. The caudate nucleus and hypothalamus were identified as key areas associated with satiety regulators in NW participants. However, conclusive findings for Obese participants were limited due to the small number of studies conducted in this area.

An MRI study was conducted to investigate the effect of a standard meal on gastrointestinal (GI) responses. The study found that GI responses including gastric content volume (GCV), GE rate, small bowel water content (SBWC), and superior mesenteric artery (SMA) blood flow, and appetite and satiety

regulators were not significantly altered by obesity following the meal. However, Obese participants showed lower satiety subjective rating, and higher insulin and triglyceride levels compared to NW participants.

Different macronutrients play distinct roles in influencing feelings of fullness and satiety, and their impact on the satiety sensation can be a valuable strategy for weight loss. In a pilot MRI study combining gut and brain imaging, responses to a high-fat (HF) emulsion drink and a carbohydrate drink that is matched in caloric content, volume, and viscosity were assessed in NW and Obese participants. Data collection in this study was significantly impacted by the COVID pandemic; therefore, findings from this work are focused on GI responses. The results suggest that the HF drink might induce higher GCV, SBWC, SMA blood flow, and subjective satiety ratings when compared to an iso-caloric, and iso-viscous high carbohydrate drink (HC) in both NW and Obese.

The final study investigated the satiating effect of acute high protein consumption compared to high carbohydrate in NW and Obese participants using ad libitum meal intake and subjective satiety ratings. This study found no significant differences in ad libitum energy intake, subjective satiety, or energy intake between the drinks in either NW or Obese participants.

This research integrated different approaches to measuring the regulation of food intake and alterations in obesity. This holistic approach facilitates a comprehension understanding of the mechanisms governing food regulation, including the impact of macronutrient composition, hormonal influences, gastrointestinal responses, neural signalling, and eating behaviours. While the studies in the thesis did not reveal significant differences in certain aspects of appetite regulation between NW and Obese, including macronutrient compositions, they did highlight several areas requiring further investigations. The complicated nature of obesity and appetite regulation necessitates continued research to better understand these complex mechanisms and inform strategies for obesity management and prevention.

Acknowledgment

I would like to take this opportunity to express my thanks to everyone who helped me produce this PhD thesis.

First, I would like to express my heartfelt gratitude to my supervisors, Prof. Andrew Salter, Dr. Sally Eldeghaidy, Prof. Dileep Lobo, and Dr. Amanda Avery, for their unwavering support, guidance, and mentorship throughout my doctoral journey. I would like to express my deepest gratitude to Prof. Andrew for his encouragement and guidance, which played a pivotal role in shaping the course of my research. I would like to extend my heartfelt appreciation to Dr. Sally Eldeghaidy for her steadfast support and invaluable guidance throughout the journey of my PhD. Working with Dr. Sally Eldeghaidy under her mentorship has been an enlightening experience, where I not only gained a wealth of knowledge but also developed essential skills that will undoubtedly shape my academic and professional future. My sincere gratitude also goes to Dr. Amanda Avery for her invaluable support in designing the intervention meals and her expert guidance in the data collection process. Also, I would like to extend my heartfelt appreciation to Prof. Dileep Lobo for his invaluable guidance in the data collection process and his unwavering encouragement to present my work at conferences.

I would like to extend my sincere thanks to Dr. Caroline Hoad for her invaluable training and unwavering support in analysing gut MRI data for my PhD thesis. She was always by my side, ready to provide assistance and guidance whenever I needed it in the field of MRI analysis. Her expertise and mentorship were instrumental in my research journey and significantly enriched the quality of my work. Special gratitude also goes to Dr. Liz Simpson for her support and wealth of knowledge provided during the first intervention study in this thesis. Working alongside Dr. Liz was a valuable learning experience and a source of skills and development that was helpful for the PhD journey. Also, I would like to express my sincere appreciation to Sarah Wolf for her invaluable assistance with blood samples during the screening visits in my research studies, which played a critical role in the advancement of my studies. I would like to extend my thanks to Dr. Randa Darwish for her invaluable assistance and guidance in conducting

the in vitro digestion experiment for my research. My sincere gratitude goes to Dr. Vincenzo Di Bari for his invaluable assistance and support in conducting viscosity measurements for my research. My special thanks also go to Sally Cordon for the analysis of blood samples for this thesis. Also, I would like to thank Prof. Andrew Cooper and Jane Paul for doing MRI scanning for this thesis.

Finally, I am grateful to my family for their unconditional support throughout my PhD journey. A special heartfelt thanks to my parents, Marzoug and Meseadah, for their love, encouragement, and belief in me during my learning journey in life. My heartfelt gratitude to my husband, Samer, for his love and support during my ups and downs throughout my learning experience. I am grateful to have a supportive partner like him in my life. Special thanks to my sisters, Amal, Arwa, and Afnan, and my brothers, Yasser, Abdullah, and Abdulaziz, for their support and staying beside me during this journey. I am truly blessed to have sisters and brothers always by my side. Lastly, I want to express my gratitude to my beautiful daughter, Tala, for providing me with joy and happiness throughout this journey, which has given me the strength to keep going.

Publications and Presentation arising from the thesis

Publication

Althubeati, S., Avery, A., Tench, C.R., Lobo, D.N., Salter, A. and Eldeghaidy, S., 2022. Mapping brain activity of gut-brain signaling to appetite and satiety in healthy adults: A systematic review and functional neuroimaging meta-analysis. Neuroscience & Biobehavioral Reviews, 136, p.104603.

Oral presentations

- The British Feeding and Drinking Group (BFDG) 46th Annual Meeting: Althubeati, S., Avery, A., Tench, C.R., Lobo, D.N., Salter, A. and Eldeghaidy, S., 2022. Functional neuroimaging meta-analysis to map brain activity of gut-brain signalling to appetite and satiety in healthy adults. Appetite, 179, p.106213.
- The 2023 Annual Meeting of the Surgical Research Society (SRS): Althubeati, S., Simpson, E.J., Bush, D., Hoad, C., Elgdeghaidy, S., Gowland, P., Macdonald, I.A. and Lobo, D.N., 2023. O032 Gastrointestinal and satiety responses to oral feeding in participants with obesity compared with those of healthy weight: an MRI study. British Journal of Surgery, 110(Supplement_3), pp.znad101-032.

Poster of distinction presentation

• The 43rd ESPEN Congress: Althubeati, S., Avery, A., Lobo, D.N., Salter, A. and Eldeghaidy, S., 2021. A neuroimaging meta-analysis to identify brain areas associated with satiety and appetite regulators. Clinical Nutrition ESPEN, 46, p.S582.

Candidate Statement

All work of this PhD thesis was conducted by the candidate unless otherwise stated.

Abbreviations

In line with current best practice, first person language 'people (adults) living with obesity' has been used in the text. However, in the studies, to assist ease of comparison with other related studies, Obese participants is the language that has been used.

ABC-Analysis of Brain Coordinates

ACC- Anterior cingulate cortex

AgRP- Agouti-related protein

ALE- Activation likelihood estimation

ANOVA- Analysis of variance

AP- Area postrema

ARC- Arcuate nucleus

ASL- Arterial spin labelling

AUC- Area under the curve

BES- Binge eating scale

BMI- Body mass index

BOLD- Blood Oxygenation Level Dependent

CART- Cocaine- and amphetamine-regulated transcript

CBF- Cerebral blood flow

CSF- Cerebrospinal fluid

CCK- Cholecystokinin

CNS- Central nervous system

CoEQ- Control of Eating Questionnaire

CSS- Composite satiety score

DE- Dextrose equivalent

DEBQ- Dutch Eating Behaviour Questionnaire

- DTE- Desire to eat
- EAT- Eating Attitude Test
- FDR- False Discovery Rate
- FFAs- Free fatty acids
- FID- Free induction decay
- fMRI- Functional magnetic resonance imaging
- FWE- Family-wise error
- FWHM- Full width at half maximum
- GCV- Gastric content volume
- GE- Gastric emptying
- GHS- Growth hormone secretagogue
- **GI-** Gastrointestinal
- GLP-1- Glucagon-like peptide-1
- GLP-2- Glucagon-like peptide-2
- HASTE- Half-Fourier Acquisition Single-shot Turbo spin Echo
- HC- High carbohydrate
- HF- High fat
- HP- High protein
- HPMC- Hydroxypropyl methylcellulose
- IQR- Interquartile range
- LBG- Locust bean gum
- LH- lateral hypothalamus
- MIP- Maximum intensity projections

MNI- Montreal neurological institute

MRI- Magnetic resonance imaging

NPY- Neuropeptide Y

NTS- Nucleus of the solitary tract

NW- Normal weight

OFC- Orbitofrontal cortex

OXM- Oxyntomodulin

PET- Position emission tomography

PFS- Prospective food intake

POMC- Pro-opiomelanocortin

PPU- Peripheral pulse-oximeter unit

PrefQuest- Food preference questionnaire

PVN- Paraventricular nucleus

PYY- Peptide tyrosine tyrosine

RF- Radiofrequency

ROI- Region of interest

rs-fMRI- Resting state functional magnetic resonance imaging

SBWC- Small bowel water content

SMA- superior mesenteric artery

SPMIC - Sir Peter Mansfield Magnetic Resonance Imaging Centre

SQ- Satiety quotient

T₅₀- Half emptying time of gastric content

TE- Echo time

TFEQ- Three Factor Eating Questionnaire

- TR- Repetition time
- VAS- Visual analogue scale
- VOI- Volume of interest
- VTA- Ventral tegmental area
- α -MSH- Alpha-melanocyte stimulating hormone

Table of content

1	I Introduction				
	1.1 T	hesis Overview	5		
2	Regul	ation of food intake	7		
	2.1 Regulation of food intake by the CNS				
	2.1.1	Homeostatic system	8		
	2.1.2	Hedonic system	10		
	2.2 R	egulation of food intake by the gastrointestinal tract	11		
	2.2.1	Food digestion and gastric emptying	11		
	2.2.1	.1 The process of gastric emptying	13		
	2.2.1	.2 Impact of gastric emptying rate on food intake regulation	14		
	2.3 A	ppetite and satiety regulators	17		
	2.3.1	Gut hormones	17		
	2.3.2	Other appetite and satiety regulators	23		
	2.4 N	utrient absorption	24		
	2.5 E	ffect of macronutrient compositions on food intake regulation	27		
	2.6 A	pproaches to measure regulation of food intake.			
	2.6.1	Appetite Rating Scales	30		
	2.6.2	Satiety quotient	31		
	2.6.3	Ad Libitum meal intake	32		
	2.6.4	Gastrointestinal responses to food intake	32		
	2.6.4	.1 Gut hormones	33		
	2.6.4	.2 Gastric emptying	33		
	2.6.5	Nutrient absorption	32		
	2.6.5	.1 Mesenteric blood flow			
	2.6.6	Neuronal signalling to food intake	36		
	2.6.7	Questionnaires and surveys to assess eating behaviour data	37		
	2.6.7	.1 Binge eating scale (BES)	37		
	2.6.7	.2 Control of eating questionnaire (CoEQ)			
	2.6.7	.3 Depression scale	38		
	2.6.7	.4 Dutch eating behaviour questionnaire (DEBQ)			
	2.6.7	.5 Eating attitude test (EAT)			

	2.6	5.7.6	Food preference questionnaire (PrefQuest)	
	2.6	5.7.7	Three-factor eating questionnaire (TFEQ)	40
2.6.7.8			Power of food scale (PFS)	40
	2.6	5.7.9	Intake24 online dietary recall survey	40
3	MR	I Theo	ory and Image Processing	42
	3.1	Basic	principles of MRI	42
	1.1	Image	formation	45
	1.2	T1-we	eighted and T2-weighted contrasts	46
4	Map	ping	brain activity of gut-brain signalling to appetite and s	satiety
in	n people	with	and without obesity: A systematic review and functio	nal
n	euroima	aging	meta-analysis	49
	4.1	Introd	uction	50
	4.2	Aims	and Hypothesis	51
	4.3 Material and m		ial and methods	51
	4.3.1 Systematic r		stematic review of the literature	51
	4.3.2	Sea	rch strategy	52
	4.3.3 Rist 4.3.4 Dat		k-of-Bias Assessment	54
			ta extraction	55
	4.3.5	Coo	ordinate-based neuroimaging meta-analysis	55
	4.3.5.1 4.3.5.2		Activation likelihood estimation meta-analysis	
			Analysis of Brain Coordinates	
	4.4	Result	ts	58
	4.4.1	Sys	stematic review	58
	4.4	4.1.1	Selection and inclusion of studies	
	4.4	4.1.2	Characteristics of included studies	
	4.4.2	Mo	dulation of brain responses to appetite and satiety regulators .	82
	4.4.2.1 4.4.2.2 1.2.1 C 4.4.2.3		Appetite regulators	
			Satiety regulators	
			ordinate-based neuroimaging meta-analysis	87
			Concurrence of brain area modulated by appetite regulators	
	4.4	4.2.4	Concurrence of brain area modulated by satiety regulators	
	4.5	Discu	ssion	91
	4.5.1	Co	ncurrence of brain area modulated by appetite regulators	91

4.5	.2 Concurrence of brain area modulated by satiety regulators				
4.6	Strengths and limitations94				
4.7	Conclusion				
5 An	MRI study to measure the effect of standard pasta meal on				
gastroi	ntestinal responses in adults with and without obesity98				
5.1	Introduction				
5.2	Aims and hypothesis				
5.3	Methods100				
5.3	.1 Study design and ethics				
5.3	.2 Eligibility criteria				
5.3	.3 Sample size calculation				
5.3	.4 Study protocol				
5.3	.5 Test meal102				
5.3	.6 Blood samples and metabolic assays103				
5.3	.7 Gut imaging protocol and analysis104				
5.3	.8 Statistical analysis				
5.4	Results107				
5.4	.1 Participant's descriptive results				
5.4	2 Gastric content volume				
5.4	.3 Superior mesenteric artery blood flow				
5.4	.4 Small bowel water content				
5.4	5 Subjective Satiety				
5.4	.6 Gut hormones				
5.4	7 Other blood results				
5.5	Discussion				
5.6	Strength and limitations of the study				
5.7	Conclusion				
6 An	6 An MRI pilot study to explore gastrointestinal responses to a high fat				
emulsion drink in people with and without obesity					
6.1	Introduction				

	6.2	А	ims and hypothesis	139
	6.3	Ν	lethods	139
	6.3	3.1	Development work on designing the study drinks	139
	6.3	3.2	Viscosity measurement	141
	6.3	3.3	In vitro digestion assessment for the fat emulsion stability	y in the gastric
	ph	ase	144	C
	64	Ν	ain "MRI" study data collection	152
	6.4	4.1	Study design and ethics	
	64	12	Fligibility criteria	153
	6.	1.2	Sample size calculation	154
	0.4	+.3	Sample size calculation	134
	6.4	1.4	Study protocol	154
	6.4	4.5	Gut imaging protocol	156
	6.4	4.6	Data analysis	157
	6.4	4.7	Statistical analysis	157
	6.5	R	esults	
		6.5.1	.1 Participant's demographic characteristics	
		6.5.1	.2 Gastric content volume	
		1.1.1	.1 Superior mesenteric artery blood flow	
		6.5.1	.3 Small bowel water content	
		6.5.1	.4 Subjective satiety	
	6.5	5.2	Eating behaviour questionnaires	179
		6.5.2	.1 Three-eating factor questionnaire	179
		6.5.2	.2 Control of eating questionnaire	
		6.5.2	.3 Binge eating scale	
		6.5.2	.4 Power of food scale	
		1.1.1	.1 Food preference questionnaire	
	6.6	D	iscussion	
	6.7	С	onclusion	
7	As	ssess	ing appetite and satiety responses to a high protein	a drink in
a	dults	with	obesity and without obesity	
	7.1	Ir	troduction	
	7.2	А	ims and hypothesis	
			JI	

	7.3	Metho	ds	190
	7.3.	1 Dev	velopment work on designing the study drinks	190
	7.4	Interve	ention Study	194
	7.4.	1 Met	thods	194
	7	.4.1.1	Study design and ethics	194
	7	.4.1.2	Recruitment	194
	7	.4.1.3	Eligibility criteria	195
	7	.4.1.4	Sample size calculation	
	7	.4.1.5	Study protocol	196
	7	.4.1.6	Test drinks	197
	7	.4.1.7	Ad Libitum lunch meal	198
	7	.4.1.8	Subjective satiety	199
	7	.4.1.9	Satiety quotient	199
	7	.4.1.10	Eating behaviour questionnaires	199
	7	.4.1.11	Statistical analysis	199
	7.5	Result	s	
	7.5.	1 Part	ticipant's descriptive results	
	7.5.	2 Ad	libitum meal intake	
	7.5.	3 Sub	jective Satiety	
	7.5.	4 Sati	ety quotient	208
	7.5.	5 Ene	ergy intake for the rest of the study day	210
	7.5.	6 Thr	ee days average intake of energy and macronutrients	211
	7.5.	7 Eati	ing behaviour scores	212
	7.6	Discus	ssion	214
	7.7	Streng	th and limitations of the study	217
	7.8	Conclu	usion	218
:	8 Co	nclusio	ns	210
	8.1	Advan	tages and limitations	212
	8.2	Recom	nmendations for future work	
	9 Ref	ference	S	
	10 Ap	pendice	es	
	10 1	Visual	analogue scale (VAS)	267
	10.1	, isual		

10.2	2 Prep	aration methods of drinks and meals used in the thesis	267
10.2.1		Materials and equipment used in preparation of drinks and meals	in
tł	ne thesis	267	
1	0.2.2	Area and equipment cleaning	268
1	0.2.3	Preparation method of the standard pasta meal in Chapter 5	269
1	0.2.4	Preparation of high-fat emulsion (High-fat drink) in Chapter 6	269
1	0.2.5	Preparation of high-carbohydrate drink in Chapter 6	271
1	0.2.6	Preparation method of high-protein drink in Chapter 7	272
1	0.2.7	Preparation method of high-carbohydrate drink in Chapter 7	272
1	0.2.8	Preparation of ad libitum pasta meal in Chapter 7	272

1 Introduction

Obesity is a global health epidemic that has risen dramatically in recent decades. According to the World Health Organisation (WHO), the global prevalence of obesity nearly tripled from 1975 to 2016, with around 650 million are affected by obesity in 2016 (WHO, 2021). In the UK, the prevalence of obesity is also increasing. The 2021 Health Survey for England estimated that 25.9% of adults in England are obese, and a further 37.9% are overweight but not obese (House of Commons Library, 2023). Obesity is characterised by an excess accumulation of body fat, and it is commonly classified using body mass index (BMI), which is a measure of an individual's body fat based on their height and weight. The WHO and many health organisations use BMI to categorise individuals into different classes of obesity. Adults with BMI \geq 30 kg/m² can be classified as living with obesity, whereas $25 \text{ kg/m}^2 \le \text{BMI} \ge 30 \text{ kg/m}^2$ is overweight, and BMI 18.5 to 24.9kg/m² is considered normal-weight (NW) (WHO, 2021). It is also important to note, there are also other methods to assess obesity including waist circumference. Generally, a waist circumference of 35 inches (88 cm) or more for women and 40 inches (102 cm) or more for men is considered indicative of abdominal obesity. Also, body composition analysis can be used to assess obesity. This includes using dual-energy X-ray absorptiometry (DXA), bioelectrical impedance analysis (BIA), and skinfold thickness measurements to assess the distribution of body fat and lean mass.

Obesity is associated with several adverse health outcomes, including an increased risk of chronic diseases such as diabetes, cardiovascular disease, and certain cancers (WHO, 2021). Causes of obesity are multifactorial, and it often results from the interplay of various genetic, environmental, behavioural, and metabolic factors. Genetic factors can play a significant role in an individual's susceptibility to obesity. Certain genes may influence metabolism, appetite regulation and the way the body stores, and utilises fat (Krude and Grüters, 2000, Locke et al., 2015). Calorie -dense diets high, such as high-fat foods and sugary beverages can also contribute to weight gain. Hormonal changes, such as insulin

resistance and leptin resistance, can disrupt appetite regulation and contribute to obesity (Considine et al., 1996, Ginsberg, 2000, Zyoud et al., 2022).

While obesity is caused by multiple factors, one of the critical elements that contributes to its development and persistence is alterations in the regulation of food intake. Hunger, satiation and satiety are cycle stages of food intake (Blundell and Halford, 1994). Appetite is defined as the desire to consume food (Heisler and Lam, 2017). Satiation is the process that results in meal termination and the disappearance of hunger (Blundell and Halford, 1994), while satiety refers to the time period between the termination of one meal and the beginning of the next (Strubbe and Woods, 2004, Cummings and Overduin, 2007). The sensation of fullness and appetite involves a complex physiological and psychological process, influenced by an interplay between various factors, including gastrointestinal (GI) responses, neural pathways, environmental cues, and individual behaviours (Dalton et al., 2013). A summary of the factors that affect the onset, timing, and length of satiety feelings are summarised in the satiety cascades which was developed by Blundell et al. (1987) and consequently updated and amended by Blundell et al. (2010) and Tremblay and Bellisle (2015). It is a theoretical framework that explains factors influencing satiety including the physiological processes, the psychological experiences, and instantaneous/immediate behaviours necessary for the eating process (figure 1.1) (Blundell et al., 2010). Particularly in relation to meals, hunger is a wellknown early 'signal' or state that initiates the eating process, whereas the release of satiety signals after food intake contributes to ending the eating event. The hunger signals that originate from the stomach send electrical signals via the vagus nerve related to feelings of emptiness or fullness, reinforced by metabolic states such as blood glucose levels (hypoglycemia) and hormone secretions such as ghrelin. Sensory and cognitive processes lead to expectations about meals with expected pleasure and reward, which help define overall meal quality and amount. The stomach and intestine distention and osmotic load provide postingestive information that gives feedback on the amount of food consumed. Gut peptide hormones such as PYY, GLP-1 and CCK released after food intake modulate medium-term satiety metabolically by inhibiting food intake (Van Kleef et al., 2012). Long-term satiety is controlled by nutrient oxidation in the

liver during the post-absorptive phase as well as the blood concentrations of amino acid, insulin, and glucose. More details about the long-term regulation are explained in section 2.1.1. The signals relating to sensory and metabolic satiety, as well as those relating to hedonic and homeostatic appetite control, are all integrated by the brain (Figure 1.1).



Figure 1.1. Satiety cascade. Figure reproduced from (Blundell et al., 2010).

Previous studies have shown that some people with obesity report a dysfunction in the link between their feelings of hunger and fullness and their eating habits (Barkeling et al., 2007, Sumithran et al., 2011). This may be related to the impaired regulation of food intake mechanisms in people with obesity. Some studies have shown lower postprandial responses of circulating satiety regulators/gut hormones, and higher concentrations of appetite regulators, in people with obesity after a standardised drink, compared with NW participants (Meyer-Gerspach et al., 2014). However, others have found no differences in the appetite and satiety regulators between NW and people with obesity after a standard liquid or breakfast meal intakes (Carroll et al., 2007, Yang et al., 2009). Regulators can be defined as factors that regulate hunger, satiation, and fullness feelings. Satiety regulators could be referred to gut peptide hormones such as PYY, GLP-1 CCK, insulin, and leptin, which released after food intake, and blood glucose levels (hyperglycaemia) that modulate satiety metabolically by inhibiting food intake. In contrast, appetite regulators refer to hormone secretions of ghrelin which rise before meal ingestion and decrease afterwards to stimulate hunger/appetite and food intake. Additionally, the metabolic state of blood glucose levels (hypoglycaemia) is related to appetite regulators as a drop in blood glucose may stimulate appetite and food intake (Wyatt et al., 2021). More details about appetite and satiety regulators are discussed in section 2.3.

With regards to the function of the GI tract, previous research has reported that the rate of gastric emptying (GE) is more accelerated in people with obesity compared with NW adults (Mora et al., 2005), leading to increased hunger sensation. However, these results conflict with the findings of Meyer-Gerspach et al. (2014) who showed that GE rate was delayed in people with obesity after eating both solid and liquid meals, compared with the NW group. Other studies have shown no differences in GE between people with obesity and NW participants after eating a standardised semi-solid meal (Pironi et al., 1993, Flint et al., 2007).

Neuronal signalling in response to food intake is a critical aspect of understanding how the brain regulates eating behavior and how alterations in this signalling can contribute to the development of obesity. Findings from neuroimaging studies showed that obesity is associated with increased activation in reward-related brain regions in response to high-calorie and palatable foods (Makaronidis and Batterham, 2018). This heightened reward response can lead to overeating. However, a systematic review by Morys et al. (2020) revealed that there is little evidence to support differences between people with obesity and NW adults in brain responses of inhibitory control and reward areas to food picture cues, and such differences might be mediated by factors that are often not considered, such as food cravings, food restraint, self-control, or age.

The conflicting nature of the evidence described above indicate the need for further study to understand changes in satiety and appetite processing in people with obesity. Therefore, the work in this PhD thesis aims to better understand the appetite and satiety mechanisms in normal-weight adults and alterations in people living with obesity. This will be extended to investigate whether there are differential responses to specific macronutrients between these groups.

1.1 Thesis Overview

All the work described in this thesis was conducted by the author, except where specifically credited to collaborators, at the Sir Peter Mansfield Imaging Centre (SPMIC), and the School of Biosciences, University of Nottingham from 2019 to 2023. The research described in this thesis is original, unless otherwise stated. The layout and content of the thesis chapters are as follows:

Chapter 2 introduces the physiological mechanisms underlying the regulation of food intake. The bidirectional interactions between the brain and the gut and the role of the GI tract in controlling food consumption are outlined. Key methods and approaches that are used in studying the regulation of food intake are highlighted. This includes appetite/satiety rating scales, ad libitum meal intake, GI responses, neuronal signalling to food intake, as well as eating behavioural questionnaires.

Chapter 3 gives a brief description of the principles of MRI and image formation. Additionally, this chapter covers briefly different MRI contrasts.

Chapter 4 employs functional neuroimaging meta-analysis to identify brain areas associated with appetite and satiety regulations in both normal-weight individuals and those with obesity.

Chapter 5 presents an MRI study to explore the gastrointestinal (GI) tract responses to a standard pasta meal in NW and alterations with obesity. The following parameters were measured: gastric content volume (GCV), GE, small bowel water content (SBWC), superior mesenteric artery (SMA) blood flow, subjective satiety, as well as plasma concentrations of appetite and satiety regulators/hormones in normal-weight participants and those living with obesity.

Chapter 6 demonstrates an MRI study to assess the interactions between the gut and brain after consuming a high fat emulsion and an isoenergic, isovolumic,

and isoviscous carbohydrate drink in normal-weight participants and those living with obesity. The chapter describes developing work conducted to optimise the fat emulsion. This is followed by presenting the results from GI responses (GCV, GE, SBWC, and SMA blood flow), and subjective satiety rating after consuming the drinks.

Chapter 7 describes an eating behaviour study in NW and individuals with obesity to measure the satiating effect of protein, which has been suggested to be the most satiating macronutrient. The study explores the impact of acute consumption of high-protein drink on satiety responses assessed by ad libitum meal intake and subjective satiety compared to high-carbohydrate drink (of similar energy content, volume, and viscosity).

Chapter 8 summarises the main findings of the thesis, outlines the advantages and limitations of the thesis, and provides recommendations for future research.

2 Regulation of food intake

The regulation of food intake is a complex process that involves a coordination between the central nervous system (CNS) and the GI tract. The bidirectional interactions between the brain and the gut are vital for ensuring that the body receives the necessary nutrients and maintains energy balance. It influences the sensation of hunger and satiety, ultimately controlling food consumption. The first section of this chapter gives an overview on the physiological mechanisms underlying the regulation of food intake. This includes how the brain and gut communicate to regulate food intake (gut-brain axis). The role of GI tract to regulate food intake is highlighted. The second section outlines key methods and approaches used in studying the regulation of food intake. This includes appetite/satiety rating scales, ad libitum meal intake, GI responses, and neuronal signalling to food intake, as well as eating behavioural questionnaires.

2.1 Regulation of food intake by the CNS

The CNS plays a central role in regulating food intake and appetite. It integrates information from various sources, including peripheral signals from the GI system and adipose tissue as well as sensory inputs, to coordinate a complex network of neural pathways that impact feeding behaviour. The coordination of food intake involves a dynamic interaction between two distinct but interconnected systems: the homeostatic system and the hedonic system. The homeostatic system plays a crucial role in maintaining energy balance and ensuring that the body's physiological needs for nutrients and energy are met (Lutter and Nestler, 2009). The hedonic system, often referred to as the reward system, plays a significant role in influencing food intake and eating behaviours (Batterham et al., 2007, Lutter and Nestler, 2009). Understanding the interplay between the hedonic and homeostatic pathways in food intake is crucial for addressing issues related to overeating, and obesity. This section gives an overview of the homeostatic and hedonic systems. Figure 2.1 illustrates brain areas associated with homeostatic and hedonic pathways for regulation of food intake.



Figure 2.1. Brain areas associated with haemostatic and hedonic pathways for food intake regulation. Figure reproduced from Francis and Eldeghaidy (2015).

2.1.1 Homeostatic system

The homeostatic system involves bidirectional interaction between the GI system and the brain (gut-brain axis). This bidirectional connection allows the gut to transmit important information to the brain regarding the presence of nutrients, the degree of fullness, and other signals related to digestion. In turn, the brain conveys this information to make decisions about hunger, satiety, and meal initiation or termination. When the gut senses a lack of nutrients or a drop in energy levels, it releases a gut peptide called ghrelin (Nakazato et al., 2001) and sends signals to the brain, which may trigger feelings of hunger and the motivation to eat. Conversely, when the gut senses that sufficient nutrients have been consumed, it signals to the brain that it's time to stop eating, leading to feelings of fullness and satisfaction. This is controlled through the enteroendocrine cells within the GI tract which stimulate the release satiety hormones such as glucagon-like peptide-1 (GLP-1), peptide tyrosine tyrosine (PYY), and cholecystokinin (CCK) (Adrian et al., 1985b, Le Quellec et al., 1992). The above signals for food intake typically convey a short-term regulation of food intake.

For the long-term regulation, insulin and leptin are two primary regulators/hormones for the long-term process to control food intake, which are produced from the pancreas and adipose tissue, respectively. They act as adiposity signals that provide negative feedback signals for energy balance to reduce food intake (Benoit et al., 2004, Niswender et al., 2004). Additionally, they positively correlate with body fat mass, and send signals to the brain about the status of energy stores. More information about the short- and long-term hormones are provided in the section 2.3.1 and 2.3.2.

Both short-term and long-term peripheral signals travel from the GI tract, pancreas, or adipose tissues to the CNS via vagal afferent nerve signalling, or directly via blood circulation (Cummings and Overduin, 2007, Ahima and Antwi, 2008). The arcuate nucleus (ARC) in the hypothalamus sense and respond to the peripheral signals by releasing modulated neuropeptide in two neuronal populations with different effects on food intake (Schwartz et al., 2000, Chaudhri et al., 2008). The ARC responds to the peripheral signals (such as leptin and ghrelin) by releasing modulated neuropeptide in two neuronal populations with different effects on food intake (Schwartz et al., 2000). Neurons in the medial part of the ARC (orexigenic neurons) express the Agoutirelated protein (AgRP) and neuropeptide Y (NPY) in response to the ghrelin hormone, which increase appetite, hunger, and food intake (Broberger et al., 1998, Hahn et al., 1998, Bewick et al., 2005). In respond to the short- and longterm satiety hormones, neurons located in the lateral ARC are anorexic, and they express cocaine- and amphetamine-regulated transcript (CART) and alphamelanocyte stimulating hormone (α -MSH) driven by pro-opiomelanocortin (POMC), which suppress appetite, and hunger and food intake (Elias et al., 1998, Batterham et al., 2002, Jobst et al., 2004). The balance between these two types of neurons 'appetite suppressants' and 'appetite stimulators' is crucial in controlling food intake. Neurons in the ACR have additional connections to the hypothalamic paraventricular nucleus (PVN) (Schwartz et al., 2000, Cone et al., 2001, Campos et al., 2022) which sends signals to higher reward areas in the brain, such as the anterior cingulate cortex (ACC), orbitofrontal cortex (OFC), nucleus accumbens and amygdala (Schwartz et al., 2000). These brain areas interact with other areas related to executive functioning, such as the dorsolateral

prefrontal cortex and ventral medial prefrontal cortex, to direct eating behaviour (Schwartz et al., 2000). The ARC also connects with the lateral hypothalamus (LH), which is associated with hunger and feeding initiation. Figure 2.2 illustrates the neuronal pathway of the homeostatic system and shows the connection between peripheral satiety signals.



Figure 2.2. The neuronal pathway of the homeostatic system shows the connection between peripheral signals to stimulate or suppress food intake. These signals can include appetite signalling through ghrelin gut hormones and satiety signalling through hormones such as glucagon-like peptide 1 (GLP-1), peptide tyrosine tyrosine (PYY), leptin, pancreatic polypeptide (PP), oxyntomodulin (OXM) and Cholecystokinin (CCK). Figure reproduced from Kim et al. (2013).

2.1.2 <u>Hedonic system</u>

Food intake is a rewarding and pleasurable experience which can override the homeostatic regulation of eating (Batterham et al., 2007, Lutter and Nestler, 2009). When individuals eat foods that are rich in taste, texture, and sensory attributes, that they find pleasurable, the reward and sensory systems are activated. Sensory information related to the food stimuli activate primary sensory areas to identifies and creates the perceptual experience of food (anterior insula and frontal operculum, and primary somatosensory cortex). The rewarding attributes of food stimuli also triggers the release of neurotransmitter dopamine and dopamine 2 from ventral tegmental area (VTA) of the midbrain. Neurons in the VTA project to various areas of the brain, including the nucleus accumbens, prefrontal cortex, and amygdala. These brain-areas assess the

rewarding aspects of the food, its hedonic value, and its potential pleasurable effects. This could lead to reinforcing the desire to continue eating such foods (Schwartz et al., 2000, Campos et al., 2022). For example, food rich in sugar (sweet) and fat are highly associated with reward, and have frequently been attributed to the hedonistic features of eating in both rodents and humans (Berridge, 1991).

Hedonic regulation can override the homeostatic route by boosting the desire to eat foods that are highly appealing and pleasant during times of relative energy excess. Dysregulation of the hedonic system can contribute to overeating and obesity. Individuals may consume calorie-dense, highly palatable foods driven by the desire for rewarding experiences, even when they are not hungry. Understanding the influence of the hedonic system on food intake is essential for addressing issues related to overconsumption and obesity. The integration of sensory and reward signals in the brain can direct behaviour and eventually food intake (Schwartz et al., 2000, Campos et al., 2022).

2.2 Regulation of food intake by the gastrointestinal tract

The GI tract plays a critical role in regulating food intake by digesting and absorbing nutrients, releasing hormones that signal hunger and satiety, and providing sensory input to the brain. This section gives an overview of how food is digested and processed in the stomach. The role of gut hormones in regulating appetite and satiety sensations is then discussed.

2.2.1 Food digestion and gastric emptying

Food digestion is a complex process that begins in the mouth and continues through the GI tract, involving various organs, enzymes, and mechanical actions (Patricia and Dhamoon, 2019). The first step in food digestion is the cephalic phase responses which are physiological responses to food cues such as sight, thought, smell, and taste of food (Powley, 1977, Teff, 2000). The role of cephalic responses is to prepare for optimal nutrients digestion and absorption (Powley, 1977, Powley and Berthoud, 1985, Nederkoorn et al., 2000, Teff, 2000). Cephalic phase responses include increased salivation, gastric motility, and pancreatic (pancreatic polypeptide, glucagon, leptin, and insulin), and gastric

(ghrelin) endocrine secretions, and bile secretions (Powley, 1977, Richardson and Feldman, 1986, Soucy and Leblanc, 1998, Mattes, 2005). Cephalic phase responses also include other non-secretory responses such as changes in respiratory quotient (McGregor and Lee, 1998), changes in cardiac measures (Nederkoorn et al., 2000), increased blood pressure (Nederkoorn et al., 2000), postprandial thermogenesis (LeBlanc and Cabanac, 1989, LeBlanc, 2000), and gastric motor activity (Mattes, 1997, Nederkoorn et al., 2000). The stimulation of cephalic responses may have a role in regulating food intake, appetite, and physiological responses. For example, a previous study by Cecil et al. (1998) has shown that oral administration has delayed gastric emptying more than gastric infusion and significantly affect satiety ratings. Additionally, they showed that a significant correlation between gastric contents and hunger (r=-0.98) as well as fullness (r=0.98) after the oral administration. This suggests that the orosensory simulation induced greater effect to reduce food intake by delaying gastric emptying.

In addition to the cephalic responses, the GI tract produces and secretes digestive enzymes to break down lipids, proteins, and carbohydrates to complete the process of digestion and, subsequently, the absorption of nutrients (MacFarlane, 2018, Patricia and Dhamoon, 2019). Salivary amylase in the mouth starts to breakdown carbohydrate and after that the digestion continues in the small intestine by pancreatic amylase. Hydrochloric acid and pepsin in the stomach break down protein when a bolus of food particles enters the stomach. Then, digestion of protein continues in the small intestine by proteolytic enzymes including trypsin, chymotrypsin, elastase and carboxypeptidase which break down protein into oligopeptides. For lipids, the primary digestion begins in the small intestine by (1) pancreatic lipase that breakdown triglycerides into 2 monoglycerides and fatty acids, (2) pancreatic phospholipase that breakdown phospholipids into head groups and fatty acids, and (3) pancreatic cholesterol esterase that breakdown cholesterol esters into component cholesterol and fatty acid. The stomach is a crucial organ in the digestive system, playing a central role in food digestion and the initial breakdown of ingested food. GE is a specific step in this process that occurs in the stomach and involves the release of partially digested food into the small intestine. This section gives an overview of the physiological mechanism of GE and the impact of the GE rate on the regulation of food intake. Below is an overview of the anatomy of the stomach and its function in food digestion.

The stomach is divided into four functional parts (Soybel, 2005). The first part, the fundus, fills with air, the second part, the body, stores food, the third part, the antrum, mixes food with digestive juice and creates fluid motion, and the fourth part, the pylorus, controls the food particles movement into the duodenum. Figure 2.3 shows a schematic diagram of a stomach anatomy. The stomach can also be divided into two parts based on the pattern of gastric movement: the proximal stomach and the distal stomach (Figure 2.3) (Soybel, 2005). The fundus and the proximal part of the corpus make up the proximal stomach while the distal part of the corpus and the antrum make up the distal stomach.



Figure 2.3. A schematic diagram of stomach anatomy and the human digestive tract. Figure reproduced from Liu et al. (2021).

2.2.1.1 The process of gastric emptying

Gastric emptying is the process by which the contents of the stomach are moved into the duodenum (Cifuentes and Acosta, 2022). This is accomplished by four mechanisms/phases: (1) tonic contractions, (2) peristaltic contractions, (3) retropulsion, and (4) emptying (Soybel, 2005, Bellmann et al., 2016) (Figure 2.4). Following food ingestion, the contents of the proximal stomach are forced

towards the distal stomach by tonic contractions. After that, the stomach surface contracts and the peristaltic waves reach the pylorus to increase the width and the indentations, frequently virtually obstructs the antral lumen. Pylorus contraction causes the sphincter to narrow, which dramatically reduces the pyloric opening when the peristaltic wave arrives. Retropulsion is the process by which the chyme is pumped back into the stomach. At this stage, food and gastric juice are thoroughly combined and emulsified, which causes crushing and rubbing between stomach walls and/or food particles. The food particles first appear to be suspended due to a process that involves repeated advances, grinding, and retreats as well as the work of acids and enzymes. Lastly, the emptying process begins when the pylorus partially opens. Fluid and small particles (between 1 and 2 mm) move continually from the stomach.



Figure 2.4. The gastric emptying process. (a) Tonic contractions, (b) peristaltic contractions, (c) retropulsion, and (d) emptying. Figure reproduced from Liu et al. (2021).

2.2.1.2 Impact of gastric emptying rate on food intake regulation

GE influences subsequent food and caloric intake. A delay in GE leads to greater satiety and lower caloric intake (Halawi et al., 2017), whilst rapid GE leads to lower satiety and higher subsequent caloric intake (Gonzalez-Izundegui et al., 2021). The rate of GE is mediated by different mechanisms including biological

factors. For example, age, sex, and gut hormones showed to have a significant effect on the GE rate. It has been suggested that elderly people have a delayed GE rate compared to young people (Brogna et al., 2006). While aging was suggested to delay GE rate, a meta-analysis by Bonner et al. (2015) conducted on 49 studies including participants from 28 weeks' gestation to adulthood demonstrated that age did not significantly affect the GE rate. Females have been shown to have a slower GE rate compared with males during specific times across the life-course. In comparison with males and postmenopausal females, it has been suggested that premenopausal females have a slower GE rate (Wang et al., 2015). This could be attributed to the effect of oestrogen during the luteal phase of the menstrual cycle (Wang et al., 2015). Gut hormones such as GLP-1 and CCK decrease GE by relaxing the proximal stomach to increase stomach capacity (Rigaud et al., 1995, Schirra et al., 2006, Schirra et al., 2009) and inhibit gastric acid secretion (Layer et al., 1995, Wettergren et al., 1997, Marciani et al., 2001). These effects are dependent on stimulating vagal afferents, which inhibit reflex vagal motor pathways (Ritter and Ladenheim, 1985, Moran et al., 1994, Li et al., 1997, Tolessa et al., 1998, Tolessa et al., 2001, Näslund et al., 2002).

The rate of GE is also mediated by the ingested food properties, including energy density, and viscosity. Increasing energy density and/or viscosity of a meal has been demonstrated to delay the GE rate (Mazzawi et al., 2019, Campos et al., 2022). Using MRI, Camps et al. (2016) demonstrated that energy density and viscosity of the ingested meal have a significant impact on GE rate in normal-weight participants. Comparing GE rate following consumption of 500 ml of 4 dairy-based shakes, with different energy density (100 kcal compared to 500 kcal) and viscosity (thin and thick), showed both the energy load and the viscosity impacted the GE rate. However, increasing the energy load was shown to be the most significant factor.

Another aspect affecting the GE rate is the separation of solid and liquid within the stomach, which is known as sieving. The sieving process happens as liquids are emptied from the stomach more quickly than solids (Hinder and Kelly, 1977, Meyer et al., 1979). This usually happens as larger solid particles of mixed solid/liquid meals are held for a longer time in the stomach to undergo grinding to reduce their size before they may be emptied after a lag phase. It has been suggested that blending or homogenising a mixed solid/liquid meal could prolong satiety, as shown by a previous study that found that, in comparison to an unhomogenised meal, a mixed solid/liquid, fat- and vegetable-rich, 2573-kJ meal that was homogenised delayed GE and enhanced satiety (Santangelo et al., 1998). Another study conducted by Marciani et al. (2012) measured the GE rate of roasted chicken with vegetables and a glass of water compared to the same meal blended into soup. The gastric volume of the soup meal decreased more slowly and linearly compared to the solid/liquid meal (77 \pm 6 min), the soup meal tended to take longer to reach half-gastric emptying (92 \pm 7 min) (P = 0.06). The soup meal was also associated with greater gallbladder contraction compared to the solid/liquid meal, suggesting more CCK secretions happen with the soup meal.

2.2.2 <u>Mesenteric blood flow</u>

The small intestine is the primary site for nutrient digestion and absorption. The digestion of the chyme released from the stomach continues in the small intestine. As the chyme moves to the small intestine, blood flow through the capillaries that supply the gut wall increases to facilitate food digestion, which can be measured as a rise in the blood flow through the superior mesenteric artery (SMA) (Jeays et al., 2007). The SMA is a major supplier of blood to various sections of the GI tract, including the duodenum, the small intestine, and the transverse colon (Jäger et al., 1986, Moneta et al., 1988). Blood flow in the SMA reflects absorptive, secretory, and motor functions in the splanchnic organs. Each of these activities rises after mealtime and subsequently causes a significant rise in splanchnic blood flow (Guyton, 2006). Blood flow in the SMA increases, reaching its peak around 30 to 45 minutes after eating, and returning to baseline levels after approximately 90 minutes (Batton et al., 1983, Sieber et al., 1991, Spencer et al., 2000). However, these changes in the SMA blood flow are influenced by various factors, such as meal composition and caloric content of the meal (Moneta et al., 1988, Sieber et al., 1991, Sieber, 1992, Sidery and Macdonald, 1994, Parker et al., 1995, Jeays et al., 2007), as well as the rate of GE (Sidery et al., 1994). Previous studies have shown that the peak blood flow

in the SMA is positively correlated with the energy content of a meal (Sidery and Macdonald, 1994, Parker et al., 1995, Someya, 2007). The mesenteric circulation supplies oxygen and nutrients to the small intestine, eliminates waste products, and transports absorbed substances away from it (Granger et al., 2011). Studying the blood flow in SMA in this thesis may offer valuable information for examining the digestive processes in both health and illness, which could have a role in food intake regulation.

2.2.3 Small bowel water content

The amount of water present in the small intestine, "small bowel water content"(SBWC), is another factor that plays a vital role in food intake regulation. On a daily basis, around 9 litres of fluid enter the small intestine, with 7 to 8 liters being reabsorbed, while only 1 to 1.5 liters progress into the colon (Volk and Lacy, 2017). Following food consumption, the presence of plasma PYY in the bloodstream facilitates the absorption of fluids and nutrients across the intestinal mucosal surface, resulting in a reduction of SBWC (Savage et al., 1987, Benelam, 2009). Consequently, this leads to the activation of the ileal brake mechanism and the release of satiety hormones (Moneta et al., 1988, Sieber et al., 1989). The ileal brake is defined as an inhibitory mechanism that controls the transit time of food moving through the small intestine to optimise nutrient absorption and digestion (Jeays et al., 2007). A previous study by Read (1992) demonstrated that nutrient infusion into the ileum leads to reduced duodenal and antral peristaltic contractions. However, no alteration was found in the transit time of the small bowel when nutrients were infused into the colon. The ileal brake occurs following the stimulation of enteroendocrine cells and mucosal afferent neurons. This system is controlled by PYY, GLP-1, and CCK. Ileal brake, is also influenced by GE rate, caloric load, and food composition (De Graaf et al., 2004).

2.3 Appetite and satiety regulators

2.3.1 <u>Gut hormones</u>

Ghrelin

Ghrelin is a 28 amino acid peptide hormone released from the gastric fundus A cells in the stomach and is currently considered the only known orexigenic

peptide hormone (Nakazato et al., 2001). In contrast to the other peptide hormones, ghrelin rises before meals ingestion and decreases afterwards, to stimulate hunger/appetite and food intake (Nakazato et al., 2001). Suppression of postprandial secretions of ghrelin is proportional to calorie content (Callahan et al., 2004). Lowest levels of ghrelin can be detected within 20-60 minutes postprandial, depending on a meal composition (Bowen et al., 2006, Carroll et al., 2007, Yang et al., 2009). Ghrelin can increase gastric acid secretion and accelerate GE of solids and liquids (Matsuda et al., 1999, Asakawa et al., 2003, Murray et al., 2005) by binding directly to the ghrelin receptor 'growth hormone secretagogue (GHS)' in the stomach, or through mediation by vagal nerve (Levin et al., 2006).

Peptide tyrosine tyrosine

PYY is a peptide hormone released from luminal cells of the ileum in the small intestine (Adrian et al., 1985b, Grandt et al., 1994). PYY can inhibit gastric acid secretions (Adrian et al., 1985a), and secretion of electrolytes and fluid in the small bowel (Savage et al., 1987). Additionally, it can act as an ileal brake which decreases motility of gastric and intestinal transit (Beglinger et al., 2001). All these mechanisms combined lead to satiety feeling. PYY is found endogenously in two forms: PYY₁₋₃₆ (36 amino acid peptide) and PYY₃₋₃₆ (34 amino acid peptide) (Grandt et al., 1994). The bioactive form of PYY is PYY₃₋₃₆, which is produced by cleavage of N terminals Tyr-Pro dipeptide from PYY1-36, by dipeptidyl-peptidase 4 (Eberlein et al., 1989, Grandt et al., 1994, Ritter, 2010). PYY performs its function by acting on the Y receptors (Y1, Y2, Y4, Y5, and Y6) (Grandt et al., 1994, Dumont et al., 1995). Endogenous PYY levels reach their lowest point during fasting and rise after eating. This increase is in proportion to calorie intake and with protein and carbohydrate rich meals (Adrian et al., 1985b, Batterham et al., 2003, Batterham et al., 2006). Postprandial PYY peaks within 30-90 minutes after food intake and is proportional to type (i.e., carbohydrate, protein, fat content) and volume of food (le Roux et al., 2006, Essah et al., 2007, Yang et al., 2009).

Cholecystokinin

CCK is a 33-amino acid peptide hormone released postprandially from luminal cells in the duodenum and jejunum (Mutt and Jorpes, 1968, Murphy and Bloom, 2006). There are two types of CCK receptors present in the peripheral tissues and the CNS: CCK1 and CCK2 (Moran and Kinzig, 2004). The forms of circulating CCK are many, ranging in chain length from eight to eighty-three amino acids. CCK-8, CCK-22, CCK-33, and CCK-58 are the major forms of CCK circulating in the blood, all having the same blinding power (attribute) (Cantor and Rehfeld, 1989, Ji et al., 1997, Luo et al., 2014). CCK1 receptors are present within CNS in areas that are involved in food intake regulation, i.e. the dorsomedial hypothalamus, the area postrema (AP) and the nucleus of the solitary tract (NTS) (Moran et al., 1986).

The release of CCK into plasma is greatly stimulated by the consumption of fat, protein, and their final digestion products, but the influence of carbohydrate consumption on CCK release is modest (Liddle et al., 1985). CCK levels rise over 10-30 minutes after food ingestion and activates the release of bile acids from the gallbladder and pancreatic enzymes, which leads to digestion of fat and protein (Liddle et al., 1985). However, CCK has a short half-life, lasting only a few minutes after the consumption of a meal (Geary, 2004). The release of CCK and signalling via CCK receptors mediate stimulation of PYY release and inhibition of ghrelin which moderates food intake and appetite (Degen et al., 2007). CCK directly activates the vagal afferents, which inhibit GE and gastric secretions (Fried et al., 1991). In animal models, peripheral administration of CCK reduces GE through interactions of a gut-brain mechanism involving oxytocin (Olson et al., 1992, Higham et al., 1997, Goyal et al., 2019). Postprandial CCK concentrations inhibits human GE by inducing pylorus contraction and antrum and proximal stomach relaxation (Liddle et al., 1986). Thus, the collective actions of CCK contributes to the feeling of fullness and satiety.

Glucagon-like peptide-1

GLP-1 is an incretin 30-amino acid peptide hormone produced by the splitting of the preproglucagon precursor molecules (Dhanvantari et al., 1996). GLP-1 is produced from luminal cells in the ileum in response to meal intake (Holst et al
2007). Receptors of GLP-1 are distributed in areas involved in food intake regulation, including islets of pancreas, the brain, and the whole GI tract (Nauck et al., 1993, Willms et al., 1996). GLP-1 performs its action through GLP-1 receptors in the NTS and AP of the brainstem, superior optic nuclei and ARC of hypothalamus, PVN, and vagus (Chaudhri et al., 2006a, Parkinson et al., 2009). GLP-1 is released 10-15 minutes postprandially and its blood levels peaks between 30-60 minutes (Flint et al., 1998) with levels remain elevated for at least 180 minutes following meals (Verdich et al., 2001a). Therefore, it plays a role in satiation during and after meal intake. Human and animal studies showed that infusions of GLP-1 induce subjective satiety and reduce food intake (Melhorn et al., 2014). GLP-1 is also an incretin hormone that stimulates the release of insulin and inhibits the production of endogenous glucose and the glucagon release (Willms et al., 1996). Regardless of the meal type, continuous administration of GLP-1 slows the rate of GE and enhance distal gastric stagnation (Näslund et al., 2004). Exogenous GLP-1 has been shown to enhance pyloric tone, reduce vestibular and duodenal motility and relax the proximal stomach during fasting and feeding states (Delgado-Aros et al., 2002, Suganuma et al., 2020). Furthermore, GLP-1 induces increased satiety and supresses pancreatic secretions (Schirra et al., 2006, Punjabi et al., 2011).

Glucagon-like peptide-2 (GLP-2)

GLP-2 is produced by modifying the splicing of pre-proglucagon and its main function is associated with trophic effects on the intestinal mucosa. GLP-2 is a 33 amino acid peptide hormone, co-secreted with GLP-1 from luminal cells in the ileum (Orskov et al., 1986). Some studies have found that GLP-2 could delay the GE of liquids in NW participants (Nagell et al., 2004, Berg et al., 2014) and reduce gastric acid production (Meier et al., 2006). GLP-2 acts on the GLP-2 receptors which are localised in subepithelial myofibroblasts, intestinal syncytium of cells located underneath the epithelium, enteroendocrine cells and enteric neurons (Yusta et al., 2000, Bjerknes and Cheng, 2001, Ørskov et al., 2005, Guan et al., 2006).

Amylin

Amylin is a 37 amino acid peptide hormone which is released into the plasma with insulin in response to food intake. It performs its action through stimulating histamine H1 receptors (Mollet et al., 2001). Amylin directly activates some brain areas that have satiating functions, including the AP and possibly the VTA. Furthermore, amylin has been shown to promote activation of other satiating hormones such as CCK, reduce gastric secretion, delay GE, reduce food intake and postprandial glucose elevation (Lutz et al., 1995, Lutz, 2006).

Oxyntomodulin (OXM)

OXM is a 37-amino acid peptide hormone produced from the splitting of preproglucagon precursor molecules (Le Quellec et al., 1992). OXM is released into the plasma postprandially from luminal cells, in response to caloric intake (Le Quellec et al., 1992). Although it has low affinity to this receptor family, OXM mediates its effects through GLP-1 receptor (Baggio et al., 2004, Lebrun et al., 2006). It also decreases secretion of gastric acid, delays GE of the stomach and reduces food intake (Schjoldager et al., 1989). Intravenous administration of OXM in humans has been shown to increase energy expenditure, supress ghrelin secretion (Chaudhri et al., 2006a, Chaudhri et al., 2006b, Wynne et al., 2006) and decrease appetite and food intake (Cohen et al., 2003, Wynne et al., 2006). Furthermore, OXM induces weight loss in both NW and people with obesity (Cohen et al., 2003, Wynne et al., 2006).

Glucagon

Glucagon is a 29-amino acid peptide hormone released from alpha cells in the pancreas (Jones et al., 2012a). Glucagon performs its action through the glucagon receptor which is located in cells in the kidney, liver, and other tissues (Svoboda et al., 1994). It is secreted into the portal vein in response to hypoglycaemia and increases blood glucose concentrations and energy expenditure (Jones et al., 2012a). Administration of glucagon intravenously has been shown to reduce the intake of food by delaying GE rate (Geary et al., 1992).

Pancreatic polypeptide (PP)

PP is a 36-amino acid peptide hormone released postprandially by pancreatic islet PP cells under vagal control, in response to calorie intake (Larsson et al., 1975, Adrian et al., 1976, Schwartz et al., 1978). Similar to PYY, PP performs its function by acting on the Y receptor family, specifically the Y4 receptor in the ARC of the hypothalamus (Chaudhri et al., 2006a). PP has a role in regulating pancreatic secretion and gastric and gallbladder motility (Adrian et al., 1979). PP induces a delay in GE, inhibits pancreatic enzymes and relaxes the gallbladder (Asakawa et al., 2003). Intravenous administration of PP increases energy expenditure and decreases GE and food intake (Clark et al., 1984).

2.3.2 Associations between gut hormones, reported appetite and measured food intake

Previous studies have measured the association between postprandial concentrations of endogenously released gut hormones, reported appetite and subsequent food intake. A study conducted by Bowen et al. (2006) administered four different types of liquid preloads (50 g of whey, soy, gluten or glucose) and measured association between postprandial concentrations of gut hormones and ad libitum energy intake (3 hours after test meals) in NW and overweight participants. They found that ad libitum energy intake was predicted by glucose (P=0.012) and ghrelin (P=0.039), and inversely predicted by CCK (P=0.056) and insulin (P=0.001) regardless of meal type. This accounted for 5.3% of the variance in ad libitum energy intake. However, other two studies found no association between gut hormones and subsequent ad libitum energy intake. A study carried out by Gibbons et al. (2016) in 16 participants with obesity showed no correlation between postprandial concentrations of CCK following highcarbohydrate (83.6% carbohydrate, 3.2 % fat & 13.2 % protein) or high-fat (39% carbohydrate, 50.3 % fat & 11.7 % protein) meals and ad libitum energy intake, which was administered 3 hours after the test meals. Another study by Hengist et al. (2023) found that subsequent ad libitum energy intake after lowfat (LF, 75% carbohydrate and 10% fat) and low-carbohydrate (LC, 10% carbohydrate and 75% fat) diets did not correlate with mean postprandial PYY (LC diet: P=0.19, LF diet: P=0.98), GLP-1 (LC diet: P=0.68, LF diet: P=0.60), GIP (LC diet: P=0.73, LF diet: P=0.34), active ghrelin (LC diet: P=0.41, LF diet:

P=0.76), total ghrelin (LC diet: P=0.19, LF diet: P=0.76), or leptin (LC diet: P=0.77, LF diet: P=0.27) in 20 NW participants. From the above studies, it seems that there are conflicting results for the association between endogenous concentrations of gut hormones and subsequent energy intake.

Some previous research has demonstrated that gut hormones administered exogenously affect appetite and subsequent food intake. For example, a study conducted by Batterham et al. (2003) reported that intravenous PYY (2 nmol per square meter of body-surface area) decreased ad libitum energy intake (P<0.001) two hours after the infusion in both NW and participants with obesity. Another study by Wren et al. (2001) reported that intravenous infusion (5pmol/kg/min) increased the VS hunger scores (P<0.05) and ad libitum energy intake (P<0.001) four hours after the infusion.

2.3.3 Other appetite and satiety regulators

Blood glucose

Glucose plays a crucial role in regulating energy consumption (Nakrani et al., 2023). Following digestion, carbohydrates are broken down into simple sugars like glucose, proteins are converted to amino acids, and lipids are broken down into fatty acids and glycerol. The body uses these smaller molecules for energy, growth, and repair when they are taken up by the bloodstream (Alberts, 2017).

Blood glucose levels rise following a meal, which activates the pancreas to secrete more insulin, which leads to glucose being stored as glycogen in the liver (Nakrani et al., 2023). When blood glucose levels decline several hours after a meal, the liver subsequently releases glucose back into the blood, reducing fluctuations in blood glucose. A drop in blood glucose may stimulate appetite and food intake (Wyatt et al., 2021) with hypoglycaemia having been reported to increase hunger feelings and the GE rate (Mayer, 1955, Schvarcz et al., 1995, Campfield et al., 1996). Blood glucose concentrations may affect hypothalamic satiety centres including the ventromedial and arcuate hypothalamic nuclei, which have important roles in glucose metabolism (Gao and Horvath, 2008). Glucose performs its action on appetite and satiety through glucoreceptors in the hypothalamic nuclei (Ritter et al., 1981, Kumar, 1999).

Insulin

Insulin is a 51 amino acid peptide hormone that is released from the beta cells of Langerhans in the pancreas. The main function of insulin is to maintain normal blood glucose levels by promoting glucose uptake by cellular tissues, regulating the metabolism of carbohydrates, lipid, and protein, and performing mitogenic effects on cell division and growth (Wilcox, 2005).

Insulin mediates appetite-suppressing effect by acting on insulin receptors in the ARC of the hypothalamus (Havrankova et al., 1978, Houten et al., 1979). In humans, intranasal infusion of insulin decreased fMRI signals in the hypothalamus (Opstal et al., 2017), which is the main area for haemostatic control, and regulates the dopaminergic reward system (Kullmann et al., 2013, Tiedemann et al., 2017, Thanarajah et al., 2019).

Leptin

Leptin is a 16 kD adipokine and acts via the leptin receptor (Chen et al., 1996). It is synthesised and secreted by white adipose tissue. Normal concentrations of leptin decrease the drive to eat and promote energy expenditure via its role in autonomic output and several neuroendocrine axes. Conversely, decreased concentrations of leptin reduces energy utilization, increases the drive to eat and leads to behavioural adaptations to the reduced energy stores (Ahima et al., 1996, Myers et al., 2009). Leptin also plays a major role in regulating metabolism by modifying the secretion of insulin, lipid metabolism, and the synthesis of hepatic glucose (Kulkarni et al., 1997, Liang and Tall, 2001, Minokoshi et al., 2002, Pocai et al., 2005, Nogueiras et al., 2007). In addition, leptin may delay GE through interacting with the CCK and the vagus nerve (Asakawa et al., 1999). Accelerated GE has been found with diminished levels of leptin (Asakawa et al., 1999).

Free fatty acids (FFAs)

FFAs are as species of lipid released during the lipolysis process from adipose tissue and other cell types. FFAs perform active roles in several biological processes in addition to their primary involvement in energy delivery and as structural elements of membranes. FFAs can have an impact on endothelial cells' (Frommer et al., 2015), adipocytes' (Schaeffler et al., 2009), or macrophages' (Håversen et al., 2009) gene expression. FFAs can also affect chemokine and

cytokine production (Frommer et al., 2015, Honda et al., 2015, Hung et al., 2015), adhesion molecule gene expression (Miles et al., 2001, Livingstone et al., 2014), and the emergence of pro-inflammatory and inflammation-promoting lipid-derived substances (Miles et al., 2001, Serhan et al., 2008, Livingstone et al., 2014, Hung et al., 2015).

It has been suggested that FFAs are involved in the regulation of energy haemostasis by stimulating secretions of insulin, GLP-1 and glucagon, and reducing intestinal motility through expression of FFA receptors (FFAR) in the pancreas and intestine (Hara et al., 2014).

Triglycerides

Triglycerides are the predominant dietary lipid in fats and oils, whether sourced from plants or animals (Lichtenstein, 2013). They are made up of three fatty acid molecules linked to a glycerol molecule. In the human body, they are the primary storage source of lipid and energy. They are predominantly synthesised through the glycerol phosphate pathway (Viecili et al., 2017). The liver and adipose tissue are the two primary locations for endogenous triglyceride production (Bayly, 2014). The proximal gastrointestinal tract hydrolyses triacylglycerols into fatty acids and monoacylglycerols that can influence GI functions, hormone release, and satiety (Armand et al., 1999). Obesity or hypertriglyceridemia are metabolic problems that can result from an imbalance in this mechanism (Viecili et al., 2017).

Fasting concentrations of triglycerides positively correlate with visceral fat area (Marston et al., 2019, Sukkriang et al., 2021). During prolonged fasting and starvation, adipose tissue mobilises triglycerides, which increase concentrations of FFAs in the blood circulation (Becker, 2001). FFAs in the blood are quickly absorbed by the liver during a fast or starvation and then return to the blood as triglycerides (Palmer et al., 1978, Guiducci et al., 2006).

Elevated triglyceride levels could induce leptin resistance, which increases the desire to eat and reduces calorie expenditures unrelated to energy need, may have developed as a signal to the brain that a person is starving. Indeed, it is believed that elevated triglycerides could impair the transport of leptin to the blood-brain barrier, causing peripheral resistance (Banks et al., 2004). In

addition, elevated blood triglycerides are frequently associated with insulin resistance as a clinical indicator of the metabolic syndrome (Grundy, 1999, Ma et al., 2020).

Table 2.1. The origin, major receptor and functions of gut hormones and appetiteand satiety regulators involved in the regulation process of food intake.

Gut	Site of	Main	Effect on	Other functions
hormone	secretion	receptors	food intake	
Ghrelin	A cells in	GHS	Increase	• Stimulate secretion of growth
	the gastric	receptor		hormones
	fundus			Stimulate gastric motility
				(emptying)
Peptide	Luminal	Y2	Decrease	• Activate the ileal brake to delay
tyrosine	cells in	receptor		motility of gastric and small
tyrpsine	ileum			intestine
(PYY)				• Reduce gastric acid secretions
	Luminal	CCK 1	Decrease	• Inhibit gastric emptying and
Cholecystok	cells in	receptor		gastric secretions
inin (CCK)	duodenum			• Induces pylorus contraction, and
	and			antrum and proximal stomach
	jejunum			relaxation
				• Increase gallbladder
				contractions
				• Increase pancreas secretions
Glucagon-	Luminal	GLP-1	Decrease	• Stimulate secretion of insulin
like-peptide	cells in	receptor		• Reduce gastric emptying
1 (GLP-1)	ileum			• Supresses pancreatic secretions
Glucagon-	Luminal	GLP-2	Decrease	• Reduce gastric emptying of
like-peptide	cells in	receptor		liquid
1 (GLP-2)	ileum			• trophic effects on intestinal
				mucosa
Amylin	Pancreas	Histamine	Decrease	Reduce gastric emptying
	beta cells	H1		Reduce gastric acid secretion
		receptor		 Increase levels of postprandial
				glucose

Oxyntomod	Luminal	GLP-1	Decrease	Reduce gastric emptying
ulin (OXM)	cells in	receptor		Reduce gastric acid secretion
	ileum			
Glucagon	Pancreas	Glucagon	Decrease	• Increase levels of blood glucose
	alpha cells	receptor		• Increase energy expenditure
Pancreatic	Pancreatic	Y4	Decrease	• Delay gastric emptying rate
polypeptide	islet PP	receptor		 Inhibition of pancreatic enzymes
(PP)	cells			• Relaxation of the gallbladder
Insulin	Langerhan	Insulin	Decrease	Maintain normal blood glucose
	s in the	receptor		levels
	pancreas			• Affect brain areas of haemostatic
				and reward mechanisms
Blood	Bloodstrea	Glucorept	Decrease	Affecting hypothalamic satiety
glucose	m after	or		centres
	food			• Hypoglycemia could activate
	intake			hunger and GE rate
Leptin	Adipose	Leptin	Decrease	• Interact with the CCK and vagus
	tissue	receptor		nerve to delay gastric emptying
				 Modifying secretion of insulin,
				synthesis of hepatic glucose,
				and lipid metabolism
Triglyceride	Liver and	-	Fasting	Inducing leptin resistance which
S	adipose		triglyceridem	lead to satiety, and calorie
	tissues		ia	expenditure reduction
			increases	
			food intake	
Free fatty	Released	FFAR	Decrease	• Stimulating secretions of insulin,
acids (FFAs)	during fat			GLP-1, and glucagon, and
	lipolysis			reducing intestinal motility
	of adipose			through expression of FFAR
	tissues			

2.4 Effect of macronutrient compositions on food intake regulation

Macronutrients have been suggested to exert different effects on appetite and satiety sensations, even when they contain the same number of calories (Holt et al., 1995, Bludell et al., 1996). Evidence suggests that there is a hierarchy in

satiating effect, with protein being more satiating than carbohydrate, and carbohydrate being more satiating than fat (Westerterp-Plantenga et al., 2006, Veldhorst et al., 2008, Johnstone, 2013). Studies have shown that individuals who follow a high-protein diet, about 30% of total energy from protein, tend to feel fuller and eat less compared to those who have a normal protein intake, about 18% of the total energy intake (Moran et al., 2005, Leidy et al., 2007). Additionally, an increased protein intake has been proposed to help in weight maintenance. In a randomized controlled trial involving 256 adults with overweight or obesity who were following a weight loss program, it was observed that weight regain was more pronounced in individuals who adhered to a low-protein diet (10–15% of total energy from protein) compared to those who followed a high protein diet (23–28% of total energy from protein) (Aller et al., 2014). Previous studies comparing protein, carbohydrate and fat suggest that protein is more satiating than carbohydrate and fat; however, there is inconsistency in findings.

The inconsistency in findings regarding the satiating effects of protein, carbohydrates, and fats highlights the need for a deeper understanding of how macronutrients influence food intake regulation. A crucial step in understanding how macronutrients influence appetite and eating behaviour is employing appropriate methodologies. The next section presents the methodologies commonly used to measure food intake regulation, which is the groundwork for the work presented in this thesis.

2.5 Effect of taste on food intake regulation

A person's choice of food is a dynamic process that evolves over the course of their life and is impacted by a variety of environmental and personal factors (Barclay and Brand-Miller, 2011). A food product's likelihood of being purchased and consumed by consumers is influenced by a variety of factors, including personal preferences, cultural, environmental, and contextual factors as well as the food's sensory qualities (Furst et al., 1996, Drewnowski et al., 1997, Olsen et al., 2012). Previous studies on food choice have indicated that people rank taste as one of the most significant considerations when choosing what to eat (Lennernäs et al., 1997, Glanz et al., 1998, Biloukha and Utermohlen,

2001). A study conducted by Kourouniotis et al. (2016) on 1306 university students reported that 82% of participants regarded taste as very important when choosing food. Participants who rated taste as very important factor also tended to consume less fruit (P= 0.03) and vegetables (P=0.05) and had a lower quality diet (P=0.001). Additionally, participants had a considerably higher likelihood of consuming foods high in salt, sugar, and fat such as fruit juice, soft drinks, cakes and puddings, sweet pastries, pizza, hot chips, and potato chips, takeout meals, pizza, chocolate, and confectionery (P=0.001). A higher intake of salt, fat and sugar has been connected to an imbalance in adult energy (Grimes et al., 2013) which is a major factor contributing to the rising rates of overweight and obesity globally (Hill et al., 2000, Bermudez and Gao, 2011, Grimes et al., 2013).

2.6 Approaches to measure regulation of food intake

The focus of this thesis is to gain insights into how food intake is regulated in adults with and without obesity and to ascertain whether there are differences in response to different macronutrients. Measuring the regulation of food intake involves assessing the various factors that influence eating behaviour including appetite and satiety. This section summarises key approaches used in measuring the regulation of food intake. This includes appetite/satiety rating scales, ad libitum meal intake, GI responses, and neuronal signalling to food intake, as well as eating behaviour questionnaires.

2.6.1 <u>Appetite Rating Scales</u>

Appetite rating scales are used to assess and quantify an individual's subjective feelings of hunger, fullness, and desire to eat. These scales are valuable in studying appetite regulation, understanding eating behaviours, and evaluating the effectiveness of dietary interventions. Appetite and satiety ratings can be measured through visual analogue scales (VAS), which were devised and validated by Blundell et al. (1987) to rate the subjective feelings of desire to eat, prospective food intake, fullness, and hunger. VAS is straightforward and commonly used tool to assess appetite and satiety sensations, and used in research and clinical settings to regularly evaluate subjective feelings (Stubbs et al., 2000). It is commonly represented as 100 mm horizontal lines that are connected at both edges by opposing descriptors, such as "extremely hungry" at one end and "completely full" at the other (Hill and Blundell, 1982). Participants self-report their level of hunger and fullness by placing a mark on a horizontal line to indicate the strength of a subjective sensation or a certain state at a specific time, allowing the sensation to be measured and quantified. Interpretation of the VAS scale is typically clear-cut as the descriptive terms are already presented at the end of each line (Stubbs et al., 2000). A variety of questions about satiety are commonly asked in VAS which comprise the following four main domains: desire to eat (DTE), hunger, fullness, and prospective food intake (PFI).

The pen and paper (P&P) format of the VAS scale are used in the interventional studies of this thesis. Previous studies used a composite satiety score to create

an overall satiety score (Van Can et al., 2014). In the composite satiety score, multiple factors or variables related to satiety and measured by the VAS are combined. In this thesis, participants were asked to report their appetite and satiety feeling by answering VAS questionnaire of DTE, hunger, fullness, and PFI (Appendix 10.1). In addition, a composite satiety score (CSS) was calculated using the following equation (Hansen et al., 2018).

 $CSS (mm) = \frac{[fullness + (100 - DTE) + (100 - hunger) + (100 - PFI)] mm}{4}$

The satiety domain was not included in the VAS as may be confused with the fullness domain. Hence, the satiety domain was not included in the calculation of the CSS. In Chapters 5, 6, and 7 of this thesis, the VAS scale (DTE, hunger, fullness and PFI domains) and CSS were applied to measure appetite and satiety responses.

2.6.2 Satiety quotient

Satiety quotient (SQ) is another tool used in appetite studies to quantify and compare satiety effects of meals or foods relative to their energy content. The tool was developed by Green et al. (1997). It aims to provide a numerical value that reflects how well a particular food or meal satisfies hunger and reduces appetite.

SQ is typically calculated for each VAS domain (DTE, hunger, fullness, and PFI) by dividing changes in the satiety VAS scales by the energy content of a meal. The following equation is used to measure SQ for each VAS domain (Drapeau et al., 2007, Drapeau et al., 2013)

 $VAS(mm/100 \text{ kcal}) = \frac{[rating pre-eating episode - rating post-eating episode] mm}{[energy intake of eating episode kcal] x 100}$

The possible range of SQ is between -20.5 and 20.5 where a higher SQ for each VAS domain represents higher appetite/satiety responses and lower values represents weaker responses (Drapeau et al., 2013). The mean of the four SQ is used to determine satiety phenotypes of participants. This typically refers to individual variations in experiencing satiety/fullness sensation after eating. Some individuals may find certain foods more filling, while others may not experience the same level of satiety from those foods. If the mean SQ is ≤ 8

mm/100 kcal then the person is classified as having a low satiety and if the mean SQ is ≥ 8 mm/100 kcal, then they would be classified as having a high satiety (Drapeau et al., 2013). Low phenotypes refer to individuals who struggle to accurately identify their hunger sensations before or after a meal (Drapeau et al., 2013). Thus, SQ measurements allow the identification of factors that contribute to individual differences in satiety and to potentially develop more effective dietary interventions to promote satiety and manage appetite. In this thesis, SQ was used in Chapter 7 to compare satiety between high-protein and high-carbohydrate drinks.

2.6.3 Ad Libitum meal intake

Ad libitum meal intake refers to a method of food consumption in which individuals are allowed to eat freely and without imposed portion control or specific restrictions. When participants are given ad libitum access to food, they can consume as much, or as little, as they want until they feel satisfied or full. An ad libitum meal intake is used to quantify satiety sensation by recording the amount and energy consumed to satiation after a test preload has been consumed for a particular period of time (Blundell et al., 2010). This approach has been shown to have a good degree of intra-individual reproducibility in normal-weight adults (Arvaniti et al., 2000, Gregersen et al., 2008, Nair et al., 2008) and those with obesity (Lara et al., 2010, Horner et al., 2014b). In this thesis, the ad libitum lunch meal was used in the Chapter 7 to compare satiety responses between high-protein and high-carbohydrate drinks.

2.6.4 Gastrointestinal responses to food intake

Alongside the subjective ratings of appetite and satiety, there is a need to measure more objective physiological responses that relate to satiety and appetite and their subjective ratings (De Graaf et al., 2004). Measuring gastrointestinal responses to food intake involves assessing various physiological and biochemical changes that occur in the digestive system in response to eating. These measurements give insights into how the body processes food, absorbs nutrients, and regulates appetite. Below are some common methods for measuring GI responses to food intake.

2.6.4.1 Gut hormones

Section 2.3.1 demonstrated the relationships between gut hormones and their role in appetite/satiety sensation. Studying gut hormones as an objective biomarkers of appetite regulation can provide valuable insights into how these hormones influence hunger, satiety, and overall appetite. These data help in the understanding of the mechanisms involved in appetite control and may have implications for managing conditions like obesity and eating disorders. Satiety/appetite research studies measuring gut hormones typically involve collecting blood samples from participants before and after an intervention, including a test meal and/or dietary changes. Changes in hormone levels in response to the interventions or across groups (for example, NW vs. Obese) are then assessed. The most used gut hormones used to assess satiety are CCK, PYY, ghrelin, and GLP-1 which are usually measured during fasting and postprandially (Delzenne et al., 2010, Yeomans et al., 2016).

Although, these hormones are considered the gold standard for assessing satiety, they have some limitations. Gut hormones degrade quickly, thus blood samples need to be mixed with enzyme inhibitor to prevent degradation (Yi et al., 2015, Malandrino and Smith, 2018). Measuring gut hormones is expensive and invasive compared with the VAS scale. However, combining gut hormone measurements with other physiological and behavioural assessments can provide a more comprehensive understanding of appetite control.

2.6.4.2 Gastric emptying

Measuring GE is important for understanding satiety sensations. Several methods are available to assess GE, including gamma camera scintigraphy, isotope respiration test (breath test), and magnetic resonances imaging (MRI). These methods are briefly described below.

Scintigraphy

Scintigraphy is a commonly used, and highly accurate, method to measure GE. In this procedure, a small amount of a radioactive substance (usually technetium-99m sulfur colloid) is mixed with a solid or liquid meal. The individual then undergoes imaging using a gamma camera at specific time intervals to track the movement of the radioactive meal through the digestive tract. The rate at which the radioactive meal leaves the stomach provides information about GE. Scintigraphy is considered a gold standard method for measuring GE. It is commonly used and highly accurate method. While it provides a direct measurement for GE, it does involve radiation exposure (Schwizer et al., 2003). Scintigraphy has been widely used to assess GE in NW and Obese participants after solid, semi-solid, and liquid meal meals (Feinle et al., 1999, Buchholz et al., 2013, Gonzalez-Izundegui et al., 2021).

Breath tests

Breath Tests are validated, reliable, and non-invasive test tool for measuring GE rates of solids and liquids without radiation exposure. They assess GE by measuring the appearance of specific gases in the breath. The individual consumes a meal or drink containing a non-absorbable, stable isotope-labelled substrate (e.g., ¹³C-labelled octanoic acid). As the labelled substrate is digested and absorbed, it is metabolised in the liver and exhaled in the breath as labelled CO₂. The rate of appearance of labelled CO₂ in the exhaled breath in the breath correlates with GE. Although the technique is non-invasive, it is an indirect measurement of GE, and highly time consuming. Similar to the scintigraphy, breath tests have been widely used NW and Obese participants after solid, semisolid, and liquid meal meals (Mora et al., 2005, Horner et al., 2014a).

Magnetic resonance imaging (MRI)

The use MRI in measuring GE and GI responses has increased in the past decade. MRI provides detailed and direct assessment of the GE (Carbone et al., 2010, Fruehauf et al., 2011). It can capture dynamic changes during the process of GE, making it suitable for studying the process under various conditions. MRI does not require radiation exposure or invasive procedures like scintigraphy, making it a safe option for repeated measurements. MRI has been validated against the gold standard methods of measuring GE (scintigraphy) in healthy participants after both liquid and solid meals (Kunz et al., 1999, Teramoto et al., 2012, Khalaf et al., 2020a) and also in patients with gastroparesis (Hayakawa et al., 2017), diabetes (Parkman et al., 2010), and dysphagia (Menys et al., 2017). Despite the great benefits of MRI to assess GE, it is expensive and may not be readily available in all clinical or research settings. Also, MRI might not be

suitable for individuals who are claustrophobic, or have MRI contraindications (i.e., pacemaker, metallic implants). Due to the relevance of MRI to the work conducted in this thesis, the principle of MRI is covered in next chapter.

2.6.5 <u>Mesenteric blood flow</u>

Blood flow in the SMA can be assessed using a diagnostic method known as Duplex ultrasonography. This approach combines conventional ultrasound imaging with Doppler ultrasound to measure the velocity and direction of blood flow in the SMA. It is a non-invasive and widely used technique for examining blood flow in various arteries and has been applied in the assessment of SMA blood flow (Ritenour, 1990, Perko, 2001). The measurement of SMA blood flow can also be performed using phase contrast MRI, a method that allows for quantitative imaging of flowing fluids. This technique relies on the principle that spins moving along a magnetic field gradient produce a phase shift that is proportionate to their velocity. Several studies have used phase-contrast MRI to evaluate changes in SMA blood flow, both during fasting and following food intake (Chowdhury et al., 2016, Terlouw et al., 2022). For example, a study conducted by Totman et al. (2009) measured SMA blood flow following an oral glucose tolerance test by phase contrast MRI in nine NW participants. After the oral glucose tolerance test, they observed a significant increase in SMA blood flow from baseline levels of 2.3 ml/s to 12.9 ml/s, with participant-specific peak times ranging from 22 to 51 minutes.

2.6.6 Small bowel water content

Small bowel water content (SBWC) represents free fluids within the small bowel (Hoad et al., 2007). SBWC is a useful biomarker that shows how secretion and absorption are balanced in health and disease, as well as how treatments are working. The visualisation and quantification of water content in the digestive system provide valuable insights into understanding the mechanisms involved in food regulation (Dellschaft et al., 2022). Previous research indicates that the balance of absorption and secretions (salivary, gastric, pancreatic, biliary, and intestinal) in response to luminal contents, as well as the underlying motility patterns—whether fed or fasting—as well as the velocity of material delivery from the stomach, will all influence SBWC (Dellschaft et al., 2022). Hence, studying the SBWC is a novel measure of GI function and pathology.

Before the development of MRI, it was difficult to measure SBWC. Previously, a variety of techniques, including radioisotopic imaging, intubation, and aspiration, have been used to conduct comprehensive research on the factors affecting pancreatico-biliary secretion and gastric emptying, which ultimately input into the small intestine (Sha, 2000, Chowdhury and Forsmark, 2003, Van Den Abeele et al., 2017). MRI is a very useful tool to measure SBWC due to its non-ionizing, non-invasive, and good sensitivity to mobile water (Dellschaft et al., 2022). Initially, echo-planar magnetic resonance imaging sequence was used to measure SBWC (Hykin et al., 1994, Adkin et al., 1995). Since the early work, modifications to coil design, sequence design, and field strength have permitted major improvements in image quality. Hoad et al. (2007) developed and validated a new technique that measures free water in the small bowel using a strongly T2 weighted single-shot fast spin echo (RARE) sequence, which is the same sequence used for biliary and pancreatic ducts, showing fluid-containing hollow organs as bright areas in MRI images (Adkin et al., 1995, Schiller et al., 2005). However, some free water in the small intestine cannot be observed with this MRI sequence. This includes free water that is trapped in mucus or in food matrices.

Quantification of SBWC in response to the consumption of liquid and solid meal has been demonstrated in several studies involving both normal weight/healthy participants (Marciani et al., 2010, Marciani et al., 2012) and individuals with altered appetite as described in Khalaf et al. (2020b). For more comprehensive information readers are referred to Dellschaft et al. (2022).

2.6.7 <u>Neuronal signalling to food intake</u>

Recent advances in neuroimaging techniques have enabled new avenues to study food regulation and neuronal signalling that influences appetite and satiety in health and disease. Among these techniques, functional MRI (fMRI) and positron emission tomography (PET) are the two most frequently employed methods to study neuronal signalling responses to food intake and reward. fMRI is a non-invasive imaging method that is widely used to measure changes in blood flow and oxygenation in the brain. PET is used to assess metabolic activity in various brain regions through the injection of a radioactive tracer into the bloodstream, which allows tracking glucose or other substances uptake in the brain. There are different approaches to measure neural response to food intake. Due to the relevance of this thesis approaches used in fMRI are outlined below.

In a typical food-related fMRI study, brain activity is measured as individuals are exposed to visual or sensory (smell or taste) cues for a period of time, which is then followed by a rest period. This design is commonly referred to as "block design" and is form of task-fMRI. fMRI is highly effective in identifying brain regions responsible for food reward and pleasure (Batterham et al., 2007, Farooqi et al., 2007), as well as for examining how cues associated with food decision-making and dietary selections influence brain activity (Wang et al., 2016). Additionally, fMRI enables the evaluation of changes in brain activity before (fasted state) and after (fed state) meals, offering valuable insights into the regulation of appetite and feelings of fullness.

One way of assessing these changes is by measuring cerebral blood flow (CBF), typically by using arterial spin labelling (ASL) technique (Page et al., 2009, Lennerz et al., 2013, Eldeghaidy et al., 2016). CBF-fMRI studies enable the evaluation of the impact of physiological states by measuring the absolute CBF at different time points, which can then be compared with a baseline CBF measure. In addition, changes in resting state fMRI network can also be evaluated at different hunger and satiety levels (Liu et al., 2000, Smeets et al., 2005) or across different groups (for example NW vs. Obese).

2.6.8 **Questionnaires and surveys to assess eating behaviour data**

2.6.8.1 Binge eating scale (BES)

The BES is based on 16 questions developed by Gormally et al. (1982) to evaluate the severity of binge eating traits with evaluation of the affective, behavioural and cognitive manifestations. Binge eating refers to compulsive eating episodes which are composed mainly of two elements: loss of control of overeating, and eating large and extreme amounts of food in a short period (American Psychiatric Association, 2013). The questionnaire includes the following domains: eight questions that reflect behavioural symptoms (such as consuming big amounts of food or eating quickly) and eight questions on accompanying feelings and cognitions (such as the worry of not being able to stop eating). Responses for each item have a range between 0 to 3 where 0 indicates no serious symptoms of BE while 3 indicates severe BE symptoms (Escrivá-Martínez et al., 2019).

Total scores for this questionnaire ranged from 0 to 46. A score of ≤ 17 suggests the absence of binge eating disorders, 18 -26 moderate disorders of binge eating and ≥ 27 indicates severe disorders (Marcus et al., 1988). BES has high specificity and sensitivity to differentiate between normal and compulsive eaters in non-clinical and clinical populations (Freitas et al., 2006, Grupski et al., 2013, Duarte et al., 2015). This scale was used for studies conducted in this thesis (Chapters 6 and 7).

2.6.8.2 Control of eating questionnaire (CoEQ)

CoEQ is a questionnaire developed by Dalton et al. (2015) with 21-items to measure experiences during the past seven days using the following domains: sweet craving (items 3, 13, 14 and 15), craving control (items 9, 10, 11, 12 and 19), savory craving (items 4, 16, 17 and 18), and positive mood (items 5,6, 7 and 8) (Dalton et al., 2017). These items are evaluated using 100-mm visual analogue scales, with higher scores in the domains representing greater characteristics. Items 1 and 2 in the questionnaire measure general appetite feelings. Items 20 and 21 are measuring individual's control level over resisting a nominated, craved food item. These four items in the questionnaire are not included in the calculation of subscales.

2.6.8.3 Depression scale

Depression scale is commonly used to measure depression in clinical and research settings (Beck et al., 1961). It is a 21-item self-reported questionnaire that measure depressive symptoms according to Diagnostic and Statistical Manual for Mental Disorders (American Psychiatric Association, 2013). Items are added together to generate an overall score with a higher score indicating greater levels of depression.

2.6.8.4 Dutch eating behaviour questionnaire (DEBQ)

DEBQ is a 33-items questionnaire developed by Van Strien et al. (1986) that assess different types of eating attitudes that may contribute to emotional, restraint and external eating, and weight gain (Brunault et al., 2015). It is a gold standard tool to assess the cognitive, behavioural, and emotional dimensions of eating behaviour with good reliability and validity (Dutton and Dovey, 2016, Arhire et al., 2021, de Carvalho et al., 2023). The response ranges from 1 (never) to 5 (very often) for each item in the questionnaire. Higher scores indicate greater alignment with each of the eating behaviours.

2.6.8.5 *Eating attitude test (EAT)*

EAT is the most standardised measure of eating disorder symptoms and concerns. However, it does not give a specific diagnosis of an eating disorder. It is a 26-item test which examines three aspects: dietary (13 items that assess the avoidance of high caloric foods and concern with becoming slimmer), bulimia and food preoccupation (6 items that assess thoughts on food and bulimia), and oral control (7 items that assess the perceived pressure from others to gain weight and self-control eating) (Garner and Garfinkel, 1979, Garner et al., 1982). Six responses are presented for each item (except for item 26): always (3) score), usually (2 score), often (1 score), sometimes (0 score), rarely (0 score), or never (0 score). Item-26 has a different scoring system with zero scores for always, usually, and often and 1 score for sometimes, 2 scores for rarely and 3 scores for never. A large body of literature has been published on its application across a wide variety of age groups and cultures (Garfinkel and Newman, 2001). EAT possesses great psychometric properties including excellent sensitivity and specificity with high validity and test-retest reliability (Garner et al., 1982, Mintz and O'Halloran, 2000, Lee et al., 2002).

2.6.8.6 Food preference questionnaire (PrefQuest)

PrefQuest, a questionnaire developed by Deglaire et al. (2012) with four scales for evaluating recalled preferences for the tastes of sweet, salt, fat and sweet, and fat and salt in adults. It is a web-based questionnaire with 83 items in total, with subscales measuring a person's preference for sweetness (21 items), saltiness (11 items), and fattiness subscales (20 items refer to fattiness and sweetness and 31 items refer to fattiness and saltiness). Each item is represented on a 9-point rating scale, with higher scores indicating higher preference to a food item.

2.6.8.7 Three-factor eating questionnaire (TFEQ)

The TFEQ is a 51-item questionnaire developed by Stunkard and Messick (1985) that assesses eating behaviour across three dimensions: cognitive restraint of eating (21 items, factor 1), disinhibition (16 items, factor 2) and susceptibility to hunger (14 items, factor 3). Cognitive restraint is defined as a conscious control of eating to promote weight loss or control body weight (score ranging from 0 to 21). Disinhibition is the overconsumption of eating in response to different stimuli such as emotional stress, which is accompanied by a lack of food intake control (score ranging from 0 to 16). Susceptibility to hunger is defined as eating in response to hunger feelings and perceptions (score ranging from 0 to 14) (Löffler et al., 2015). This questionnaire has been validated for normal-weight people, people with obesity, the general adult population (Hyland et al., 1989), and different ethnic populations (Bardone-Cone and Boyd, 2007). All scales demonstrated adequate test–retest reliability (Laessle et al., 1989). All TFEQ items are assigned either 0 or 1 points. Higher scores in the domains represent greater characteristics.

2.6.8.8 Power of food scale (PFS)

The PFS is a 21-item scale developed by Cappelleri et al. (2009) with good validity and reliability that assesses the psychological impact of an environment with an abundancy of 'palatable' foods (Lowe et al., 2009, Andreeva et al., 2019, Torelli et al., 2022). The scale assesses appetite for palatable food but not consumption at three food proximity levels. First, food available level which assess general food related thoughts. Second, food physically present level which measures a person's desire for foods that are physically present in front of them. The last level is food tasted which assesses the desire for and enjoyment of food when first tasted.

2.6.8.9 Intake24 online dietary recall survey

This survey is an open source self-completed computerised system (https://intake24.co.uk/) designed and validated by Newcastle University (Intake24, 2023) based on multiple-pass 24-hour recall (Raper et al., 2004). It includes a database of more than 2500 foods linked to food composition codes (Roe et al., 2015). It was originally designed for participants aged between 11-24 years and subsequently extended for the general adult aged 11-88 years

(Rowland et al., 2018). The system allows participants to record all drink and food ingested with a series of pictures to help for estimating portion sizes. Participants also could verify their responses at each step. The system will ask some questions if some food items considered to be missing such as 'do you have butter on toast'. This method was used to obtain 3-days dietary intake as well as quantifying energy intake for the rest of study days for the Chapter 7.

To conclude, approaches to measuring food intake regulations, there are a variety of methods available to measure appetite, satiety, and food intake. In choosing the ones to be used in trials undertaken in this thesis, a combination of factors, including cost, invasiveness, and facilities available, were all taken into account.

3 MRI Theory and Image Processing

MRI is widely recognised as one of the most powerful imaging techniques for measuring the structural and functional aspects of the GI tract and brain, and their responses to food consumption and regulation of food intake. MRI was used extensively in the studies conducted within this thesis. The following section gives a brief description of the principles of MRI. For more details the reader is referred to McRobbie (2003).

3.1 Basic principles of MRI

MRI is a non-invasive imaging technique that generates detailed threedimensional detailed anatomical images of the internal structures of the body. It relies on the principles of nuclear magnetic resonance, which involves the interaction of certain atomic nuclei with strong magnetic fields and radiofrequency (RF) pulses (McRobbie, 2003). In MRI, hydrogen nuclei (protons), are primarily used because their abundance in the body and the highwater content of tissues. Protons also have a magnetic property referred to as "spin". These spins behave like tiny magnets.

MRI machines generate strong, static magnetic fields (B₀) using superconducting magnets. When a person is placed in the external magnetic field of the MRI machine, the protons "spin" around their axis and align themselves with the direction of the magnetic field (i.e., parallel to external magnetic field B₀, longitudinal plane), with a net magnetisation M₀, as shown in Figure 3.1. The protons also process, or spin, around the B₀ field at a characteristic frequency known as the "Larmor frequency". The Larmor frequency is directly proportional to the strength of the magnetic field and the gyromagnetic ratio of the ¹H for protons.



"Net Magnetization"

Figure 3.1. The process of protons in the absence of the magnetic field (the left side, protons randomly oriented), and when a strong magnetic field B_0 is applied (the right side, protons aligned parallel to the magnetic field). Figure reproduced from Caligari Conti (2016).

To manipulate these aligned protons, RF coils within the MRI machine emit short bursts of RF energy at "Lamour frequency". These RF pulses are applied at an angle relative to the magnetic field, typically 90 degrees, and are used to excite the protons (Figure 3.2), causing them to temporarily deviate from their aligned positions (i.e., to move away from the main magnetic field, B₀). Because the RF pulse matches the resonant frequency of the protons in the magnetic field, it causes the protons to flip into the transverse plane (My). This phenomenon is known as "nuclear magnetic resonance". When the RF pulse is switched off, the excited protons gradually realign themselves to their original aligned state, parallel to the main field B₀ (Figure 3.2). During the return of the proton's equilibrium alignment, protons emit RF signals.

The process of returning to equilibrium is termed relaxation, and the process involves two relaxation times. T1 also known as spin-lattice relaxation or longitudinal relaxation time, and it is the time taken for the protons' longitudinal magnetisation (Mz) to recover to its equilibrium value (M₀). The other relaxation time is known as T2, or spin-spin relaxation, also known as transverse magnetisation. This is the time takes for the protons' transverse magnetisation (My) to decay. As the protons relax and return to equilibrium, they emit a signal known as Free Induction Decay (FID), which contains information about the characteristics of the tissue being imaged, such as its proton density, T1 and T2 relaxation times, and other tissue properties. The FID signal is detected by the MRI machine's receiver coil. The detected signal is then processed and used to create the final MRI image. Figure 3.2 gives an illustration of the excitation and relaxation process of the protons.



Figure 3.2. The diagram illustrating the main concept of MRI. Upper section: The magnetisation (M_0) is rotated to the transverse plane by the radiofrequency (RF) pulse. Lower section: Following the switching off the RF pulse, the protons slowly reach equilibrium, and emit a signal known as Free Induction Decay (FID) that is detected by the MRI machine's receiver coil and contains information about the characteristics of the tissue being imaged. Figure reproduced from Dulińska-Litewka et al. (2019).

3.2 Image formation

Gradient coils within the MRI machine generate additional, spatially varying magnetic fields. These gradient fields are used to encode spatial information into the MR signals. There are three types of gradient fields: slice selection (z-axis), phase encoding (y-axis), and frequency encoding (x-axis).

Slice selection gradient (Gz) is applied along the z-axis (the direction of the main magnetic field, B_0). It creates variations in the magnetic field strength along the z-axis and is used to select a specific slice or section of the body for imaging. By applying Gz, MRI can capture images of individual slices within the body. *Phase encoding gradient* (Gy) is applied along the y-axis (perpendicular to both the z-axis and the main magnetic field B_0), and it creates variations in the magnetic field to encode spatial information along the y-axis. Gy determines the position of data in the phase-encoding gradient (Gx) creates variations in the magnetic field strength along the x-axis and is used to encode spatial information along the spatial information along this axis. It determines the frequency of the signals collected and helps create the two-dimensional image (along with the Gy).

The coordinated application of these gradient coils during the MRI scan allows the collection of data from different spatial positions within the body. By changing the strengths and timing of the gradient fields, the MRI machine can encode the spatial location of signals and create images with specific contrasts and spatial resolutions.





Figure 3.3. Representation of gradient fields, applied in MRI scanner, with illustration of the direction of each gradient: (a) slice selection (G_z), (b) phase encoding (G_x), and (c) frequency encoding (G_y). Figure adapted from IMAIOS (2023).

3.3 T1-weighted and T2-weighted contrasts

Different tissues in the body have varying relaxation times (T1 and T2), resulting in variations in signal intensity and contrast in the MRI images. These differences in contrast are essential for identifying and distinguishing different tissues. Figure 3.4 illustrates the variation in T1 and T2 relaxation times across different tissue type (fat, muscles, and fluid). Various MRI pulse sequences are used to highlight specific tissue properties and create different types of images. In T1-weighted contrast, tissues with longer T1 times, such as fat, appear bright, while tissues with shorter T1 times, like muscle, appear darker. In T2-weighted contrast, tissues with longer T2 times, such as cerebrospinal fluid (CSF), appear bright, while tissues with shorter T2 times, such as bone, appear darker. Figure 3.5 show an example of abdominal image illustrating the difference appearance of fat and water in T1-weight and T2-weights MRI scans.



Figure 3.4. T1 and T2 relaxation times for fat, fluid, and muscle. Figure adapted from Ridgway (2010).



Figure 3.5. Examples of T1 and T2 weighted scans illustrating abdominal MRI image of T2-weighted (left side) and T1-weighted contrast (right side). Figure adapted from Santos et al. (2017).

Contrast enhancement of MRI images strongly depends on echo time (TE) and repetition time (TR). TE controls the timing of signal acquisition after the application of the RF pulse. It is adjusted in MRI sequences to emphasise specific tissue properties and create contrast in the resulting images. TR is the time between the applications of the RF excitation pulse to the application of the next pulse. It also influences the appearance of MRI images and is tailored to the tissue to be examined. Adjusting the TE and TR in MRI sequences can influence image/tissue contrast. Longer TE values enhance T2 contrast, while shorter TE values emphasize T1 contrast. Longer TR values can increase T1 contrast.

In this thesis, MRI was employed extensively to investigate and understand the GI responses to food consumption and the regulation of food intake in normalweight participants and individuals with obesity. This is demonstrated in Chapter 5 and Chapter 6. Chapter 5 assesses the GI responses to a standard pasta meal, while Chapter 6 uses advances in MRI to assess food intake to different macronutrients.

4 Mapping brain activity of gut-brain signalling to appetite and satiety in people with and without obesity: A systematic review and functional neuroimaging meta-analysis

Understanding how neurohormonal gut-brain signalling regulates appetite and satiety is vital for the development of therapies for obesity and altered eating behavior. However, reported brain areas associated with appetite or satiety regulators show inconsistency across functional neuroimaging studies. The work in this chapter aimed to systematically assess the convergence of brain regions modulated by appetite and satiety regulators, and to generate quantitative brain activation maps of neurohormonal gut-brain signalling in normal-weight adults that can be used to define alterations with eating behaviour.

The work detailed in this chapter was published in Neuroscience & Biobehavioral Reviews. It was also presented as an oral presentation at the 46th British Feeding and Drinking Group, and as a poster of distinction at the 43rd ESPEN Congress.

Althubeati, S., Avery, A., Tench, C.R., Lobo, D.N., Salter, A. and Eldeghaidy, S., 2022. Mapping brain activity of gut-brain signaling to appetite and satiety in healthy adults: A systematic review and functional neuroimaging meta-analysis. Neuroscience & Biobehavioral Reviews, 136, p.104603.

The British Feeding and Drinking Group (BFDG) 46th Annual Meeting: Althubeati, S., Avery, A., Tench, C.R., Lobo, D.N., Salter, A. and Eldeghaidy, S., 2022. Functional neuroimaging meta-analysis to map brain activity of gut-brain signalling to appetite and satiety in healthy adults. Appetite, 179, p.106213.

The 43rd ESPEN Congress: Althubeati, S., Avery, A., Lobo, D.N., Salter, A. and Eldeghaidy, S., 2021. A neuroimaging meta-analysis to identify brain areas associated with satiety and appetite regulators. Clinical Nutrition ESPEN, 46, p.S582.

4.1 Introduction

Recent advances in non-invasive neuroimaging techniques have allowed the study of neurohormonal gut-brain signalling pathways that modulate appetite in health and disease (Gibson et al., 2010). A number of studies report attenuation in appetite-related brain areas including the insula, amygdala, OFC, ventral striatum, ACC, caudate, parahippocampal cortex, thalamus, and hypothalamus, when humans are fed, compared with when they are hungry (Jakobsdottir S, 2012, Sun et al., 2014). In contrast, studies investigating endogenously released appetite/satiety regulators, following food ingestion, or exogenously administered hormones with brain responses conducted on NW adults show inconsistency in the reported brain areas. For instance, some studies demonstrate increased activation in the insula with increasing concentrations of satiety regulators (Batterham et al., 2007, Schilling et al., 2014) while others found decreased activation in the insula with decreasing concentrations of satiety regulators (Spetter et al., 2014). There are different reasons for these discrepancies, including different designs (endogenously released vs. exogenous administration) or different paradigm/stimulation (no-task 'rest', visual-task "food images", taste-task) used during brain imaging examination. In people living with obesity, functional neuroimaging studies have shown that the brain activity in these people is significantly different from that in NW adults in several brain regions. Brain regions implicated in food reward responses have greater activation in these people paired with hypoactivity in areas associated with homeostatic satiety (Rothemund et al., 2007, Stoeckel et al., 2008, Szalay et al., 2012). Hence, separating brain data from participants living with obesity from NW participants is essential for an accurate understanding of brain activation associated with appetite and satiety processing.

By pooling data from published work on the interplay between appetite and satiety regulators and the brain in modulating appetite, a more accurate picture of regional brain activation associated with appetite and satiety processing can be established. Coordinate-based neuroimaging meta-analyses including activation likelihood estimation (ALE) methodology (Turkeltaub et al., 2002) and Analysis of Brain Coordinates (ABC) (Tench et al., 2021) allow the

identification of consistent brain activations across studies. These techniques use coordinates in standard anatomical space reported by neuroimaging studies to assess the agreement, or overlap, in activation patterns and infer a quantitative brain map of the overlapped regions (Muller et al., 2018, Tench et al., 2021). Thus, by using these methods a quantitative brain activation map of the neurohormonal gut-brain interactions could be generated.

At the time of writing this thesis, there was only one published systematic review that investigated the effects of appetite/satiety regulators on brain regions involved in appetite and satiety (Zanchi et al., 2017). However, neuroimaging data in this previous review were combined from adults with normal-weight and those with obesity and did not conduct functional neuroimaging meta-analysis to quantitively determine the concurrence/overlap of brain areas activated in response to appetite and satiety regulators across studies.

4.2 Aims and Hypothesis

Primary aim

To provide a comprehensive analysis of the functional neuroimaging literature on brain areas associated with changes in appetite and satiety regulators in NW and Obese adults.

Secondary aim

To produce a quantitative brain activation map of the neurohormonal gut-brain interactions, using coordinate-based neuroimaging meta-analysis.

Hypothesis

The hypothalamus, insula and caudate nucleus are key brain areas regulating appetite and satiety and are positively associated by appetite regulators and negatively by satiety regulators in NW adults. Such regulation is disrupted in adults with obesity.

4.3 Material and methods

4.3.1 Systematic review of the literature

A comprehensive search was carried out in the MEDLINE, EMBASE and Cochrane Central Register of Controlled Trials (CENTRAL) databases between November 2019 and January 2021 to identify relevant studies using keywords from functional neuroimaging techniques and appetite and satiety responses including terms related to satiety and/or appetite regulators. In addition, a manual searching process was used to find relevant studies in the reference lists of all included studies.

4.3.2 <u>Search strategy</u>

The full search strategy is described below in Table 4.1. This systematic review was conducted in accordance with the Preferred Reporting Items for Systematic Reviews and Meta-Analyses (PRISMA) guidelines (Liberati et al., 2009). The protocol was registered on the International Prospective Register of Systematic Reviews (PROSPERO) (https://www.crd.york.ac.uk/prospero/) with registration number CRD42020223921. Searches were restricted to human studies published in the English language but not restricted to publication dates. After duplicates were removed, all records were screened for title and abstract. The remaining publications were then reviewed independently for eligibility based on full texts by three authors using the set criteria. The eligibility criteria were based on the PICO (Population-Intervention-Comparator-Outcomes) model, summarised in Table 4.2.

Table 4.1. Search Strategy used in the systematic review.

- 1. "gut peptide" [MeSH terms]
- 2. "gut peptides" [MeSH terms]
- 3. "gut hormone" or "gastrointestinal hormone" [MeSH terms]
- 4. " gut hormones" or "gastrointestinal hormones" [MeSH terms]
- 5. "peptide YY" [MeSH terms]
- 6. "PYY" [MeSH terms]
- 7. "ghrelin" [MeSH terms]
- 8. "cholecystokinin" [MeSH terms]
- 9. "CCK" [MeSH terms]

10. "glucagon like peptide 1" [MeSH terms]

11. "GLP-1" [MeSH terms]

12. "satiety response" or "satiety" [MesSH terms]

13. "appetite" [MeSH terms]

14. "brain" [MeSH terms]

15. "imaging" [MeSH terms]

16. "neuroimaging" or "functional neuroimaging" or "neuroimaging" [MeSH terms]

17. "MRI" or "nuclear magnetic resonance imaging" [MeSH terms]

18. "fMRI" or "functional magnetic resonance imaging" [MeSH terms]

19. "PET" or "positron emission tomography" [MeSH terms]

20. "food" [MeSH terms]

21. "nutrient" [MeSH terms]

22. "nutrients" [MeSH terms]

23. "food intake" [MeSH terms]

24. "caloric intake" or "nutrient intake" [MeSH terms]

25. "meal" [MeSH terms]

26. "meals" [MeSH terms]

27. "meal intake" [MeSH terms]

28. "drink" [MeSH terms]

29. "drinks" [MeSH terms]

30. #1 or #2 or #3 or #4 or #5 or #6 or #7 or #8 or #9 or #10 or #11

31. #12 or #13

32. #14 or #15 or #16 or #17 or #18 or #19

33. #20 or #21 or #22 or #23 or #24 or #25 or #26 or #27 or #28 or #29

34. #30 and #31 and #32 and #33

35. #30 and #31 and #32

Inclusion	Population	• Normal-weight (18.5 to 24.9 kg/m ²) or adult humans with			
criteria		obesity (BMI \ge 30 kg/m ²) between 18-65 years old with no			
		medications that would influence appetite or metabolism			
	Intervention	• Acute macronutrients interventions (carbohydrate, protein,			
		or fat) consumed as a drink or a meal or			
		• Exogenous infusion of appetite/satiety regulators			
		• No restrictions were applied on the amount of			
		macronutrients given, the level of hormone infusion, the			
		number of hours fasted, the consumption/infusion and the			
		route of macronutrient ingestion (oral or gastric) or gut			
		hormone infusion (bolus, intravenous or subcutaneous			
		injection)			
	Comparator	• No specific comparators with controls such as water,			
		placebo, saline or fasting included. Most studies are			
		"before and after" studies where the participants serve as			
		their own controls			
	Outcomes	• Primary outcome: concurrence of brain regions modulated in response to appetite and satiety regulators in NW and Obese			
		• Secondary outcome: quantitative brain-activation maps			
		generated from coordinate based meta-analyses to assess the			
		concurrence of brain regions modulated in response to			
		appetite and satiety regulators			
	Study	Controlled trials, randomized controlled trials, randomized			
	design	cross-over design trials, and cohort studies			
		• Studies that involved participants with gastrointestinal,			
Exclusion		endocrine, and neurological diseases, or adolescents			
criteria		Publications with no direct correlation analysis performed			
		between brain responses and satiety/appetite regulators or with			
		long-intervention studies			
		• In-vitro studies			
		• Reviews			

Table 4.2. Eligibility criteria based on the PICO (Population-Intervention-Comparator-Outcomes) model.

4.3.3 <u>Risk-of-Bias Assessment</u>

The quality of the included papers was assessed for potential risk of bias using the Cochrane collaboration to assess the risk of bias in randomised cross-over and randomised controlled trials (ROBINS-I) (Ding et al., 2015, Higgins et al., 2019).

4.3.4 Data extraction

For each study, the following information was extracted: authors, year of publication, total number of participants, participant details [mean age, sex and body mass index (BMI), time of first brain imaging scan after treatment administration, intervention, administration method (e.g. oral, gastric or intravenous cannula), assessed appetite/satiety regulators, neuroimaging modality and brain stimulation method (e.g. gustatory, visual) and correlation results between brain areas and appetite and satiety regulators. Extracted data were grouped into 1) brain areas correlated positively and/or 2) correlated negatively with appetite regulators, 3) brain areas correlated positively and/or 4) correlated negatively with satiety regulators. In the appetite analysis, data were analysed during the fasting state or for contrast fasted>fed. In the satiety state analysis, data were derived from a direct contrast between fed state versus fasted/hunger state (fed>fasted) or data assessed postprandially within 1.5 hours following the last consumption. Brain areas from each of the sub-grouped data were then pooled and common brain areas across studies were evaluated. To illustrate the concurrence of brain areas generated from the systematic review, anatomically defined masks for overlapped brain areas were generated using WFU PickAtlas toolbox (Maldjian et al., 2003) in SPM software (https://www.fil.ion.ucl.ac.uk/spm/software/spm12/). The generated masks were displayed using the MRIcroGL software (Rorden et al., 2007) and overlaid on brain template in Montreal Neurological Institute (MNI) space.

4.3.5 <u>Coordinate-based neuroimaging meta-analysis</u>

Neuroimaging studies included in the meta-analysis were pooled from those included in the systematic review. Studies that did not report coordinates for brain activations in response to appetite/satiety regulators in the article or supplementary material were excluded from the meta-analysis. The recently suggested standard protocol for neuroimaging meta-analysis by Eickhoff et al. (2016) was followed and therefore included only neuroimaging studies that
reported brain activation using whole-brain voxel wise analyses (Turkeltaub et al., 2012). In addition, seed-based functional connectivity analysis in rs-fMRI studies were excluded from the meta-analysis, as they usually focus on particular areas in the brain. Coordinates of brain regions that are directly correlated with satiety and appetite regulators were manually extracted. Extracted coordinates were checked and when inconsistencies between the coordinates reported in the original study were identified coordinates were rechecked and corrected. Studies that reported coordinates in Talaraich space (Talairach and Tournoux, 1988) were converted to the standard space of the MNI (Evans et al., 1993) using the icbm2taal algorithm implemented in the Ginger ALE toolbox (Laird et al., 2010).

There are multiple algorithms for performing coordinate-based neuroimaging meta-analysis, each has different empirical parameters and assumptions, and each can produce different results conditional on the assumptions. Therefore, to obtain robust results of brain areas associated with appetite and satiety regulators, two different tools were employed in the current neuroimaging meta-analysis: the ALE and ABC meta-analysis methods. The next section describes in the ALE and ABC analyses conducted in this chapter.

4.3.5.1 Activation likelihood estimation meta-analysis

The ALE-approach is the most popular method of performing neuroimaging meta-analysis. The algorithm takes into account the number of participants in each study to apply a relevant smoothing, resulting in a higher specificity of the actual overlap between studies (Eickhoff et al., 2009). However, in order to produce results that are not overly representative of single studies, it is recommended that at least 17–20 experiments should be included in the analysis. In addition, the ALE algorithm does not allow assessment of the effect sign associated with the coordinates when a decrease/increase brain activity are combined in a single analysis. This recommendation is based on the finding that, from this size on, the average contribution of the most dominant experiment to any above-threshold cluster is less than half, and the average contribution of the two most dominant experiments to any above-threshold cluster is less than 80% when controlling for multiple comparisons using cluster-level family wise error correction (Eickhoff et al., 2016). ALE meta-analyses were performed using

Ginger ALE version 3.0.2 (http://www.brainmap.org/ale). ALE analysis uses the reported activation peaks from the individual studies as a three-dimensional Gaussian probability distribution (kernel) centered at the given coordinates to create a modeled activation (MA) map for each study. Individual MA-maps are then combined to calculate statistical ALE maps and ALE values for each cluster. These calculations are confined to a grey matter mask provided by the Ginger ALE software. The ALE maps indicate areas of the brain where convergence between activation foci is greater than would be expected by chance (i.e., a null distribution of clusters). The recommendations of Eickhoff et al. (2016) for all analyses was followed. The statistical significance threshold of ALE maps was assessed and corrected for multiple comparisons by employing a cluster-level family-wise error (FWE) at P <0.05, following an initial cluster forming threshold of uncorrected P <0.001 Eickhoff et al. (2016). The P-value was calculated for each voxel based on probabilities of reaching an ALE value that differed from that of the corresponding voxel on a nulldistribution map, via random permutation. Five-thousand permutations were used to generate the P-values (Laird et al., 2010).

Discriminating between close and distant coordinates was crucial for accurately assessing spatial convergence across studies. A voxel-wise approach was utilised where each study's foci were modelled as Gaussian probability distributions centred on the reported coordinates. The size of the Gaussian kernel was determined based on empirical assessments of spatial uncertainty in neuroimaging, typically ranging between 4- and 20-mm full width at half maximum (FWHM) (smoothing data for each study are shown in Table 4.3). This modelling allows for the assessment of the likelihood that each voxel is activated, accounting for both the precision of coordinate reporting and the spatial resolution of the studies. Overlapping probabilities from these Gaussian kernels were then accumulated across experiments to create a statistical ALE map (Eickhoff et al., 2009, Turkeltaub et al., 2012, Eickhoff et al., 2016). Thresholding this map based on cluster size correction and permutation testing, as described above, ensured that only significant clusters of spatially convergent activity were considered, effectively discriminating between closely located foci and more dispersed activations

The generated meta-analysis maps from the ALE methods were displayed using the MRIcroGL software (Rorden et al., 2007) and overlaid on brain template in MNI space.

4.3.5.2 Analysis of Brain Coordinates

ABC requires a minimum of 5 studies and does not take any account of the sample size in each study. A further difference between the two algorithms is in the thresholding for statistical significance, where ALE uses a cluster level family wise error rate method and ABC directly relates the threshold to the aim of detecting replicated results. ABC methodology (Tench et al., 2021) was performed using the ABC toolbox implemented in the NeuRoi image analysis software

(www.nottingham.ac.uk/research/groups/clinicalneurology/neuroi.aspx). The algorithm of this recently developed model-based method uses the density of coordinates from independent studies as its statistic and requires only the human grey matter volume (one parameter). Statistical thresholding is performed by requiring a minimum proportion of the studies contributing to a cluster and is generally more conservative than false discovery rate (FDR<0.05). Importantly, this method, in contrast to the ALE-approach, does not require the empirical choice of Gaussian smoothing kernel to extrapolate coordinates to voxel-wise activation maps or the randomization of the coordinates in the empirical space to define the statistical threshold.

4.4 **Results**

4.4.1 <u>Systematic review</u>

4.4.1.1 Selection and inclusion of studies

Of the 1390 studies identified in the initial search, 81 were selected for full text assessment (Figure 4.1). A total of 31 eligible studies (25 studies with NW participants and 6 studies with Obese participants) were included. Characteristics of the included studies are summarised in Table 4.3. The quality assessment of each paper is shown below in Table 4.4.



Figure 4.1. PRISMA Diagram

4.4.1.2 Characteristics of included studies

Of the 31 studies included, 17 investigated the effect of endogenously released appetite and satiety regulators and 11 investigated the effect of exogenously administered regulators on brain responses. Twenty-two studies used fMRI, of which eight used food picture task-fMRI (Malik et al., 2008, De Silva et al., 2011, Page et al., 2011, Jakobsdottir et al., 2012, Kroemer et al., 2013, Goldstone et al., 2014, Heni et al., 2015, Dorton et al., 2017), five studies used taste stimuli (Liu et al., 2000, Li et al., 2012, Spetter et al., 2014, Sun et al., 2014, Eldeghaidy et al., 2016), four studies assessed neurological responses across a time course "physiological fMRI design"(Batterham et al., 2007, Lassman et al., 2010, Jones et al., 2012b, Little et al., 2014), four studies used resting state fMRI (Page et al., 2013, Wolnerhanssen et al., 2015, Zhang et al., 2015, Al-Zubaidi et al.,

2019), three studies used ASL (Page et al., 2009, Lennerz et al., 2013, Schilling et al., 2014) and three studies used PET imaging technique (Tataranni et al., 1999, Gautier et al., 2000, Pannacciulli et al., 2007).

Across the included studies, brain responses for the hungry state were assessed following fasting that ranged between 4 and 14 hours, whereas for the satiety state brain responses were assessed within 1.5 hours postprandially. Seven studies administered standard meals with different amounts of protein, fat and fibre, containing ingredients such as soya bean, beef or milkshake (Tataranni et al., 1999, Gautier et al., 2000, Pannacciulli et al., 2007, Jakobsdottir et al., 2012, Spetter et al., 2014, Sun et al., 2014, Al-Zubaidi et al., 2019). Eight studies administered target nutrients such as whey protein solution, glucose drink, soybean oil emulsion or flavoured fat emulsion samples (Liu et al., 2000, Li et al., 2012, Kroemer et al., 2013, Page et al., 2013, Heni et al., 2015, Wolnerhanssen et al., 2015, Eldeghaidy et al., 2016, Dorton et al., 2017). Seventeen studies reported that nutrients were administered orally (Tataranni et al., 1999, Gautier et al., 2000, Liu et al., 2000, Pannacciulli et al., 2007, De Silva et al., 2011, Jakobsdottir et al., 2012, Li et al., 2012, Kroemer et al., 2013, Lennerz et al., 2013, Page et al., 2013, Spetter et al., 2014, Sun et al., 2014, Heni et al., 2015, Zhang et al., 2015, Eldeghaidy et al., 2016, Dorton et al., 2017, Al-Zubaidi et al., 2019), while three studies administered nutrients via the intragastric route (Little et al., 2014, Spetter et al., 2014, Wolnerhanssen et al., 2015). Spetter et al. (2014) reported that they administered nutrients by both the nasogastric tube and the oral routes. Eleven studies administered exogenous appetite and satiety regulators including PYY, GLP-1, ghrelin, insulin, glucose and CCK by intravenous (Batterham et al., 2007, Malik et al., 2008, Page et al., 2009, De Silva et al., 2011, Page et al., 2011, Jones et al., 2012b, Goldstone et al., 2014, van Bloemendaal et al., 2014), intranasal (Schilling et al., 2014) or intra-gastric infusion (Lassman et al., 2010, Little et al., 2014). Details of the included studies are provided in Table 4.3.

Reference	Subjects	Mean age (years), gender & BMI (kg/m ²)	Time of brain imaging & type of brain analysis	Intervention	Adminis tration	Hormone investigate d	Neuroimaging modality & paradigm	Smoothing and threshold level	Results
						Endogenously r	eleased regulators stua	lies	
Al- Zubaidi et al. (2019)	n= 24	- Age: 24.3±1.3 - Sex: All M - BMI: 23.4 ±1.4	 After 20 minutes of the nutrient intake Whole brain 	 300 ml of glucose (75 g) ingestion Fasted state 	Orally	Glucose Insulin	rs-fMRI	-6 mm FWHM -FDR at P = 0.05	After glucose ingestion relative to fasting (hunger > satiety): - insulin levels → superior frontal gyrus ↓, posterior insula ↓ - glucose → fusiform gyrus ↑
Dorton et al. (2017)	n= 22	- Age: 21.2 ± 2.1 - Sex: 10 M & 12 F - BMI: 22.6 ± 1.9	 After 20 minutes of the nutrient intake ROI [ventral striatum (nucleus accumbens) and bilateral 	 300 ml of glucose (75 g) ingestion 300 ml of water ingestion 	Orally	GLP-1 PYY	Task-fMRI (food picture paradigm)	-5 mm FWHM -P < 0.05, corrected for multiple comparisons	After glucose intake: - GLP-1 \rightarrow the dorsal striatal \downarrow

Table 4.3. Characteristics and main results of the included studies

			dorsal						
			Sulatuill						
			(caudate/put						
			amen)]						
Eldeghai	n= 17	- Age: 25 ± 2	- Immediately	- Two emulsion	Orally	ССК	Task-fMRI (taste	- 8 mm FWHM	Responses to the CS and FS after the
dy et al.		- Sex; 11 M &	W/h ala huain	stimuli;			stimuli paradigm)	D < 0.05	high fat meal:
(2010) *		6 F	- whole brain	stimulus (FS)				-P < 0.05,	- $CCK \rightarrow primary somatosensory$
		- BMI: 22.4 ±		& flavoured				corrected	φ gyrus \perp middle and posterior insula
		0.8		not fat control				conceted	L temporal gyrus L thalamus L.
		0.8		stimulus (CS)					cerebellum \downarrow , operculum \downarrow
				following:					
				■ 250 ml					
				of high					
				fat					
				drink/lo					
				ad					
				(22%) fat)					
				■ 250 ml					
				of					
				water					
				load					
Gautier	n= 22	- Age: 35 ±8	- After 25	- Liquid	Orally	Insulin	PET	-Smoothing level is	After the liquid meal ingestion in NW
et al.	(11 NW	(NW), 27±5	minutes of	formula		GLP-1		not mentioned	participants:
(2000) *	& 11	(Obese)	the meal	meal (1.5		Leptin			
	Obese)	- Sex: All M	intake	kcal/ml		FFA			

- BMI: ≤25			Ensure-Plus:		-P < 0.005,	- insulin \rightarrow posterior OFC \downarrow ,
	(NW), ≥ 35	- Whole	15% protein,		uncorrected for	hippocampus/ parahippocampus \downarrow ,
	(Obese)	brain+ ROI	53% carb &		multiple	putamen \downarrow , thalamus \downarrow , precuneus \uparrow
		(hypothalam	32% fat)		comparisons	- FFA \rightarrow DLPFC \uparrow
		us,				After the liquid meal ingestion in Obese
		thalamus,				participants:
		DLPFC,				- insulin \rightarrow posterior OFC \downarrow ,
		anterior				hippocampus/ parahippocampus \downarrow ,
		prefrontal				precuneus \downarrow , putamen \downarrow , thalamus \downarrow .
		cortex,				
		ACC,				
		insular				
		cortex,				
		posterior				
		orbitofrontal				
		cortex,				
		hippocampu				
		s/parahippoc				
		ampal				
		gyrus,				
		caudate				
		ventricle,				
		precuneus,				
		putamen,				
		parietotemp				
		oral cortex,				
		occipital				
		cortex,				

			cerebellum						
			& midbrain)						
Heni et	n=12	- Age: 23±2	- After 30	- 300 ml of	Orally	GLP-1	Task-fMRI (food	-6 mm FWHM	After glucose intake:
al.		- Sex: 6 M & 6	minutes of	glucose (75			picture paradigm)		- insulin \rightarrow OFC \downarrow
(2015) *		F	the meal	g) ingestion				-P<0.05, corrected	
		- BMI: 21.2 ±	intake	- 300 ml of				for multiple	
		1.1		water				comparisons	
			- Whole brain	ingestion					
Jakobsdo	n= 15	- Age: 23.4 ±	- One hour of	- Standard	Orally	Glucose	Task-fMRI (food	-6 mm FWHM	After satiation with the standard meal:
ttir et		3.5	the meal	meal		Insulin	picture paradigm)		- leptin \rightarrow hippocampus \downarrow , insula \downarrow ,
al.		- Sex: All M	intake	consisted of		Ghrelin		-FDR at P<0.05	temporal lobe bilaterally \downarrow , frontal gyrus
(2012) *		- BMI: 22.4 ± 2		1600 kcal,		TAG			\downarrow
			- Whole brain	15.8 %		Leptin			
				protein,					
				44.4%					
				carbohydrate					
				and 39.8%					
				fat					
Kroemer	n=26	- Age: 24.4 ±	- After 5	- 300 ml of	Orally	Ghrelin	Task-fMRI (food	-8 mm FWHM	During fasting:
et al.		3.4	minutes of	glucose (75			picture paradigm)		- ghrelin \rightarrow middle frontal gyrus \uparrow ,
(2013) *		- Sex: 13 M &	the meal	g) ingestion				-Whole brain	midbrain [↑] , superior/medial frontal
		13 F	intake					uncorrected	gyrus \uparrow , inferior frontal gyrus \uparrow , medial
		- BMI: 21.1 ± 2						uncorrected	occipital/temporal gyrus ↑,
			- Whole brain					P < 0.001/ROIs	hypothalamus \uparrow , subthalamic nucleus \uparrow ,
								FWE correction	fusiform gyrus \uparrow , thalamus \uparrow , superior
									occipital gyrus , inferior frontal gyrus \uparrow ,
									middle frontal gyrus ↑, pallidum,

									amygdala ↑, inferior frontal gyrus,
									caudate body ↑, inferior temporal g.,
									fusiform gyrus ↑, middle/superior frontal
									gyrus ↑, thalamus (anterior nucleus) ↑,
									medial/superior frontal gyrus, <i>fanterior</i>
									cingulate ↑, postcentral (supramarginal
									gyrus & rolandic operculum ↑)
Lennerz	n=12	- Age: 29.1	- Four hours	- Low	Orally	Glucose	rCBF/ASL	-8 mm FWHM	High GM vs low GM:
et al.		- Sex: All M	of the meal	glycemic		Insulin			- glucose and insulin \rightarrow nucleus
(2013)		- BMI: 32.9	intake	meal (37%				-Whole	accumbens ↑
				of predictive				brain/Bonferroni	
			- Whole brain	glucose)				corrected P < 0.002	
			+ ROIs	- High					
			(ventral	glycemic					
			striatum,	meal (84%					
			hypothalam	of predictive					
			us,	glucose)					
			midbrain)	- Both meals:					
				500 kcal, ~					
				68.9 g					
				carbohydrate					
				,~13 g fat,					
				~18 g					
				protein.					
Li et al.	n= 14	- Age: 23	- After 6	- 300 ml of	Orally	GLP-1	Task-fMRI (taste	-8 mm FWHM	After whey protein ingestion:
(2012) *		- Sex: All M	minutes of	whey	-	Ghrelin	stimuli paradigm)		- GLP-1: lateral orbito-frontal cortex↓
		- BMI: 21.2	meal	protein (257		Glucose			- insulin: caudate↓
			ingestion			Insulin			- CCK; thalamus \downarrow

[с /Г.)		CCV		$\mathbf{D} < 0.05$ some stad	abualing annyadala 🕇
				g/L)		CCK		-P < 0.05, corrected	- gnrein: amygdala
			- ROIs	ingestion				with Monte Carlo	After glucose ingestion:
			(thalamus,	- 300 ml of				simulations	 insulin: thalamus↓, middle insula↓,
			hypothalam	soybean					amygdala↓, lateral OFC ↓
			us, insula,	emulsion					- glucose: thalamus↓
			parahippoca	(11 g/L)					- CCK: caudate ↓
			mpal/hippoc	- 300 ml of					- GLP-1: Latera OFC ↓, middle insula↓
			ampal	glucose (250					 ghrelin: middle insula ↑, later OFC ↑
			cortex,	g/L)					After fat ingestion:
			putamen,	ingestion					- CCK: caudate \downarrow , thalamus \downarrow
			caudate	- Water: 300					 ghrelin: amygdala ↑, middle insula ↑,
			OFC &	ml					lateral OFC ↑
			amygdala)						
Liu et al.	n= 21	- Age: 34 ± 3	- After 10	- 296 ml of	Orally	Insulin	rs-fMRI	Smoothing and	- Fasting insulin \rightarrow hypothalamus \downarrow ,
(2000)		- Sex: 11 M &	minutes of	dextrose (75				threshold levels are	somatosensory cortex \downarrow , SMA \downarrow ,
		10 F	meal intake	g)				not mentioned	cerebellum \downarrow , anterior cingulate \downarrow , OFC \downarrow
		- BMI: not		- 300 ml of					
		available	- Whole brain	distilled					
		available		water					
Page et	n= 20	- Age: 31 ± 7	- After 60	- 300 ml of	Orally	GLP-1	rs-fMRI & fMRI-	-6 mm FWHM	After glucose ingestion:
al.		- Sex: 10 M &	minutes of	glucose (75		PYY	ASL		- insulin \rightarrow caudate \downarrow , putamen \downarrow
(2013) *		10 F	meal intake	g) ingestion		Ghrelin		-FWE whole- brain	
		- BMI: 22± 2.5		- 300 ml of				correction of	
			- Whole	fructose (75				P<0.05	
			brain+ ROI	g) ingestion					
			(hypothalam	<i>U</i> , <i>U</i>					
			us)						
	1	1	,			1	1	1	

Pannacci	n= 42	- Age: 31 ± 8	- After 25	- Standard	Orally	GLP-1	PET	-15 mm FWHM	After the liquid meal ingestion:
ulli et		- Sex: 22 M &	minutes of	liquid					- GLP-1 \rightarrow hypothalamus \uparrow , inferior
al.		20 F	the meal	formula				$-P \le 0.001$,	frontal gyrus \uparrow , middle frontal gyrus \uparrow ,
(2007) *		- BMI: 31 ± 9	intake	meal (1.5				uncorrected for	
				kcal/ml				multiple	
			- Whole brain	Ensure-plus,				comparisons	
				15% protein,					
				53%					
				carbohydrate					
				and 32% fat)					
Spetter et	n= 14	- Age: 24.6±	- After 5	- Oral	Orally &	Glucose	Task-fMRI (taste	-8 mm FWHM	During nasogastric infusion of chocolate
al.		3.8	minutes of	chocolate	nasogastri	Insulin	stimuli paradigm)		milk:
(2014) *		- Sex: All M	the meal	milk	c tube	Ghrelin		-FWE whole- brain	- insulin \rightarrow middle and posterior Insula \downarrow ,
		- BMI: 22.3±	intake	- Nasogastric		CCK		correction of	putamen↑
		1.6		chocolate				P<0.05	
			- Whole	milk					
			brain+ ROI	infusion					
			(hippocamp	per 100ml;					
			us insula,	84.6 kcal,					
			amygdala,	16% protein,					
			midbrain,	56.7%					
			putamen,	carbohydrate					
			caudate,	and 26% fat)					
			pallidum,	- Nasogastric					
			nucleus	water					
			accumbens						
			and						

			hypothalam						
			us)						
Sun et al.	n= 32	- Age: 25.3 ±	- After 30	- Milkshake	Orally	Glucose	Task-fMRI (taste	-6 mm FWHM	Responses to milkshake after the fixed
(2014) *		5.6	minutes	(per 945 ml;		Insulin	stimuli paradigm)		meal:
		- Sex: 15 M &	of the	918 kcal,		Ghrelin		-FWE at P<0.05	 ghrelin: amygdala[↑], midbrain[↑],
		17 F	meal	10.7%		TAG			insula [†] , pallidum [†] , hippocampus [†] .
		- BMI: 25.3 ±	intake	protein,					 triglycerides: midbrain ↓, insula
		4.5		52.4%					\downarrow ,hippocampus \downarrow , putamen \downarrow ,
				carbohydrate					pallidum ↓
			- ROI	and 25% fat)					
			(hippocamp	during;					
			us insula,	- Fasting					
			amygdala,	- Satiation					
			midbrain,	with fixed					
			putamen,	lunch meal					
			caudate,	(425 kcal for					
			pallidum,	women &					
			nucleus	625 kcal for					
			accumbens	men) and					
			and	satiation					
			hypothalam	with ad lib					
			us)	lunch meal					
Tatarann	n= 11	- Age: 34 + 3	- 25 minutes	- Liquid	Orally	Glucose	PET	-20 mm FWHM	After the liquid meal ingestion:
i et al.	n— 11		of the meal	formula	Orany	Insulin	121	20 1111 1 10 1101	- insulin \rightarrow OFC , insula
(1999) *		- Sex: All M	intake	meal (1.5		Leptin		-P<0.005.	- FFA \rightarrow OFC , insula , DLPFC \uparrow
(1777)		- BMI: not	mune	kcal/ml		FFA		uncorrected for	
		available		Keul/IIII		1171			

		19±6% body	- Whole	Ensure-Plus:				multiple	
		fat	brain	15% protein,				comparisons	
				53% carb &					
				32% fat)					
Wolnerha	n= 12	- Age: 24.8	- After 5	- 300 ml of	Nasogas	Insulin	Rs-fMRI	-5 mm FWHM	After glucose ingestion relative to
nssen et		years	minutes	glucose (75	tric tube	Glucose			placebo:
al.		- Sex: All M	of the	g)		GLP-1		-P< 0.05,	- insulin \rightarrow caudate \uparrow , pallidum \uparrow , OFC \uparrow
(2015) *		- BMI: 22.9	meal	- 300 ml of				uncorrected	
			intake	fructose (25					
				g)					
			- Whole brain	- 300 ml tap					
				water					
Zhang et	n=40 (20	- Age: 20-28	- Immedia	- Liquid	Orally	Glucose	rs-fMRI	-8 mm FWHM	After the liquid meal:
al.	NW &	- Sex: All M	tely	formula		Insulin			- insulin \rightarrow dACC \downarrow
(2015)	20	- BMI: 21.5 ±		meal (1.5				-P< 0.05. Monte	
	Obese)	1.4 (NW),	- ROIs	kcal/ml					
		33.6 ± 3.5	(dACC	Ensure-Plus:				Carlo corrected	
		(Obese)	and	15% protein,					
			precuneu	53% carb &					
			s)	32% fat)					
					1	Exogenously adr	ninistered regulators st	udies	
D (1 1	0	A 20 C	T 1 1	DVV	T /	CI			
Batterha	n=8	- Age: 29.6 ±	- Immediately	- PYY	Intraven	Glucose	Physiological fMRI	-4 mm FWHM	After PY Y infusion:
m et al.		2.1	XX /1 1	infusion	ous	Insulin			• ghrelin: hypothalamus \uparrow , VIA \uparrow &
(2007) *		- Sex: All M	- Whole	- Placebo				-Uncorrected for	brainstem ↑
		- BMI: 21.7 ±	brain +	(saline)		Ghrelin		ghrelin effect &	• PYY: globus pallidus ↑,
		0.7	ROIs	infusion				P<0.05 cluster-	middle frontal gyrus ↑, anterior lobe
			(solitary	Immediately				level corrected	cerebellum \uparrow , anterior cingulate \uparrow ,

			nucleus	after				for other	inferior parietal lobule ↑, medial
			and tract,	infusion				hormones effect	superior frontal gyrus ↑, substantia
			parabrac						nigra \uparrow , OFC \uparrow , peri- aqueductal grey
			hial						\uparrow , VTA \uparrow , precentral gyrus \uparrow ,
			nucleus,						parabrachial nucleus \uparrow , insula \uparrow ,
			substanti						putamen \uparrow , hypothal- amus \uparrow , superior
			a nigra,						tem- poral gyrus ↑, middle frontal
			nucleus						gyrus ↓, angular gyrus ↓
			accumbe						
			ns &						
			hypothal						
			amus)						
De Silva	n=16	- Age: 29.5	- After 90	Saline	- Orally	GLP-1	fMRI- food picture	-Smoothing level is	In response to fed saline state:
et al.		- Sex: 11 M	minutes of	infusion	for the	PYY	paradigm	not mentioned	- PYY infusion \rightarrow OFC \downarrow , nucleus
(2011) *		& 5 F	the infusion	Standard	breakfast				accumbens ↓
		- BMI: 22.1		breakfast	-			-P<0.05, cluster-	- GLP-1 infusion \rightarrow insula \downarrow
			- ROI	(579 kcal)	Intraveno			level corrected	
			(bilaterally	and then	us for				
			amygdala	saline	hormones				
			insula, OFC,	infusion,					
			nucleus	0.8					
			accumbens,	pmol/kg/min					
			caudate &	of GLP-17-					
			putamen)	36 amide					
				- 0.3					
				pmol/kg/min					
				of PYY 3-36					

					Combined PYY3-36 & GLP-17-36 amide & (0.3 pmol/kg/min & 0.8					
					pmol/kg/min					
Jones et al. (2012) *	n= 20	- Age: 34.1 - Sex:7 M & 5 F - BMI: 25.1	-	Immedia tely Whole brain	 Fasting state: 1. 1.25 or 5 pmol/kg/min of ghrelin injection intragastric lipid (dodecanoat e, C12) + ghrelin Postprandial state: 0.3 mmol/kg of ghrelin bolus saline 	Intraven ous	Ghrelin	Physiological fMRI	-Smoothing level is not mentioned -FWE whole- brain correction of P<0.05	 Ghrelin (pre-prandial state): thalamus ↑, hypothalamus (upper) ↑, midbrain, cerebellum↑, medulla and pons regions of the brainstem↑, parahippocampal gyrus (amygdala/ hippocampus) ↑, insula↑, precentral gyrus ↑, postcentral gyrus ↑ Ghrelin (post-prandial state): thalamus (ventral anterior nucleus) ↓, parahippocampal gyrus (amygdala/ hippocampus) ↓, insula↓, hypothalamus (upper)↓, midbrain and pons regions of the brainstem↓, medulla ↓, postcentral gyrus ↓, cerebellum↓ precentral gyrus & motor cortex ↑

Lassman	n=19	- Age: 37	-	Immediately	- 250 ml of	Intraven	ССК	Physiological fMRI	-Smoothing level is	- CCK: hypothalamus \downarrow , medulla \downarrow ,
et al.		- Sex:13 M		-	lipid	ous			not mentioned	midbrain \downarrow , precuneus \downarrow , cerebellum \downarrow ,
(2010) *		& 6 F	-	Whole	dodecanoic					cingulate gyrus \downarrow , caudate \downarrow , thalamus \downarrow ,
		- BMI: 25.4		brain	acid				-Uncorrected	temporal gyrus ↓
					- 250 ml of				P<0.005	
					saline (0.9%					
					control) or					
					- oral CCK					
					receptor					
					anatagonist					
					dexloxiglum					
					ide (600					
					mg),					
					administrate					
					d orally 1					
					hour before					
					the					
					intragastric					
					infusion					
Malik et	n=21	- Age: 24.1 ±	-	Immediately	- Ghrelin	Intraven	Insulin	Task-fMRI (food	-6 mm FWHM	• After ghrelin infusion \rightarrow
al.		1.1			infusion	ous	Glucose	picture paradigm)		hippocampus \uparrow , amygdala \uparrow , OFC \uparrow ,
(2008) *		- Sex: All M	-	Whole brain	- Placebo				-P< 0.001	caudate ↑, pulvinar ↑, VTA ↑,
		- BMI:			(saline)				uncorrected	substantia nigra ↑, insula ↑, occipital
		22.3 ± 0.7			infusion.					gyrus↑, fusiform ↑
					Two ghrelin					
					infusion (0.5					
					mg/kg for					
					each time					

				for 20					
				minutes)					
Page et	n= 9	- Age: 28 ± 5	- 30 minutes	- Euglycemia	Intraven	Glucose	fMRI-ASL	-Smoothing and	Hypoglycemia relative to euglycemia
al.		- Sex: 8 M	at the start	(2 mU/kg/	ous	Insulin		threshold levels	(hypoglycemia > euglycemia):
(2009) *		& 1 F	of	min of				are not	- hypothalamus ¹ , inferior frontal
()		- BMI: 23.6 ± 2	hypoglycem	insulin+ 20				mentioned	$g_{VIIIS} \uparrow ACC \uparrow caudate^{\uparrow} pars$
		2000	ic session.	% glucose					$triangularis L \uparrow superior$
			- 90 minutes	adjusted to					temporal gyrus \uparrow visual
			during the	achieve					association cortex \uparrow nutamen \uparrow
			session of	euglycemia					nars opercularis medial
			euglycemic	(plasma					frontal gyrus $ $ cerebellum $ $
			- Immediately	glucose- 95					nontai gyrus _‡ , cerebenum _‡
			after	mg/dL)					
			infusion	Hypoglycom					
			infusion	- Hypogryceni					
			Whole	ia (piasina					
			- whole	$g_{10}\cos e = 50$					
			brain+ ROI	mg/dL)					
			(nypotnalam						
			us)						
	21		T 1 1		.				
Page et	n= 21	- Age: $31.4 \pm$	- Immediately	- Euglycemia	Intraven	Insulin	Task-fMRI-	-6 mm FWHM	• Euglycemia relative to hypoglycemia
al.		7.9		(2 mU/kg/	ous	Ghrelin	(food picture		(euglycemia > hypoglycemia):
(2011) *		- Sex: 12 M	- Whole	min of		Leptin	paradigm)	-FWE whole-	anterior cingulate cortex [↑] ,
		& 9 F	brain	insulin+ 20				brain correction	ventromedial- prefrontal cortex ↑
		- BMI: 25.2±4		% glucose				of P<0.05	
				adjusted to					
				achieve					

			euglycemia					
			(plasma					
			glucose= 95					
			mg/dL)					
			- Hypoglycem					
			ia (plasma					
			glucose = 50					
			mg/dL)					
n = 22	- Age: 25.9 +	- After 95	- Fasted-	Intraven	Glucose	Task-fMRI (food	-6-mm FWHM	• Ghrelin $\rightarrow OFC \uparrow$ hippocampus \uparrow
	17	minutes of	saline	0115	Insulin	nicture		
	- Sex: 17 M	the meal	injection	ous	GI P-1	produce paradigm)	-FDR at P<0.05	
	& 5 F	intaka	Fed saline		DVV	paradigini	1 DIX at 1 <0.05	
	\mathbb{R} \mathbb{R} \mathbb{R} \mathbb{R} \mathbb{R} \mathbb{R} \mathbb{R} \mathbb{R} \mathbb{R} \mathbb{R} \mathbb{R} \mathbb{R} \mathbb{R}	intake	- Peu sainte		Chrolin			
	- DIVII. 23.9 \pm	Whole	witti		trialyzarid			
	0.0	- whole	standard		trigiycend			
		brain+	breakfast		es			
		ROI	(730 kcal,					
		(nucleus	55% CHO,					
		accumbe	31% fat &					
		ns,	14%					
		caudate,	protein)					
		anterior	Fed ghrelin					
		insula,	injection					
		amygdal	with					
		a,	standard					
		hippoca	breakfast					
		mpus,	(730 kcal,					
		OFC)	55% CHO,					
		*	31% fat &					
r	1= 22	$h = 22 - Age: 25.9 \pm 1.7 - Sex: 17 M & 5 F - BMI: 23.9 \pm 0.6$	$n=22 - Age: 25.9 \pm 1.7 + 1.7$	$= 22 - Age: 25.9 \pm 0.6 + 0.6$	$= 22 - Age: 25.9 \pm 0.6 + 0.6$	$= 22 - Age: 25.9 \pm 0.6 + 0.6$	$= 22 + Age: 25.9 \pm 1.7 + After 95 + Hypoglycem ia (plasma glucose= 95 mg/dL) - Hypoglycem ia (plasma glucose= 50 mg/dL) + Hypoglycem ia (plasma glucose= 50 mg/dL) + Hypoglycem ia (plasma glucose= 50 mg/dL) + Hypoglycem is a (plasma glucose= 50 $	$= 22 + Age: 25.9 \pm 1.7 + After 95 + Fasted injection = 0 + FDR at P<0.05 + FDR at P<0.05 + FDR at P<0.05 + FOR at P<0.05 + F$

				14%					
				protein)					
Little et	n=12	- Age: 38 ± 3.4	- One hour	- 250 ml of	Intraven	Glucose	Physiological	-4 mm FWHM	- Glucose vs saline:
al. (2014)		- Sex: 7 M	of the	glucose	ous &	Insulin	fMRI		- glucose: hypothalamus↓, brainstem
		& 5 F	meal	(45g)	intragastri	CCK		-FDR at P<0.05	↓, medulla↓, pons↓, cerebellum↓
*		- BMI: 19.7–	intake	following 2	c infusion	GLP-1			cerebellum anterior \downarrow , lingual \downarrow ,
		28.9		placebo					fusiform↓, thalamus↓
			- Whole	tablets					 insulin: hypothalamus↓, brainstem↓,
			brain	- 250 ml of					medulla↓, pons↓, cerebellum↓
				glucose					cerebellum anterior \downarrow , lingual \downarrow ,
				(45g)					fusiform \downarrow , thalamus \downarrow .
				following					- Glucose + dexloxiglumide vs saline:
				600 mg of					
				CCK1					 insulin: cerebellum↓, lingual gyrus
				receptor					\downarrow , cuneus \downarrow
				antagonist					- GLP-1: cerebellum \downarrow , lingual gyrus
				(dexloxiglu					\downarrow , cuncus \downarrow
				mid)					
				- 250 ml of					
				saline					
				(0.9%,					
				control)					
				following 2					
				placebo					
				tablets					

Schilling	n=48	- Age: 23.96 ±	- After 30	- Oral cortisol	Intraven	GLP-1	fMRI-ASL	-12mm FWHM	• Intranasal insulin infusion \rightarrow
et al.		3.4	minutes	vs.	ous	Insulin			putamen ↑, insula↑, inferior frontal
(2014) *		- Sex: All M	of	intranasal				-P= 0.05 corrected	gyrus↑, caudate nucleus↑
		- BMI: 20 <	infusion	insulin				for multiple	
		BMI < 25		- Oral cortisol				comparisons	
			- Whole brain	vs. oral					
			+ ROIs	placebo					
			(hippocamp	- Oral vs.					
			us, insula,	intranasal					
			putamen)	placebo.					
				- Intranasal					
				insulin vs.					
				intranasal					
				placebo					
				- Insulin (100					
				I.E. /ml) &					
				cortisol (30					
				mg)					
van	n= 32	- Age: 57.8 ±	- Immedia	1. Exenatide,	Intraven	GLP-1	Task-fMRI (food	-8 mm FWHM	• Obese healthy: exenatide vs placebo
Bloeme	(16 NW	1.9 (NW), 58	tely	2. Exenatide	ous	Insulin	picture		→ amygdala \downarrow , insula \downarrow & OFC \downarrow .
ndaal et	& 16	± 2.1 (Obese)		together			paradigm)	-FWE at P<0.05	
al.	Obese)	- Sex: 8 M & 8	- ROIs	with the				corrected for	
(2014)		F for each	(insula,	GLP-1				multiple	
		group	striatum,	receptor				comparisons	
		- BMI: 23.2 ±	amygdal	antagonist					
		0.4 (NW),	a, and	exendin 9-					
		32.6 ± 0.7	OFC)	39, or					
		(Obese)							

3. Placebo		
(Fig. 1A).		
The par-		
ticipants		
were		
blinded to		
the type of		
infusions.		

* Indicates included studies in the Activation Likelihood Estimation meta-analysis

ACC, anterior cingulate cortex; ASL, arterial spin labelling; BMI, body mass index; CBF, cerebral blood flow; CCK, cholecystokinin; DLPFC, Dorsolateral prefrontal cortex; Dex, dexloxiglumide; FDR, false discovery rate; FFA, free fatty acids; fMRI, functional magnetic resonance imaging; F, female; FEW, family wise erroe; FWHM, full width at half maximum; GLP-1, Glucagon-like peptide-1; M, male; NGT, nasogastric tube, OFC, orbitofrontal cortex; PET, position emission tomography; PYY, peptide YY; rs-fMRI, resting state fMRI; rCBF, regional cerebral blood flow; ROI, region of interest; SMA, supplementary motor area; VTA, ventral tegmental area.

Table 4.4. Cochrane risk of bias assessment of the included studies in the systematic review.

Author., (year)	Appropriatecross-over design †	Randomized treatment order	Carry-over effect †	Unbiased data †	Allocation concealment	Blinding of participants and personnel	Blinding of outcome assessment	Incomplete outcome data	Selective outcome reporting	Other bias
Al-Zubaidi et al. (2019) *	Low	Unclear	Low	Low	Unclear	Unclear	Unclear	Unclear	Low	Low
Batterham et al. (2007) *	Low	Unclear	Low	Low	Unclear	Unclear	Unclear	Unclear	Low	Low
De Silva et al. (2011) *	Low	Unclear	Low	Low	Unclear	Unclear	Unclear	Unclear	Low	Low
Dorton et al. (2017) *	Low	Unclear	Low	Low	Unclear	Unclear	Unclear	Unclear	Low	Low
Eldeghaidy et al. (2016) *	Low	Unclear	Low	Low	Unclear	Unclear	Unclear	Low	Low	Low
Gautier et al. (2000) **	NA	Unclear	NA	NA	Unclear	Unclear	Unclear	Low	Low	Low

Heni et al. (2015) **	NA	Unclear	NA	NA	Unclear	Unclear	Unclear	Unclear	Low	Low
Jakobsdottir et al. (2012) *	Low	Unclear	Low	Low	Unclear	Unclear	Unclear	Low	Low	Low
Jones et al. (2012) *	Unclear	Unclear	Unclear	Low	Unclear	Unclear	Unclear	Unclear	Low	Low
Kroemer et al. (2013) **	NA	Unclear	NA	NA	Unclear	Unclear	Unclear	Unclear	Low	Low
Lassman et al. (2010) *	Unclear	Unclear	Unclear	Low	Unclear	Unclear	Unclear	Unclear	Low	Low
Lennerz et al. (2013)	Low	Unclear	Low	Low	Unclear	Unclear	Unclear	Low	Low	Low
Li et al. (2012) *	Low	Unclear	Low	Low	Unclear	Unclear	Unclear	Unclear	Low	Low
Little et al. (2014) *	Unclear	Low	Unclear	Low	Unclear	Unclear	Unclear	Low	Low	Low
Liu et al. (2000) *	Low	Unclear	Unclear	Low	Unclear	Unclear	Unclear	Unclear	Low	Low
Malik et al. (2008) **	NA	Unclear	NA	NA	Unclear	Unclear	Unclear	Low	Low	Low
Page et al. (2009) *	Low	Unclear	Low	Low	Unclear	Unclear	Unclear	Unclear	Low	Low

Page et al. (2011) **	NA	Unclear	NA	NA	Unclear	Unclear	Unclear	Unclear	Low	Low
Page et al. (2013) *	Low	Low	Low	Low	Unclear	Unclear	Unclear	Unclear	Low	Low
Pannacciulli et al. (2007)	NA	Unclear	NA	NA	Unclear	Unclear	Unclear	Unclear	Low	Low
**										
Schilling et al. (2014) *	Unclear	Unclear	Low	Low	Unclear	Unclear	Unclear	Unclear	Low	Low
Spetter et al. (2014) *	Low	Unclear	Low	Low	Unclear	Unclear	Unclear	Low	Low	Low
Sun et al. (2014) *	Low	Unclear	Low	Low	Unclear	Unclear	Unclear	Unclear	Low	Low
Tataranni et al. (1999) **	NA	Unclear	NA	NA	Unclear	Unclear	Unclear	Unclear	Low	Low
van Bloemendaal et al.	Low	Unclear	Low	Low	Unclear	Low	Unclear	Low	Low	Low
(2014)										
Wolnerhanssen et al.	Low	Unclear	Low	Low	Unclear	Unclear	Unclear	Low	Low	Low
(2015) *										
Zhang et al. (2015) **	NA	Unclear	NA	NA	Unclear	Unclear	Unclear	Low	Low	Low

[†]Domains specific only for cross-over study design. *Cross-over trials. **Controlled trials. NA; not applicable.

4.4.2 <u>Modulation of brain responses to appetite and satiety regulators</u>

As a first step, the data extracted from the 31 studies were grouped into brain areas that associated: 1) positively and/or 2) negatively with appetite regulators, 3) associated positively and/or 4) negatively with satiety regulators. Data from brain areas from each of the sub-groups were then pooled and common brain areas across studies evaluated. Changes in brain responses from the baseline were not shown in this review as it was not reported of the authors of the studies included. A full list of overlapped brain areas is reported in Table 4.5. Figure 4.2 illustrates the concurrence of key brain areas commonly reported across studies based on the findings from the systematic review for NW adults.

4.4.2.1 Appetite regulators

In NW adults, eight studies reported positive association of ghrelin with concurrence in brain activation (Batterham et al., 2007, Malik et al., 2008, De Silva et al., 2011, Jones et al., 2012b, Li et al., 2012, Kroemer et al., 2013, Goldstone et al., 2014, Sun et al., 2014) mostly found in the amygdala (five studies) (Malik et al., 2008, Jones et al., 2012b, Li et al., 2012, Kroemer et al., 2013, Sun et al., 2014), OFC (five studies) (Malik et al., 2008, De Silva et al., 2011, Li et al., 2012, Goldstone et al., 2014, Sun et al., 2014), insula (four studies) (Malik et al., 2008, Jones et al., 2014, Sun et al., 2014), insula (four studies) (Malik et al., 2008, Jones et al., 2012b, Li et al., 2014), insula (four studies) (Malik et al., 2008, Jones et al., 2012b, Li et al., 2012, Sun et al., 2014), and hippocampus (four studies) (Malik et al., 2008, De Silva et al., 2011, Jones et al., 2012b, Goldstone et al., 2014), Figure 4.2A. A single study reported negative association with ghrelin concentrations in the caudate nucleus, hypothalamus, insula, amygdala, hippocampus, and thalamus (Jones et al., 2012b).

Regarding Obese adults, no study examined brain areas associated by appetite regulators in Obese adults.

4.4.2.2 Satiety regulators

In terms of satiety regulation in NW adults, eight studies reported positive correlation in response to satiety regulators with concurrence in the ACC (three studies) (Batterham et al., 2007, Page et al., 2009, Page et al., 2011) and putamen (three studies) (Batterham et al., 2007, Page et al., 2009, Schilling et al., 2014),

Figure 2.2B. Fifteen studies reported attenuation in activity in various brain areas (Tataranni et al., 1999, Gautier et al., 2000, Liu et al., 2000, Page et al., 2009, Lassman et al., 2010, De Silva et al., 2011, Jakobsdottir et al., 2012, Li et al., 2012, Page et al., 2013, Schilling et al., 2014, Spetter et al., 2014, Sun et al., 2014, Heni et al., 2015, Eldeghaidy et al., 2016, Al-Zubaidi et al., 2019). Most studies showed concurrence in the insula (eight studies) (De Silva et al., 2011, Jakobsdottir et al., 2014, Sun et al., 2014, Eldeghaidy et al., 2016, Al-Zubaidi et al., 2019, Most studies) (Liu et al., 2012, Schilling et al., 2014, Spetter et al., 2014, Sun et al., 2014, Eldeghaidy et al., 2016, Al-Zubaidi et al., 2019), hypothalamus (five studies) (Liu et al., 2000, Page et al., 2009, Lassman et al., 2010, Page et al., 2013, Spetter et al., 2014), OFC (four studies) (Gautier et al., 2000, De Silva et al., 2011, Li et al., 2012, Heni et al., 2015), thalamus (four studies) (Gautier et al., 2000, Lassman et al., 2010, Li et al., 2012, Eldeghaidy et al., 2016), putamen (four studies) (Gautier et al., 2000, Page et al., 2000, Page et al., 2013, Spetter et al., 2014, Sun et al., 2014), and caudate nucleus (three studies) (Lassman et al., 2010, Li et al., 2012, Page et al., 2013), Figure 4.2C.

In Obese adults, one study out of the four studies reported positive correlation in response to satiety regulators which was correlated with the nucleus accumbens (Lennerz et al., 2013). Two studies reported negative association in response to satiety regulators in the OFC (Gautier et al., 2000, van Bloemendaal et al., 2014). A single study reported negative correlation in the insula and amygdala (van Bloemendaal et al., 2014), ACC (Zhang et al., 2015) and hippocampus, precuneus, putamen and thalamus (Gautier et al., 2000).

Table 4.5. List of reported brain areas in the systematic review. This list included brain areas correlated with endogenously released or exogenously administered appetite or satiety regulators.

Brain areas	Number of studies	Reference						
	(percentage)							
Brain areas correlated positively with appetite regulators (8 studies)								
Amygdala	5 (62.5%)	Malik et al. (2008); Jones et al. (2012);						
		Kroemer et al. (2013); Li et al. (2012); Sun						
		et al. (2014)						

OFC	5 (62.5%)	Malik et al. (2008); De Silva et al. (2011); Li
		et al. (23); Goldstone et al. (2014); Sun et al.
		(2014)
Insula	4 (50%)	Malik et al. (2008); Jones et al. (2012); Li et
		al. (2012); Sun et al. (2014)
Hippocampus	4 (50%)	Malik et al. (2008); De Silva et al. (2011);
		Jones et al. (2012); Goldstone et al. (2014)
Palldium	3 (37.5%)	Kroemer et al. (2013); Li et al. (2012); Sun et
		al. (2014)
Midbrain	3 (37.5%)	Jones et al. (2012); Kroemer et al. (2013); Sun
		et al. (2014)
Hypothalamus	3 (37.5%)	Batterham et al. (2007); Jones et al. (2012);
		Kroemer et al. (2013)
Fusiform gyrus	2 (25%)	Malik et al. (2008); Kroemer et al. (2013)
Thalamus	2 (25%)	Jones et al. (2012); Kroemer et al. (2013)
Caudate	2 (25%)	Malik et al. (2008); Kroemer et al. (2013)
Brainstem	2 (25%)	Batterham et al. (2007); Jones et al. (2012)
VTA	2 (25%)	Malik et al. (2008); Batterham et al. (2007)
Brain areas	correlated positivel	y with satiety regulators (12 studies)
ACC	3 (25%)	Batterham et al. (2007); Page et al. (2009);
		Page et al., (2011)
Putamen	3 (25%)	Batterham et al. (14); Page et al. (26);
		Schilling et al. (29)
Insula	2 (16.6%)	Batterham et al. (2007); Schilling et al. (2014)
Caudate	2 (16.6%)	Page et al. (2009); Schilling et al. (2014)
Hypothalamus	2 (16.6%)	Batterham et al. (2007); Page et al. (2011)
Inferior frontal gyrus	2 (16.6%)	Page et al. (2009); Schilling et al. (2014)
Superior	2 (16.6%)	Batterham et al. (2007); Page et al. (2009)
gyrus		
Nucleus accumbens	1 (8%)	Lennerz et al. (2013)
OFC	2 (16.6%)	van Bloemendaal et al. (2014); Gautier et al.
		(2000)
I		

Brain areas	negatively correlat	ed with satiety regulators (15 studies)
Insula	9 (60%)	Al-Zubaidi et al., (2019); De Silva et al.,
		(2011); Eldeghaidy et al., (2016); Jakobsdottir
		et al., (2012); Li et al., (2012); Schilling et al.,
		(2014); Spetter et al. (2014); Sun et al. (2014);
		van Bloemendaal et al. (2014)
Hypothalamus	5 (33%)	Lassman et al. (2010); Liu et al. (2000); Page
		et al. (2009); Spetter et al. (2014); Page et al.
		(2013)
OFC	4 (26%)	De Silva et al. (2011); Heni et al. (2015); Li et
		al. (2000); Gautier et al. (2000)
Thalamus	4 (26%)	Eldeghaidy et al. (2016); Lassman et al.
		(2010); Li et al. (2012); Gautier et al. (2000)
Putamen	4 (26%)	Spetter et al. (2014); Page et al. (2013); Sun et
		al. (2014); Gautier et al. (2000)
Caudate	3 (20%)	Lassman et al. (2010); Li et al. (2000); Page et
		al., (2013)
Cerebellum	3 (20%)	Eldeghaidy et al. (2016); Lassman et al.
		(2010); Page et al. (2009)
Temporal gyrus	2 (13%)	Eldeghaidy et al. (2016); Lassman et al.
		(2010)
ACC	1 (7%)	Zhang et al. (2015)
Amygdala	1 (7%)	van Bloemendaal et al. (2014)
Hippocampus	1 (7%)	Gautier et al. (2000)
Precuneus	1 (7%)	Gautier et al. (2000)

ACC, anterior cingulate cortex; OFC, orbitofrontal cortex; VTA, ventral tegmental area.



Figure 4.2. Results of the systemic review showing concurrence of key brain areas commonly reported across studies conducted on NW adults. (A) Brain areas positively correlated with appetite regulators, showing the concurrence in the insula, amygdala, hippocampus, and orbitofrontal cortex (OFC). (B) Brain areas positively correlated with satiety regulators showing the concurrence in the anterior cingulate gyrus (ACC) and the putamen. (C) Brain areas negatively correlated with satiety regulators showing the concurrence in the insula, caudate, thalamus, hypothalamus, OFC, and putamen.

4.4.3 <u>Coordinate-based neuroimaging meta-analysis</u>

In a second step, the concurrence/overlap in brain regions activated in response to changes in appetite and satiety regulators in NW adults were examined quantitatively using neuroimaging meta-analysis. In determining the total number of studies required for the coordinate-based meta-analysis in this thesis, the established guidelines for ALE meta-analyses were adhered. The literature recommends a minimum of 17 to 20 experiments to ensure sufficient statistical power and robustness in detecting convergent activation patterns across studies (Muller et al., 2018). After reviewing the available studies, 20 experiments were included in the analysis, aligning with the upper range of this recommendation to maximise the analysis's reliability and validity. Meta-analysis was not conducted for Obese adults because of the low number of studies including this population. Studies were initially grouped following the same methods used for the systemic review: 1) brain areas correlated positively with appetite regulators (4 studies), 2) and/or correlated negatively with appetite regulators (1 study), 3) brain areas correlated positively with satiety regulators (8 studies), 4) and/or correlated negatively with satiety regulators (12 studies). However, due to the small number of studies (less than the 17 required for the ALE analysis), for each of these subgroups, separate meta-analysis was not performed. Instead, two primary metaanalyses were performed: one for appetite regulators and the other for satiety regulators, each combined across studies reported negative and positive correlation with brain responses.

4.4.3.1 Concurrence of brain area modulated by appetite regulators

Of the five studies eligible for the meta-analysis with appetite regulators, four assessed positive correlation (Malik et al., 2008, Jones et al., 2012b, Kroemer et al., 2013, Goldstone et al., 2014), and a single study assessed negative correlation (Jones et al., 2012b). Due to the low number of investigations, this analysis was not performed.

4.4.3.2 Concurrence of brain area modulated by satiety regulators

In terms of the satiety analysis, the ALE and ABC meta-analyses across 14 independent studies (20 experiments combined across increased/decreased brain activation to satiety regulators) included 212 NW adults and 123 foci provided convergent results, revealing the same cluster (Figures 4.3, and 4.4) in the caudate nucleus. For the ALE-analysis, the caudate cluster was centered at MNI (-10,12,6) and for the ABC analysis at MNI (-12,10,8). Four studies contributed to the caudate cluster in the ALE-analysis, while five studies contributed to the ABC-analysis (Table 4.6). The forest plot of the ABC approach (Figure 4.5) demonstrated that two studies reported positive correlation (increase in caudate activity) while three studies reported negative correlation (decrease in caudate activity) with satiety regulators. In addition, ALE analysis revealed additional cluster in the hypothalamus centered at MNI (2, -4, -12), with five studies contributed to this cluster (Table 4.6, and Figure 4.3).

Table 4.6. Studies and relative foci coordinates in MNI space contributing to the identified clusters in employed meta-analyses on satiety regulators, using the activation likelihood estimation (ALE) method and Analysis of Brain Coordinates (ABC) approach.

ALE	ABC	Coordi	NI space							
		x	У	Z						
Contributors to cuadate cluster										
Page et al. (2009)	Page et al. (2009)	-10.5	8.2	12.7						
Page et al. (2013)	Page et al. (2013)	-10.4	9.9	4.4						
Lassman et al. (2010)	Lassman et al. (2010)	-14.1	18.4	5.1						
Wolnerhassen et al.	Wolnerhassen et al.	-11.3	10.4	10.8						
(2015)	(2015)									
	Little et al. (2014)	-14.1	4.3	4.0						
Ca	ontributors to hypothala	mus cluster								
Lassman et al. (2010)		-9	0.0	-7.7						

Batterham et al. (2007)	-6	-11.94	-10.23
Page et al. (2009)	-3	-5.7	-9.8
Pannacciulli et al.	-4	-4.0	-19.32
(2007)			
Little et al. (2014)	-4	-1.0	-13



Figure 4.3. Results of the ALE meta-analysis showing convergent clusters with significant ALE values correlated with satiety regulators showing correlation in (A) the caudate nucleus centered at MNI (-10,12,6), Z = 4.62, ALE value= $1.5 \times 10-2$, cluster volume= 1000 mm3, and (B) the hypothalamus centered at MNI (2, -4, -12), Z = 4.21, ALE value= $1.32 \times 10-2$, cluster volume= 1728 mm3. Maps are family-wise error (FWE)-corrected for multiple comparisons P < 0.05.



Figure 4.4. Results of the ABC meta-analysis, showing convergent clusters in the caudate nucleus centred at MNI (-12, 10, 8), False Discovery Rate (FDR) corrected for multiple comparisons <0.05.



Figure 4.5. The forest plot from the ABC analysis illustrating the effect sign associated with studies contributing to the increased and/or decreased caudate activity in response to satiety regulators.

4.5 Discussion

This systematic review included studies that directly assessed brain activation in response to appetite/satiety regulators endogenously released, following acute food ingestion, and/or exogenously administered regulators in NW and Obese adults. In addition, two different coordinate-based meta-analysis approaches (ALE and ABC) were employed to reveal convergent brain areas of neurohormonal gut-brain signalling to appetite/satiety regulators.

4.5.1 <u>Concurrence of brain area modulated by appetite regulators</u>

The systematic review revealed that the insula was one of the most reported regions across studies in NW adults, with an overlap in 50% of the studies in response to appetite regulators. These results supported the hypothesis and are in agreement with those of the previous systematic review (Zanchi et al., 2017). The insula is an important relay that connects the hypothalamus, OFC, and limbic system. It is often referred to as "ingestive cortex" because it contains primary taste neurons, projecting from the oral cavity (Scott and Plata-Salamán, 1999), as well as primary visceral afferents from the gut (Craig, 2002). The insula encodes multi-modal sensory features of foods (de Araujo et al., 2012), and its activity is modulated by hunger and satiety (de Araujo et al., 2006).

The systemic review also identified the amygdala, and OFC as key brain areas associated with appetite regulators in 62% of the studies in NW adults. The amygdala, and OFC encode motivation value of food cues (Jay et al., 2003). More specifically, the amygdala pass information about sensory cues onto the OFC and has an important role in reward processing (Padoa-Schioppa and Assad, 2008). These results agree with the literature (Zanchi et al., 2017). However, these analyses to assess the overlap quantitively in appetite regulators were not conducted due to the small number of studies identified for the coordinate based meta-analysis. No study examined brain areas associated by appetite regulators in Obese adults.

4.5.2 <u>Concurrence of brain area modulated by satiety regulators</u>
The results from the systematic review and meta-analysis revealed strong concurrence across studies in NW adults and confirms the role of hypothalamus in appetite regulation which support the hypothesis. The hypothalamus is widely recognised as the gatekeeper to control food intake, highly influenced by nutrients, it is physically connected to other areas involved in maintaining homeostatic energy balance and receives projections from the gastrointestinal tract via the brainstem (Blouet and Schwartz, 2010). The results from the systematic review revealed a negative correlation with satiety regulators, which is inconsistent with the previous systematic review that reported associations in opposite directions. These discrepancies might be due to variations in the inclusion criteria. Unlike Zanchi et al. (2017), the work in this chapter did not combine NW and Obese adults or include studies with participants below the age of 18 years. In addition, the current focus was on the neuromodulation of appetite/satiety regulators in response to acute food intake, hence long-duration intervention studies were not included, unlike Zanchi et al. (2017), which may have also led to differences in findings.

The caudate nucleus is associated with perception of food stimuli, reward processing, and cognitive appetite control (Chen and Zeffiro, 2020). The negative correlation of the caudate nucleus with satiety regulators reported in the current review support the study's hypothesis and was confirmed by the ALE and ABC meta-analysis methods. However, it is important to note that the results from the meta-analyses were combined across negatively and positively correlated studies. The relation of caudate nucleus to hunger and satiety is not yet clear, which perhaps explains the findings in this study. While some neuroimaging studies show reduction in caudate activity (Batterham et al., 2007, De Silva et al., 2011) in response to satiety regulators, others show increased activity (Page et al., 2009, Wolnerhanssen et al., 2015). Increases in caudate activity after a meal could reflect top-down attentional control (Balleine et al., 2007), whereas the suppression after meal termination could be due to the dopamine-driven inhibition response (Mehta et al., 2012).

This systematic review found the thalamus to be another key area of the brain involved in correlating satiety regulators in NW adults. Thalamic brain activity has been reported to vary as a function of hunger or satiety (Tataranni et al., 1999), ghrelin application (Higgins et al., 2007) and glucose infusion (Jones et al., 2012b, Little et al., 2014). The thalamus is the gateway to sensory perception, and it plays a major role in integrating proprioceptive information from the gastrointestinal tract (Kelley et al., 2005, Little et al., 2014) through the vagus nerve (Coss-Adame and Rao, 2014). The results from this systematic review demonstrate a negative correlation between thalamic activity with satiety regulators, which may reflect the role of food stimulation in modulating thalamus activity. This could be due to the role of the thalamus in integrating sensory perception (visual and taste cues) or the connection with the vagus nerve which sends information regarding the meal size and physical characteristics. The systematic review also revealed the association of insula with satiety regulators with 53% of the studies reporting decreases in insula activation in response to satiety regulators.

In terms of studies in Obese adults, only four studies were included in this review, and no meta-analysis was performed due to the limited number of the studies. Brain areas including those reported in the NW studies (insula, ACC, OFC, thalamus, putamen, hippocampus, and precuneus) also showed to be associated by satiety regulators. However, due to the limited number of studies and diversity of results, no conclusion can be made in Obese adults regarding the concurrence of brain regions associated by appetite and satiety regulators. Hence, further studies that recruit Obese adults are required to elucidate the altered mechanism of appetite in people with obesity.

4.6 Differences between endogenously released and exogenously administered hormones

The systematic review's findings indicated that certain brain regions were more affected by endogenously generated hormones than by hormones that were given externally. For instance, it demonstrated a greater role of endogenously released appetite regulators in the amygdala and OFC, and endogenously released satiety regulators in the insula and thalamus modulation. The following statements could be used to explain these observations. The thalamus and insula's involvement are most likely a result of their functions as multimodal food processing regions. The thalamus plays a role in integrating sensory perception (visual and taste cues), which sends information regarding the meal size and physical characteristics (Kelley et al., 2005, Little et al., 2014). The insula is thought of as a multi-modal sensory area encoding food stimulation (de Araujo et al., 2006). The motivation value of food signals is encoded by the amygdala and OFC in relation to their engagement with appetite endogenous regulators ((Gottfried et al., 2003). More specifically, the OFC, which is crucial to reward processing, receives information regarding sensory cues from the amygdala (Padoa-Schioppa and Assad, 2008). These findings could demonstrate that food stimulation has a role in modulating the activity of the above brain regions.

For the exogenously administered hormones, the coordinates meta-analysis shows that exogenously induced satiety hormones modify the hypothalamus and caudate nucleus responses. As previously mentioned, the caudate nucleus is involved in reward processing and cognitive appetite control, while the hypothalamus is thought of as the gatekeeper for controlling food intake (Chen and Zeffiro, 2020). These result demonstrate the effectiveness of exogenously administered appetite/satiety regulators in regulating energy intake in human.

4.7 Strengths and limitations

At the time of writing this thesis, this is the first study to employ functional neuroimaging meta-analysis to quantitatively define the overlap of brain areas associated with satiety regulators across studies. The results from the generated activation maps of the meta-analysis (in our case from a total of 212 participants across the included studies) are more robust than those of any individual imaging study. The generated activation map from the NW adults can be used as a reference or baseline to compare alterations with obesity, or in people with altered eating

behaviour. Another strength is the stringent and well-defined inclusion and exclusion criteria, which enabled an unbiased assessment of the central mechanisms regulating satiety and appetite. The present systematic review provides data and a clear overview of appetite neuroimaging findings in NW adults.

This study has a number of limitations as described below:

- it included a very limited number of studies recruiting adults living with obesity.
- it included relatively small number of appetite/satiety studies in the metaanalysis, a consequence of the strict inclusion selection criteria. This did not allow sub-analyses on appetite/satiety regulators positively/negatively correlated with brain responses to be performed, or to investigate possible differences in neurohormonal gut-brain signalling in response to endogenously released appetite and satiety regulators compared with exogenously infused regulators.
- the small number of studies included in the meta-analysis might be responsible for the absence of other key brain areas related to appetite and satiety regulations including the insula, thalamus and amygdala, OFC and ACC. The location of brain clusters/foci for the OFC and ACC activations varies widely across studies and this might also explain the lack of their concurrence in the analysis.
- Although a strict selection criterion was employed in this study, the presence of individual differences in food preferences is possible.
- There are variations in the imaging modalities (task-based fMRI, resting state fMRI and cerebral blood flow), study designs (endogenously released or exogenous administration), and food-related paradigms/stimulations (i.e., food images or taste stimuli) in the studies included, which might introduce some heterogeneity. While the ideal would be to separate the studies based on stimulation type and/or study design and/or imaging modality and performed separate analyses, this was not possible due to the small number of studies. Despite these variations and drawbacks, the robustness of the

findings is supported by similar results obtained with two different approaches (ALE and ABC), particularly for the caudate convergence. In addition, variations/heterogeneity in the included studies may reduce the sensitivity of the meta-analysis, making it likely to be conservative, rather than causing false positive results/activations.

• Finally, another limitation of this review was that the risk of bias analysis was performed using the Cochrane risk of bias guideline (Higgins et al., 2019). As this tool is not specifically designed for neuroimaging studies, the risk of bias might be inappropriately estimated (Acar et al., 2018).

4.8 Conclusion

This systematic review and quantitative meta-analyses add to the growing body of evidence describing brain areas involved in appetite and satiety processing. Robust evidence was presented from the systematic review and two different coordinate based meta-analysis approaches/methods (ALE and ABC) for the importance of the hypothalamus and caudate nucleus in association with appetite and satiety processing. Although there was a clear correlation between these brain regions and satiety regulators, the results of this study could be affected by additional factors that were not examined, such as psychological disorders like mood disorders or any prior dietary preferences of the participants. No possible conclusion could be made for the adults living with obesity due to the very limited number of studies conducted on these people. Hence, more work is needed to fully elucidate the complex interactions associated with the central regulation of appetite/satiety in adults with normal-weight and those with obesity. This study could help future work to understand the underlying mechanism of appetite regulation in normal-weight participants. Moreover, the results of the meta-analysis can be used for comparison in future studies to define alterations with obesity or altered eating behaviours and develop new treatment strategies. Future research should take advantage of these appetite-regulatory systems pharmacologically which could mimic some of the significant impacts of bariatric surgery.

5 An MRI study to measure the effect of a standard pasta meal on gastrointestinal responses in adults with and without obesity

Advantages in MRI allows exploring the gastrointestinal function, by combining different measurements in a single scan session. The Nottingham GI-MRI research group has developed non-invasive MRI techniques to assess post-prandial GE and SBWC, and SMA blood flow in NW participants. To date, there been no research conducted to assess SBWC and SMA in individuals with obesity after food consumption. This study aims to building on previous work in NW participants to explore for the first-time alterations in GI physiology (GE, SBWC, and SMA) in people with obesity, using advanced MRI techniques.

The work detailed in this chapter was initiated by colleagues in the University of Nottingham Medical School under the direction of Dr Liz Simpson. Approximately 70% of data collection was completed before the author became involved. The author was directly involved in collection of the remaining data and analysis and interpretation of all data is her own work. Data was presented as an oral presentation at the 2023 Annual Meeting of the Surgical Research Society (SRS).

Althubeati, S., Simpson, E.J., Bush, D., Hoad, C., Elgdeghaidy, S., Gowland, P., Macdonald, I.A. and Lobo, D.N., 2023. O032 Gastrointestinal and satiety responses to oral feeding in participants with obesity compared with those of healthy weight: an MRI study. British Journal of Surgery, 110(Supplement_3), pp.znad101-032.

5.1 Introduction

There is conflicting evidence to suggest that obesity is associated with specific changes in the GI tract (Sam et al., 2012). Previous studies have shown that Obese participants have alterations in their GI functions, including altered responses of gut hormones and GE rate (Mora et al., 2005, Meyer-Gerspach et al., 2014). However, other studies showed no differences in GE and gut hormone responses between

Obese and NW participants (Pironi et al., 1993, Carroll et al., 2007, Flint et al., 2007, Yang et al., 2009).

In addition to the above factors, as explained in section 2.4, SBWC and SMA blood flow influence the process of food regulation and might have an impact on how people with obesity regulate food intake. Previous studies have reported that SBWC decreased after both liquid and solid meal intake (Marciani et al., 2010, Marciani et al., 2012) and have showed an increased SMA blood flow after consuming liquid meals in adults with a NW (Moneta et al., 1988, Sieber et al., 1989). Measuring SBWC and SMA in people with obesity can provide insights into how the gastrointestinal system functions in individuals with excess body weight. Understanding these dynamics may contribute to a better understanding of the metabolic and digestive alterations that often accompany obesity.

5.2 Aims and hypothesis

Primary aim

To compare the effect of ingestion of a mixed meal on GE rate between the NW and Obese.

Secondary aim

To compare the effect of ingestion of a mixed meal on GCV, SMA blood flow, SBWC, subjective satiety measured by the VAS, plasma hormone concentrations (insulin, PYY, ghrelin, GLP-1, and GLP-2), blood glucose, triglycerides and FFAs between NW and Obese.

Hypothesis

Obese participants will have lower GCV and faster GE rate, as well as lower SBWC, SMA blood flow, postprandial hormone concentrations, and satiety feelings following the ingestion of a mixed meal compared with NW adults.

5.3 Methods

5.3.1 Study design and ethics

This observational, unblinded study was carried out at Sir Peter Mansfield Imaging Centre (SPMIC) at the University of Nottingham. Ethical approval for the study was obtained from the Medical School Research Ethics Committee at the University of Nottingham, reference A16042015. Registered at https://ClinicalTrials.gov refence NCT03860623.

5.3.2 Eligibility criteria

5.3.2.1 Inclusion criteria

Healthy participants with no history of gastrointestinal motility disorders (e.g., irritable bowel syndrome, gastroesophageal reflux disease, gastroparesis, sphincter of Oddi dysfunction, etc.), diabetes, or previous thoracic or abdominal surgery and were non-smokers were recruited. Participants aged 18–60 years old with a BMI of normal-body weight (BMI 18.5-24.9 kg/m2) or Obese (BMI 30–40 kg/m2) who self-reported stable weight in the previous 3 months. Posters on University Campuses, local community spaces (post offices, libraries, community boards at supermarkets), and local social media platforms (Facebook) were used to advertise the study and recruit participants.

5.3.2.2 Exclusion criteria

Those with acute illness in the preceding 6 weeks, on regular medication, with a history of substance abuse or having factors precluding safe MRI were excluded. Additional exclusion criteria included a history of gastrointestinal motility disorders (e.g. gastroesophageal reflux disease irritable bowel syndrome, gastroparesis, sphincter of Oddi dysfunction, etc.), previous thoracic or abdominal surgery.

5.3.3 <u>Sample size calculation</u>

No previous data were available for GE data after consumption of a solid meal in Obese to inform sample size requirements for this study. Therefore, data from a previous study conducted by Marciani et al. (2013a) on 12 NW participants, who consumed a mixed solid/liquid 542 kcal meal with half emptying time of gastric content (T₅₀) of 132 ± 26 minutes (mean \pm SD), was used to calculate sample size for this study. Employing this in the sample size calculation, 11 participants in each independent group would give the 90% statistical power (α = 0.05) to detect a 29% difference in T₅₀ in the Obese group.

5.3.4 <u>Study protocol</u>

Participants were invited to attend 2 visits: a screening visit at the David Greenfield Physiology Unit (DGPU) and a single MRI study visit at the SPMIC as described below. Participants were asked to consume a standard diet for the 3 days immediately before the MRI study visit. This normalised pre-study diet, which was based on foods in their normal diet, was designed to provide 15% of total daily energy intake as protein, 35% as fat and 50% as carbohydrate, calculated from their estimated energy requirements. Participants were also asked to refrain from nicotine, caffeine, and alcohol from 6 pm prior to the study day, and to fast overnight.

Screening visit

Healthy volunteers, who had expressed their interest in the study, were invited to attend a screening visit. If they met the eligibility criteria and were willing to participate, participants signed a consent form, and completed a health questionnaire, an MRI safety questionnaire, and an abnormal scan form. Participants were then asked to come at the DGPU for blood samples to measure their fasting glucose levels, urea and electrolytes, liver function, thyroid function, and full blood count. Additionally, their weight, height, and blood pressure were recorded. Participants were then asked to fill out 4-days food intake diary using household measures. Dietary data were then analysed using the Nutritics software (https://www.nutritics.com/en/).

MRI visit

The GI-MRI protocol used in this study was based on protocol previously developed and used by our GI MRI research group (Hoad et al., 2007, Totman et al., 2009, Chowdhury et al., 2016). On the study day, participants reported at SPMIC at 8:00 and were cannulated by a trained research nurse, using venous cannula retrograde

into a dorsal foot vein, for blood sampling. Baseline measurements (T0) of MRI scans (GCV, SBWC, and SMA blood flows), blood samples (PYY, GLP-1, GLP-2, ghrelin, glucose, and insulin) and satiety VAS questionnaires were collected. Participants were then given a test meal (~500 g portion) with 250 ml tap water, and they were asked to consume the meal within 15 minutes in an upright, seated position outside the scanner. Following ingestion of the test meal (T45), participants were scanned at 30 minutes intervals for 4 hours to assess GCV, SBWC, and SMA blood flow. Satiety VAS questionnaires were collected immediately after eating the pasta meal (T45), and then every 60 minutes (at T105, T165, T225, T285 and T345) throughout the experiment (five hours). During the MRI session, a 0.9% saline infusion was used to maintain the patency of the cannulation line, and the foot with the cannula was placed in a warming box compatible with MRI (air temperature 50-55oC). The warming box was used to collect arterialised-venous blood sampling which is important in measuring blood glucose concentrations. Arterialized venous blood is consisted of pure arterial blood mixed with a particular amount of idealised venous blood (Collis and Neaverson, 1967). Monitoring the arterial blood glucose concentration is necessary to precisely study how glycaemic change affects physiological responses and hormone concentrations (Liu et al., 1992). In addition, sampling from arterialized blood could obtain higher probability of identifying small variations in postprandial GLP-1 availability with interventions (Chen et al., 2018).

5.3.5 Test meal

The test meal was comprised of white pasta, tomato-based pasta sauce, olive oil and cheddar cheese (Astbury et al., 2011, Astbury et al., 2014). Appendix 10.2.3 gives a full description of the pasta meal preparation. The energy and macronutrient composition of the meal is shown in Table 5.1.

Nutrient	Nutrition information per 100 g
Energy (kcal)	145

Table 5.1. Nutrient composition of the pasta meal.

Carbohydrate (g)	19.3
Total sugar (g)	2.9
Fat (g)	5.1
Protein (g)	5.7
Fibre (g)	1.7

5.3.6 Blood samples and metabolic assays

Blood samples were drawn via a three-way tap; the first 2 ml were discarded to remove saline from the sampling line. Blood was drawn at baseline (T0), and then at 15 minutes (T60) and 30 minutes (T75) following the ingestion of the test meal. After that, every 30 minutes until 5 hours postprandial (T315). Blood samples were processed as soon as possible after collection (unless stated). Processing of samples were carried out as described below.

- Blood glucose: analysed immediately after drawing using YSI 2300 STATplus (Yellow Springs Instruments) and the remaining blood was dispensed.
- Ghrelin and PYY: 2 ml of blood were added to ethylenediaminetetraacetic acid (EDTA) (purple top) vacutainer[™] tubes containing 100µL of Trasylol (aprotinin) and then centrifuged at 3000 g and 4°C for 10 minutes.
- GLP-1: 1 ml of blood was added to EDTA purple vacutainer[™] tubes containing 10 µL of dipeptidyl peptidase-4 and then centrifuged at 8000 g and room temperature for 2 minutes.
- Insulin, triglycerides, and GLP-2: 5 ml of blood were added to (gold top) SSTTM VaccutainerTM tubes at baseline and 3 ml were used for subsequent analysis. Samples were left to clot for 15-20 min and then centrifuged at 3000 g and room temperature for 10 minutes. Triglycerides were only measured at baseline fasting state.

FFA: 1.5 ml of blood were added to a lithium heparin microtube containing 50 µl of Ethylene Glycol Tetraacetic Acid (EGTA)/glutathione and 10µl tetrahydrolipstatin (THL) preservative and then centrifuged at 8000 g and room temperature for 2 minutes. It was only measured at baseline fasting state.

Plasma was then removed and stored in a -80° C freezer for later analysis of the GLP-1, ghrelin, PYY and FFA. Aliquots of serum were stored in the -80° C freezer for later analysis for GLP-2, triglycerides, and insulin. Analysis was carried out by the research technicians (Sally Cordon) based in the Medical School using commercially kits as descried below.

- Insulin: Invitrogen enzyme-linked immunosorbent assay (catalogue no: ELISA KAQ1251) kits.
- Total PYY Merck ELISA kits (catalogue no: EZHPYYT66)
- Total GLP-2: Merck ELISA kits (catalogue no: EZGLP2-37K)
- Active GLP-1: Merck ELISA kits (catalogue no: EGLP-35K)
- Total ghrelin: Invitrogen ELISA kits (catalogue no: BMS2192)
- FFA: WAKO reagents and plate reader assays
- TAG: Horiba Medical reagents and Pentra-400 analyser

5.3.7 Gut imaging protocol and analysis

MRI scanning was carried out using a 3T scanner (Ingenia Philips, the Netherlands). dStream Anterior Coil was used and placed over the participant's abdomen. The participant's foot was placed in a warming box compatible with MRI to allow for blood sampling. The following scan protocols were used to measure GCV, SBWC, and SMA blood flow.

GCV: was collected using a sequence of coronal Half-Fourier Acquisition Singleshot Turbo spin Echo (HASTE) acquiring two stacks of 14 contiguous axial slices with a slice thickness of 10 mm with no gaps between slices and reconstructed inplane resolution 0.78 mm \times 0.78 mm and SENSE of 2.0. The parameters of the sequence were: flip angle 90, TR=840 ms, TE=60 ms. Data were acquired during 2 breath-hold of 10 s each.

Data were analysed using Medical Imaging processing, analysis, and visualization software (MIPAV, Center for Information Technology, National Institute of Health, Bethesda, Maryland, U.S, https://mipav.cit.nih.gov/). Gastric content MRI images were analysed using the Analyze7.5 through 28 image slices for each time point. Within each image slice, a volume of interest (VOI) was manually drawn around GCV. The GCV was then calculated from the sum of the volumes measured from each image slice. T₅₀ was calculated when GCV dropped to 50% of initial postprandial gastric volume (V₀). The rate of GCV T₅₀ was defined as the rate of volume change in ml/min at the calculated T₅₀ volume (Parker et al., 2016). T₅₀ was calculated using the GE modelling by the fitted data using the following equation V(t)=V 0f(1+kt/t-empt) exp((-t)/t-empt)+(1-f)(1-Gt) (Parker et al., 2016). All data were included for fitting until the volume was less than 100 ml.V₀ is the GCV at time 0, and f, k, tempt, and G are fitted models. The parameters V₀, t_empt, k, G and f are all variable parameters that are fitted during the modelling process which will be different for each individual emptying curve. In general, they have the following characteristics for the curve.

- V0 is the volume of the meal at Time 0
- *t_empt* is the exponential decay curve time constant (exp = exponential)
- *k* is the parameter which allows some lag time before the decay and allows for the volume to go up which means that secretions can also be accounted for in the early phase of the emptying
- *G* is the gradient of the linear component of the decay which tends to be dominant at the end of emptying
- *f* is the fraction of the decay that behaves like an exponential decay with (1f) being the fraction that behaves like a linear decay.

SBWC: was acquired as previously described (Hoad et al., 2007, Marciani et al., 2013b) using a sequence of coronal single-shot fast spin echo acquiring two stacks of 14 contiguous coronal slices with slice thickness 7 mm with no gaps between slices, reconstructed in-plane resolution 0.78 mm \times 0.78 mm and SENSE of 2.2, flip angle 90, TE=400 ms, TR=1728 ms. The data were acquired during 2 breath holds of 24 s each.

Data were analysed using a previously described and validated technique by Hoad et al., (2007) in the Matlab® platform (The Mathworks Inc, Cambridge, UK, https://uk.mathworks.com/). This technique identifies fluid in the MRI scans based on the intensity of CSF, which was used in the analysis to calculate a threshold value. Free fluids within the small bowel above the threshold value were segmented and measured. The segmented data were saved as part of the analysis and volumes of fluid were measured from the segmented data across the small bowel.

SMA blood flow: was collected using a sequence of phase-contrast MRI angiography measurements with 20 cardiac phases acquiring 14 contiguous coronal slices with slice thickness 6 mm with no gaps between slices, reconstructed in-plane resolution $1.17 \text{ mm} \times 1.17 \text{ mm}$, and SENSE of 3, flip angle 25° , TE (shortest), TR (shortest), and velocity encoding 140 cm/s for baseline and 200 cm/s for other time points. The data were acquired during a breath hold of under 20 s and varied in length due to differences in heart rate between participants, as it depends on the cardiac cycle for each participant. An MRI-compatible peripheral pulse-oximeter unit (PPU) was used to allow cardiac gating for the SMA blood flow scan.

Data were analysed using the SEGMENT software (Medviso AB, Lund, Sweden, https://medviso.com/segment/). A region of interest (ROI) was manually drawn around the superior mesenteric artery when it was best defined (much brighter or darker than the grey background). The software determined the velocity, flow rate, and net volume of blood flow.

5.3.8 Statistical analysis

GraphPad Prism 9.5 for Mac OS X was used to analyse the data and make graphs (GraphPad Software, La Jolla, CA, USA, <u>www.graphpad.com</u>). Unless stated,

parametric data are expressed as mean and standard deviation (SD), whereas nonparametric data are expressed as the median and interquartile range (IQR). The normality of the data was tested by the Shapiro-Wilk test. To test for any significance of differences between groups in variables measured at a single time point, an unpaired t-test was used for parametric data and the Mann-Whitney U-test for non-parametric data. Two-way analysis of variance (ANOVA) was used to identify differences between groups for repeated measurements, (NW vs. Obese). If significant differences between the groups were detected by the 2-way ANOVA analysis, these were further probed using the Bonferroni post hoc test. Differences were considered significant at P<0.05. The total area under the curve (AUC) for postprandial responses was calculated using the trapezoid rule, which is a numerical integration technique to approximate the integral or the area under a curve (Yeh, 2002). The Trapezoidal Rule divides the whole area into smaller trapezoids in order to evaluate the area under the curves instead of using rectangles. This integration estimates the area by using a trapezium to calculate the area under the curve. The postprandial AUC was calculated by the GraphPad Prism for the area under the reading curve down to fasting baseline values (Shiang, 2004). The peaks that were below the baseline value were also considered in the AUC calculation.

To test correlation between GCV and CSS, Pearson test for parametric data and Spearman test for non-parametric data were used. Additionally, a univariate general linear model (multiple linear regression) was calculated using IBM SPSS Statistics (version 28) to evaluate the variance in the CSS determined by the independent variables [GCV and body weight]. First, correlation between CSS (a dependent variable) and GCV (independent variable) was analysed including all participants. Then, correlation between CSS (a dependent variable), GCV (first independent variable) and body weight (second independent variables).

5.4 Results

5.4.1 <u>Participant's descriptive results</u>

Figure 5.1. illustrates the recruitment process. One hundred sixty-five participants contacted the research team and were assessed for eligibility. Thirty-one participants were screened for the study, of which three were excluded after screening and seven withdrew. Twenty male participants were included in the final analysis for the study, 10 NW and 10 Obese. Table 5.2 summarises the demographic characteristics of the participants.

Some data were missing and were not included for the final study analysis for the following reasons. One of the Obese participants had MRI data available only up to 180 minutes because of technical issues with the Magnet at the start of the study day which doesn't allow for final scans. Also, baseline data for SMA blood flow was not collected because of technical issues, hence the SMA data for this participant was not included when calculating the AUC. Another Obese participant had missing SBWC data for the timepoint of 30 minutes because of technical issues. To resolve this issue, average data was calculated for the data at timepoint 0 minutes and timepoint 60 minutes to estimate the missing value. Breathing data for 4 participants (one NW and 3 Obese) was not of sufficient quality to analyse GCV. To remove this effect, GCV was analysed using SBWC MRI images which is less affected by poor breathing.



Figure 5.1. Recruitment flow diagram

Table 5.2. Summary demographic characteristics of the participants. Data are presented as the mean \pm SD.

	NW group	Obese group	P-value
Age	37.8 ± 14.2	38.5 ± 13	0.91
Height (cm)	174.7 ± 7.1	176 ± 3.7	0.62
Weight (kg)	72.6 ± 8.2	103.6 ± 7	< 0.0001
BMI (kg/m ²)	23.8 ± 1.9	33.6 ± 3.3	< 0.0001
Systolic blood pressure	126.7 ± 10.9	134.9 ± 27.3	0.38
(mmHg)			
Diastolic blood pressure	76 ± 10.2	81 ± 13.4	0.36
(mmHg)			

5.4.2 Gastric content volume

Examples of stomach MRI images of the GCV in NW and Obese participants at fasting/baseline and after consumption of the test meal are shown in Figures 5.2 and 5.3. The time course values for GCV when fasted and following the pasta meal in each group is shown in Figure 5.4a. There were no significant differences in fasted GCV between the NW and Obese groups (P=0.96, Table 5.3). Gastric volume was significantly increased by eating in both groups (NW: 873% \pm 175%, P<0.0001 vs. Obese: 859% \pm 156%, P<0.0001). Mean values of GCV peaked at 0 minutes and levels returned to baseline by 240 minutes in both groups (Figure 5.4a). Average T₅₀ time were not different between the NW and Obese group (148 \pm 16 minutes vs. 170 \pm 15 minutes, respectively, P=0.87) with an average T₅₀ rate of 1.7 ml/min for the NW group and 1.8 ml/ minute for the Obese group (P=0.89). No significant group × time interactions (P=0.33) or main effect of the group (P=0.26) were found in GCV. A significant main effect of the time was observed (P<0.0001). AUC for postprandial GCV (T0-T275) did not differ between the NW and Obese groups (Table 5.4, P=0.22).

The GCV was correlated with CSS in each group. The results showed a significant positive correlation between GCV and CSS in both groups (CSS: NW: r = 0.65, <0.0001; Obese: r=0.55, P<0.0001; Figures 5.4b). In addition, results from the multiple linear regression model showed that R² was equal to 0.30 for relationship between CSS vs. GCV for all participants. However, R² improved significantly to 0.44 (P<0.001) for the relationship between CSS vs. GCV vs. body weight. This give indication that 30% of variance in the CSS is explained by GCV and 14% by body weight.





Figure 5.2. MRI images of stomach of a normal-weight participant at fasting baseline and after consumption of the test meal.



Figure 5.3. MRI images of stomach of a participant with obesity at fasting baseline and after consumption of the test meal.

	NW group	Obese group	P-value
MRI measurements			
Fasting GCV (ml)	59.2 ± 32.6	58.6 ± 29.5	0.96
(mean ± SD)			
Fasting SMA blood flow (ml/s)	7.7 ± 2.5	6.5 ± 2.7	0.30
(mean ± SD)			
Fasting SBWC (ml)	84 (107.3)	118.5 (184.7)	0.35
[median (IQR)]			
Satiety VAS scores			
Fasting DTE (mm)	57.7 ±29	66.2 ± 14.7	0.20
(mean ± SD)			
Fasting hunger (mm)	55.9 ± 27.1	64.2 ± 18.8	0.43
(mean ± SD)			
Fasting fullness (mm)	25.3 ± 26.5	25.6 ± 19.9	0.97
(mean ± SD)			
Fasting PFI (mm)	60.2 ± 19.2	71.6 ± 16.7	0.17
(mean ± SD)			
Fasting CSS (mm)	39.1±22.6	30.9 ± 13.5	0.33
(mean ± SD)			
Blood measurements			
Fasting PYY (pg/ml)	103 ± 49.6	88 ± 32.4	0.41
(mean ± SD)			
Fasting active GLP-1 (pmol/l)	7.2 ± 11.5	1.4 ± 2.5	0.15
(mean ± SD)			
Fasting GLP-2 (ng/ml)	1.82 ± 1	2.10 ± 1.3	0.72
(mean ± SD)			
Fasting total ghrelin (pg/ml)	3202 ± 1134.6	2857 ± 923.2	0.46
(mean ± SD)			
Fasting insulin (mIU/l)	12.5 ± 5.8	17.6 ± 11.3	0.21
(mean ± SD)			
Fasting blood glucose (mmol/l)	4.4 ± 0.9	4.1 ± 0.4	0.41
(mean ± SD)			
	1	1	1

Table 5.3. Summary of baseline (fasting) characteristics of participants.

Fasting triglycerides (mmol/l)	0.64 ± 0.17	1.7 ± 0.82	0.0009
(mean ± SD)			
Fasting FFAs (mmol/l)	0.28 ± 0.16	0.37 ± 0.09	0.11
(mean ± SD)			

BMI; body mass index, CSS; combined satiety scores, DTE; desire to eat, FFAs; free fatty acids, GCV; gastric content volume, GLP-1; glucagon-like peptide-1, NW; normal-weight group, PFI; prospective food intake, PYY; peptide YY, SBWC; small bowel water content, SMA; superior mesenteric artery, VAS; visual analogue scale.



Figure 5.4. (a) The gastric content volume (GCV) over time following the consumption of the pasta meal in normal-weight (NW) and Obese participants. Values are presented as the mean and SD, n=10 in each group. (b) Correlation of GCV with composite satiety scores (CSS) in the NW group (r=0.65, $R^2=0.43$) and the Obese group (r=0.55, $R^2=0.30$).

	NW group	Obese group	P-value
MRI measurements (0-285 min)			
GCV (ml)	2476±477.8	2734± 444.6	0.22
SMA blood flow (ml/s)	33.3 ± 25.9	50.2 ± 26.9	0.17
SBWC (ml)	772.9± 342.8	927.4 ± 395.6	0.36
Satiety VAS scores (0-345 min)			
PFI (mm)	140± 91.6	99.2±122.7	0.33
Hunger (mm)	162.1±79.6	113.8 ± 121.1	0.30
DTE (mm)	155.3±86.9	122.5±108.9	0.46
Fullness (mm)	166.1±112.8	102.7±120	0.23
CSS (mm)	154.9±70.9	107.9±97.3	0.23
Blood measurements (0-315 min)			
Total PYY (pg/ml)	182 ± 314.6	175.4 ± 246.5	0.95
Active GLP-1 (pmol/l)	11.8 ± 80.23	16.2 ± 21.55	0.86
GLP-2 (ng/ml)	2.93 ± 6.8	1.85 ± 7.9	0.75
Total ghrelin (pg/ml)	3375±7950	2211±6587	0.73
Insulin (mIU/l)	222.3 ± 104.4	433.2 ± 263.6	0.03
Blood glucose (mmol/l)	6.6 ± 6.8	7.3 ± 4.5	0.79

Table 5.4. Total area under curve (AUC). Data are presented as the mean ± SD.

CSS; combined satiety scores, DTE; desire to eat, GCV; gastric content volume, GLP-1; glucagon-like peptide 1, NW; normal-weight group, PFI; prospective food intake, PYY; peptide tyrosine tyrosine, SBWC; small bowel water content, SMA; superior mesenteric artery, VAS; visual analogue scale.

5.4.3 Superior mesenteric artery blood flow

There were no differences in fasted SMA blood flow between NW and Obese groups (P= 0.30, Table 5.3). The flow in SMA increased significantly by eating in both groups (NW: $121\% \pm 34\%$; P<0.0001 vs. Obese: $136\% \% \pm 36\%$, P<0.0001) reaching a peak at 0 minutes for the NW group and 30 minutes for Obese group with no difference between them (P=0.71, Figure 5.6). SMA blood flow dropped to baseline levels by 210 minutes in both groups (Figure 5.5). No significant group × time interactions (P=0.23) or main effect of the group (P=0.70) was shown. A significant main effect of the time was observed (P<0.0001). Postprandial AUC (T0-T275) did not differ between the groups (P=0.17, Table 5.4).



Figure 5.5. Superior mesenteric artery (SMA) blood flow values over time for normalweight (NW) and Obese participants following the consumption of the pasta meal. Values are presented as the mean and SD, n=10 in each group.

5.4.4 Small bowel water content

There were no differences between the NW and Obese groups in the fasted state (P=0.35, Table 5.3). Highest volumes of SBWC were shown in the fasted state in both groups and then volumes started to decrease after feeding toward lowest values at 150 minutes in both groups (Figure 5.6). No significant group × time interactions (P=0.99) was shown. There was a trend for a group main effect for higher SBWC for the Obese compared with the NW group (P=0.097). A significant main effect of the time was observed (P<0.0001). Total AUC of postprandial SBWC (T0-T275) did not differ significantly between the groups (P=0.37, Table 5.4). Examples of maximum intensity projections (MIP) images of the segmented SBWC of NW and Obese participants at fasting baseline and after consumption of the test meal are shown in Figures 5.7 and 5.8.



Figure 5.6. Small bowel water content (SBWC) volume over time for normal-weight (NW) and Obese participants following the consumption of the pasta meal. Values are presented as the median and IQR, n=10 in each group.



Figure 5.7. Examples of maximum intensity projections (MIP) images of small bowel water content of a normal-weight (NW) participant at fasting/baseline and at different time points following the consumption of the pasta meal.



Figure 5.8. Examples of maximum intensity projections (MIP) images of small bowel water content of a participant with obesity at fasting/baseline and at different time points following the consumption of the pasta meal.

5.4.5 <u>Subjective Satiety</u>

The four domains (DTE, hunger, fullness, and PFI) of the VAS questionnaire to assess feelings of satiety and the CSS when fasted and following the pasta meal in NW group and Obese group are illustrated in Figure 5.9. No differences were found in fasted values between the NW and Obese groups for all VAS domains (DTE, P=0.20; hunger, P=0.43; fullness P=0.97, and PFI, P=0.17) and CSS (P=0.33).

There was no significant group \times time interactions for all VAS domains and CSS was shown. However, there was a significant group main effect observed for DTE, hunger, PFI, and CSS. The Obese group had significantly higher scores for the DTE, hunger, and PFI and lower CSS compared to the NW group (group effect: all P=0.02, Figure 5.9a, 5.9b, 5.9d, and 5.9e). For the fullness domain, there was a trend for higher scores for the Obese group compared to the NW group (group main

effect: P=0.09, Figure 5.9c). A significant main effect of the time was observed for all VAS domains and CSS (all P<0.0001). Postprandial AUC (T0-T335) did not differ between the groups in all VAS domains (DTE, P=0.46; hunger, P=0.30; fullness P=0.23, and PFI, P=0.33; Table 5.4) and CSS (P=0.23, Table 5.4). For the Bonferroni post hoc test, a significant difference was seen between the NW and Obese groups at 120 minutes for the PFI domain (P=0.04, Figure 5.9d). As expected, all postprandial ratings in both groups of PFI, and DTE decreased significantly after eating, reaching their lowest value at 0 minutes. Ratings of fullness increased after eating, reaching a peak at 0 minutes in both groups.







Figure 5.9. Subjective satiety ratings over time using Visual Analogue Scale (VAS) scores of (a) desire to eat, (b) hunger, (c) fullness, (d) prospective food intake, and include composite satiety scores (e) at fasted and following the consumption of the pasta meal in normal-weight (NW) and Obese participants. Values are presented as the mean and SD, n=10 in each group. *Values are significant different between the two groups (P<0.05).

5.4.6 Gut hormones

Total PYY

No differences were seen in fasting PYY concentrations between the NW and Obese groups (P=0.41, Table 5.3). Also, no significant group \times time interactions (P=0.91) or the group main effect (P=0.34) was shown. A significant main effect of the time was observed (P=0.0001). As shown in Figure 5.10, no differences were shown in total PYY concentrations between the two groups in response to eating (group effect: P=0.34, Figure 5.10). No differences were shown for total AUC of postprandial (T0-T305) concentrations between the NW and Obese groups (P=0.95, Table 5.4). The time to the peak were 90 minutes for the NW and 30 minutes for the Obese group. Concentrations then return to baseline values by 300 minutes in both groups.



Figure 5.10. Total peptide YY (PYY) levels over time at fasted and following the consumption of the pasta meal in normal-weight (NW) and Obese participants. Values are presented as the mean and SD, n=10 in each group.

Active glucagon-like peptide-1

No differences were seen in fasting GLP-1 concentrations between the NW and Obese groups (P=0.19, Table 5.3). No significant group × time interactions (P=0.30) or the group main effect (P=0.16, Figure 5.11) was observed. A significant main effect of the time was observed (P<0.0001). No differences were also shown for AUC of postprandial concentrations (T0-T305) between the NW and Obese groups (P=0.86, Table 5.4). The NW group had a huge variability in GLP-1 concentrations at fasted and following the consumption of the pasta meal compared with the Obese group. The time to the peak was 90 minutes for the NW group, and 15 minutes for the Obese group. Concentrations then return to baseline values at 300 minutes for the NW group and after 300 minutes for the Obese group.



Figure 5.11. Active glucagon-like-peptide-1 (GLP-1) levels over time at fasted and following the consumption of the pasta meal in normal-weight (NW) and Obese participants. Values are presented as the mean and SD, n=10 in each group.

Glucagon-like peptide-2

No differences were seen in fasting GLP-2 concentrations between the NW and Obese groups (P=0.72, Table 5.3). No significant group × time interactions (P=0.58) or the group main effect (P=0.92, Figure 5.12) was found. A significant main effect of the time was observed (P<0.0001). No differences were shown for total AUC for

postprandial concentrations between the NW and Obese groups (P=0.75, Table 5.4). The time to the peak was 60 minutes for both groups. Concentrations then returned to baseline values at 210 minutes in both groups.



Figure 5.12. Glucagon-like peptide-2 (GLP-2) levels over time at fasted and following the consumption of the pasta meal in normal-weight (NW) and Obese participants. Values are presented as the mean and SD, n=10 in each group.

Total ghrelin

No differences were seen in fasting ghrelin concentrations between the NW and Obese groups (P=0.46, Table 5.3). No significant group× time interactions (P=0.77) or the group main effect (P=0.77, Figure 5.13) was shown. A significant main effect of the time was observed (P<0.0001). No differences were shown for total AUC for postprandial concentrations between the NW and Obese groups (P=0.73, Table 5.4). The time for postprandial nadir (minimum concentrations of ghrelin concentrations) was 60 minutes for the NW group and 90 minutes for the Obese group. Concentrations then return to baseline values at 300 minutes in NW group and 210 minutes in the Obese group



Figure 5.13. Total ghrelin levels over time at fasted and following the consumption of the pasta meal in normal-weight (NW) and Obese participants. Values are presented as the mean and SD, n=10 in each group.

5.4.7 Other blood results

Insulin

No differences were seen in fasting insulin concentrations between the NW and Obese groups (P=0.21, Table 5.3). There was a trend for group × time interactions (P=0.07) and a significant group main effect (P=0.015). In response to eating, the Obese group had higher insulin concentrations compared to the NW group across the 5 hours (Figure 5.14). A significant main effect of the time was observed (P<0.0001). Total AUC for postprandial concentrations was significantly higher for the Obese group than the NW group (P=0.03, Table 5.4). The time to the peak was 30 minutes for both groups. Concentrations then return to baseline values by 300 minutes in both groups.


Figure 5.14. Insulin levels over time at fasted and following the consumption of the pasta meal in normal-weight (NW) and Obese participants. Values are presented as the mean and SD, n=10 in each group. *Values are significant different between the two groups (P<0.05).

Blood glucose

No differences were seen in fasting blood glucose concentrations between the NW and Obese groups (P=0.41, Table 5.3). In response to eating, no differences were shown in blood glucose concentrations between the two groups (Figure 5.15). There were no significant group × time interactions (P=0.57) or the group main effect (P=0.47). A significant main effect of the time was observed (P<0.0001).No differences were shown for total AUC for postprandial concentrations between the two groups (Table 5.4, P=0.79). The time to the peak was 30 minutes for the NW group and 60 minutes for the Obese group. Concentrations then return to fasted values at 210 minutes in the NW group and 300 minutes in the Obese group.



Figure 5.15. Blood glucose levels over time at fasted and following the consumption of the pasta meal in normal-weight (NW) and Obese participants. Values are presented as the mean and SD, n=10 in each group.

Triglyceride

The triglyceride concentration was measured only in the fasted state. The Obese group had significantly higher concentrations compared with the NW group (P=0.0009, Table 5.3, Figure 5.16).



Figure 5.16. Fasting triglyceride concentrations in normal-weight participants (NW group) and Obese participants. Values are presented as the mean and SD, n=10 in each group. *** Values are significant different between the two groups (P<0.0001).

Free fatty acids

Concentrations of FFAs were measured only at the fasting level. There were no differences between the two groups (P=0.11, Table 5.3, Figure 5.17).



Figure 5.17. Fasted free-fatty acids (FFAs) concentrations in normal-weight participants (NW group) and Obese participants. Values are presented as the mean and SD, n=10 in each group.

5.5 Discussion

This is the first study, to our knowledge, that uses MR methods to measure the GI functions of GCV, GE, SMA blood flow, and SBWC as well as gut hormone release, blood glucose, triglycerides, FFAs, and subjective satiety in NW and Obese groups. The results showed no differences between the NW and Obese groups in terms of GCV, GE, SMA blood flow, and SBWC, as well as all hormones. Except for insulin, and blood glucose. However, subjective satiety ratings were significantly lower in the Obese group compared with the NW group.

GE is controlled by gut hormones and other factors, and it is considered a critical factor in regulating food intake and satiety. Consequently, it may play a role in the pathogenesis of obesity and this study hypothesised that participants with obesity would have a higher GE rate compared to the NW participants, as reported by Mora et al. (2005). However, the observations of Mora et al. (2005) are not unequivocal and the present study adds to previous findings, suggesting no differences in GE rate between NW and Obese participants following food intake were found (Buchholz et al., 2013, Seimon et al., 2013). The present study, and the study of Seimon et al. (2005), included only male participants, while the study of Mora et al. (2005), included both male and female participants which could explain the inconsistency seen in the results.

Investigating SBWC could help to a better understanding of food intake regulation in people with obesity, as it reflects small intestine functions (Dellschaft et al., 2022). Postprandial plasma concentrations of PYY facilitate fluid absorption via the intestinal mucosal surface to increase small intestine transit time, hence reducing SBWC (Savage et al., 1987, Benelam, 2009). In this study, PYY concentrations were increased slightly, while the SBWC decreased after both groups ate the pasta meal. This is consistent with previous studies, in which SBWC decreased after both liquid and solid meal intake in NW participants (Marciani et al., 2010, Marciani et al., 2012). This is the first study that compared SBWC between NW and Obese participants after feeding, hence, the results of the Obese participants couldn't be compared to the previous studies.

SMA blood flow is known to increase after food intake to aid nutrient absorption. Previous research has shown that gastric distension could stimulate SMA blood flow (Vanis et al., 2010, Vanis et al., 2012). In the present study, SMA blood flow increased significantly after eating in both groups with no differences between them. The lack of differences between groups may be a result of the same energy content being consumed by both groups, as previous studies have suggested that blood flow is positively correlated with energy content (Sidery and Macdonald, 1994, Parker et al., 1995, Someya, 2007). These findings of higher postprandial responses of SMA blood flow support its role in food digestion and absorption (Jeays et al., 2007). Hence, SMA blood flow after eating may be an important marker of digestive processes in both health and illness, which could have a role in food intake regulation.

To date, no study has compared SMA blood flow between NW and Obese participants following food intake. The results of the NW group support those of previous studies, wherein SMA blood flow increased after liquid and solid meals in NW participants (Moneta et al., 1988, Sieber et al., 1991).

Some previous studies have found that circulating postprandial plasma concentrations of PYY and GLP-1 are reduced, while ghrelin is increased in people with obesity, when compared to NW individuals, suggesting that these hormones have a role in the pathophysiology of obesity (Batterham et al., 2003, Meyer-Gerspach et al., 2014). However, GLP-1, PYY and ghrelin were not different in this study between the groups at fasting and in response to the pasta meal intake. The findings of our study are consistent with are those of Adam and Westerterp-Plantenga (2005), who observed no differences between the fasting and postprandial responses of GLP-1 between NW and Obese participants after eating a standard breakfast meal with water. Similarly, a study conducted by Yang et al. (2009) did

132

not find differences in fasting and postprandial PYY responses between NW and Obese participants following high-carbohydrate (88% carbohydrate, 4% fat and 8% protein) and high-fat (25% carbohydrate, 4% protein and 71% fat) breakfast meals. Moreover, Carroll et al. (2007) reported no differences between NW and Obese participants in their fasting and postprandial responses regarding GLP-1 and ghrelin after a soy liquid meal. These results conflict with another study that showed higher GLP-1 and lower ghrelin responses in the Obese participants compared with NW participants after three protein liquid meals (soy, whey and gluten) (Bowen et al., 2006). However, they reported that Obese participants had higher fasting GLP-1 and lower ghrelin concentrations than NW participants, which might influence the postprandial responses. Of note, test meals varied widely in the previous studies and included liquid, semi-solid and solid meals with different macronutrient compositions, which make for inconsistent results across the studies.

There is a lack of studies investigating the relationship between endogenous GLP-2 and satiety and GE in humans. Unlike GLP-1, GLP-2 has no incretin effect on blood glucose homeostasis and insulin. The main effects of GLP-2 are seen to maintain mucosal morphology, intestinal integrity, and nutrient and energy absorption. Previous studies showed no influence of GLP-2 infusion on GE (Schmidt et al., 2003, Meier et al., 2006), while one study showed it slowed GE (Nagell et al., 2004),

Blood glucose concentrations may have a role in changes to appetite, as a reduction in blood glucose concentrations can increase appetite and initiation of food consumption, while an increase in feelings of satiety has been shown to occur with hyperglycaemia (Mayer, 1955). However, it is difficult to distinguish whether changes which occur to satiety, as a result of elevated blood glucose concentration, stem from hyperglycaemia alone or from other metabolic changes / processes which co-exist, such as endogenous insulin production. Fasting and postprandial concentrations of blood glucose were not different between groups in this study. These results confirm the findings reported by Carroll et al. (2007), which did not find differences between NW and Obese participants in their postprandial concentrations of blood glucose following a liquid meal.

Previous studies that examined the relationship between circulating insulin concentrations and food intake suggested that insulin has an appetite-suppressing effect on NW participants and, to a lesser extent, Obese participants (Verdich et al., 2001b). As expected, insulin concentrations in the current study were higher in the Obese compared to the NW group. These results are consistent with previous studies, which showed greater increases in circulating insulin concentration in the Obese compared to NW participants after carbohydrate intake (van Vliet et al., 2020, Wikarek et al., 2020). Elevated insulin concentrations seen in the Obese group suggests insulin resistance, which may lead to further weight gain (Verdich et al., 2001b, Zyoud et al., 2022). Insulin resistance is believed to be linked to weight gain by increasing appetite and basal lipolysis process (Girousse et al., 2013), and decreasing the thermic effect of food (Camastra et al., 1999).

The processes involved in feeling sated include behavioural and psychological events, metabolic processes and peripheral physiology, as well as metabolic and neurotransmitter interactions in the brain (Dalton et al., 2013). The present study adds to a previous study conducted by Adam and Westerterp-Plantenga (2005), which observed lower satiety scores in the Obese compared with the NW group following a standard breakfast. Lower satiety scores lead to eating more and, hence, can contribute to the development of obesity. With regard to the correlation between GE and satiety, the present study found fullness and CSS were positively correlated with GCV in both groups, which is consistent with previous studies (Marciani et al., 2015, Gonzalez-Izundegui et al., 2021).

In this study, triglycerides and FFA were measured once during the fasting state and there was no difference in FFA between the groups. Fasting plasma triglyceride was higher in the Obese compared with the NW group, which was expected, as it positively correlates with visceral fat area (Marston et al., 2019, Sukkriang et al., 2021) and is associated with insulin resistance. Elevated triglyceride levels could induce leptin resistance, which increases the desire to eat and reduces calorie

134

expenditures unrelated to energy need, may have developed as a signal to the brain that a person is starving (Banks et al., 2004). Hence, hypertriglyceridemia seen in the Obese group could explain the lower satiety feelings that they reported compared to the NW group.

5.6 Strength and limitations of the study

This is the first study that used MR methods to measure the GI functions of GCV, GE, SMA blood flow, and SBWC in NW and Obese participants. It included only male participants to remove the effect of oestrogen during the luteal phase of menstrual cycle as it could delay GE rate (Wang et al., 2015). It used identical test meals for both groups, so the results were directly comparable.

The present study has some limitations as listed below.

- Sample size was calculated based on NW participants, which may be underpowered to detect differences between groups.
- It included only male participants; hence, we cannot apply our findings to females.
- Despite being a research strength, the test meal was the same for all groups instead of being adjusted for body weight. Given that breakfast meals account for 18%–20% of daily energy intake (Trumbo et al., 2002), this could cause some participants to overestimate or underestimate their energy requirements. It is important participants are meeting their energy requirement, as this is necessary for basic metabolic processes such hormone and enzyme production, metabolism, and brain function (de Nava and Raja, 2022).
- Total ghrelin concentrations were measured as opposed to the acylated ghrelin form and total PYY concentrations were examined as opposed to PYY₃₋₃₆. This suggests that variations in total PYY and ghrelin and concentrations may not precisely show variations in PYY₃₋₃₆ and active ghrelin and concentrations. Nevertheless, a substantial correlation exists

between acyl- and total ghrelin (Marzullo et al., 2004). Whereas it was previously believed that only the acylated ghrelin was the active form, it is now believed that unacylated ghrelin also applies some biological activities (Broglio et al., 2004, Asakawa et al., 2005), thereby supporting the idea that total ghrelin is more significant overall. Meanwhile, research indicates that the PYY₁₋₃₆ to PYY₃₋₃₆ ratio is comparable in NW and Obese (le Roux et al., 2006).

 Ad libitum meal intake was not measured, which could have more objectively measured satiety than through the VAS questionnaire, although the VAS questionnaire has been validated and is commonly used to assess satiety feelings (Flint et al., 2000).

5.7 Conclusion

In conclusion, the present study showed that GI functions, gut hormones, and other satiety regulators (except for insulin and triglycerides) were not altered with obesity. The Obese group had higher insulin and triglycerides concentrations and lower satiety feelings compared to the NW group, which could be factors for weight gain. However, the brain plays a major role in regulating food intake and future studies should include brain measurements in response to food intake to allow better understanding of the interactions between gut and brain to tackle obesity.

6 An MRI pilot study to explore gastrointestinal responses to a high fat emulsion drink in people with and without obesity

The gut-brain axis plays a major role in the regulation of food intake. However, the interaction between the brain and gut to regulate appetite and satiety is not yet fully understood, particularly in obesity. Understanding the physiological mechanisms to regulate appetite and satiety could lead to the development of food products that enhance satiety feeling. The imaging team at the SPMIC have developed an MRI protocol that combines gut and brain imaging in a single MRI scan session. The primary objective of the work described in this chapter was to employ this innovative imaging protocol to investigate satiety and appetite responses in individuals with and without obesity, using a high-fat emulsion. The initial work described in this chapter involves the formulation of high-fat emulsion and control beverages with equivalent caloric and viscosity profiles.

6.1 Introduction

The effect of macronutrients on satiety is an important aspect of nutrition and weight management. Different macronutrients, which include carbohydrates, fats, and proteins, can have distinct impacts on satiety due to their various metabolic and physiological effects (Westerterp-Plantenga et al., 2006, Veldhorst et al., 2008, Johnstone, 2013). The satiety effect of fats versus carbohydrates is a subject of ongoing research and can vary depending on several factors, including meal composition and the type of fats and carbohydrates consumed (Yang et al., 2009, Gibbons et al., 2016). Obesity is generally caused by the overconsumption of high caloric food (Mendoza et al., 2007). Several studies reported that people with obesity highly prefer fatty foods compared with normal-weight participants (Dressler and Smith, 2013). This is mainly driven by the hedonic/reward attributes of fat, which plays as key factor in food choice. Neuroimaging studies have demonstrated that taste and texture of oral fat activate taste and reward-related brain

areas (Rolls and McCabe, 2007, Grabenhorst et al., 2010, Eldeghaidy et al., 2011, Stice et al., 2013). These rewarding attributes of fatty food may lead to overconsumption, and could results in weight gain and, in extreme cases, obesity (Golay and Bobbioni, 1997, Bray et al., 2004). Although fatty foods have high caloric content, they have been suggested to have a relatively low satiety effect compared with matched caloric content from protein and carbohydrate (Blundell and MacDiarmid, 1997).

The impact of carbohydrates on satiety can vary depending on their type. Simple carbohydrates, like sugars and refined grains, can lead to rapid changes in blood sugar levels, potentially causing hunger soon after consumption (Bornet et al., 2007, Maki and Phillips, 2015). The consumption of palatable foods high in refined/simple carbohydrate may also be a strong contributor to overeating. Several animal and human studies have reported the activation of the mesolimbic dopaminergic reward pathway in response to sweet intake and, over time, lead to overeating and obesity (Thornley et al., 2011, Thanarajah and Tittgemeyer, 2020). In contrast, complex carbohydrates, especially those rich in fibre, have a more significant satiating effect. They provide sustained energy and reduce overeating by promoting fullness.

Previous MRI studies have investigated the impact of carbohydrate and fat meals on satiety responses. These studies often focus on understanding the brain responds to different carbohydrate-rich or fat-rich foods and how these responses influence feelings of fullness (Eldeghaidy et al., 2016, Al-Zubaidi et al., 2019). Other studies have assessed the GI responses and subjective ratings of satiety (Goetze et al., 2007, Marciani et al., 2015). This study aimed to use advances in MRI methods to combine brain and gut imaging to explore the interactions between the gut and brain in response in NW and Obese participants to a high fat emulsion drink compared to a high carbohydrate drink, that is matched in caloric content and viscosity. Unfortunately, the Covid-19 pandemic significantly disrupted these research activities and, due to the time restrains for the PhD, data available for inclusion in this chapter is only focused on the GI responses to fat and carbohydrate in NW and Obese participants. Hence, no brain data is included in this chapter as there was not enough time to acquire the skills needed to analyse the brain data.

6.2 Aims and hypothesis

Primary aim

To compare GCV and the sensation of postprandial fullness (measured by VAS) between NW and Obese participants following the ingestion of high-fat (HF) and high-carbohydrate (HC) drinks.

Secondary aims

- To compare SBWC and SMA blood flow between NW and Obese participants following the consumption of HF and HC drinks.
- To compare eating behaviours measured by questionnaires between NW and Obese participants.

Hypothesis

This study hypothesises that Obese participants will have a lower GCV compared to NW participants following the consumption of HF drink compared to the HC drink.

6.3 Methods

This section starts by describing the development work on designing the fat emulsion and the carbohydrate drinks, followed by the methods used for the "main MRI study".

6.3.1 <u>Development work on designing the study drinks.</u>

Designing the fat emulsion drink.

The initial design of the HF emulsion drink in this study was based on a previous work by Eldeghaidy et al. (2016) and Eldeghaidy et al. (2011) that measured brain responses and gut hormone levels (gut-brain axis) following the consumption of a high fat emulsion in NW adults. In Eldeghaidy et al. (2016), the 250 ml HF drink

was composed of 22% rapeseed oil, 1% sucrose stearate emulsifier (E-473), and mineral water, in addition to flavours and 9% sucrose. The types of the oil and emulsifier were articularly chosen because of their low odour and taste properties (Miettinen et al., 2002, De Araujo and Rolls, 2004), which is an important factor to consider when designing a brain imaging study. The fat level in the emulsion drink was chosen as it is typically found in high-fat food products such as mayonnaise and salad dressing (Hussein et al., 2015, Eldeghaidy et al., 2016).

In the previous studies of Eldeghaidy et al. (2011) and Eldeghaidy et al. (2016), the HF drink was prepared by mixing sucrose stearate, mineral water and oil using a high-shear blender (Silverson) followed by high-pressure homogenizer (at pressures 500 and 50 bar for the first and second stages respectively) to produce a fat emulsion with a small droplet size $(0.4-\mu m)$ (Hollowood et al., 2008). However, the design of the current study differs from that of Eldeghaidy et al. (2016), as the aim was to combine brain and gut imaging to assess the gastric volume content of the HF drink in NW and Obese participants, in addition to rather than just exploring brain responses of the drinks. Therefore, other considerations in designing the HF drink were taken into account including the GE rate and stability of the HF emulsion. Work by Hussein et al. (2015) demonstrated that a small droplet size (0.4-µm) of 20 % fat emulsion takes a long time (T₅₀: 330 \pm 61 minutes) to empty from the stomach compared with a large (6- μ m) droplet size (T₅₀: 230 ± 22 minutes). A long scan time was not desirable in this study, as it could be uncomfortable for participants, particularly for the Obese group. Hence, in this study the fat emulsion was designed to generate a large $(6-\mu m)$ droplet size. Appendix 10.2.4 gives a full description of the fat emulsion preparation.

Designing carbohydrate drink

One of the aims of this study was to assess the effect of macronutrient on the gutbrain axis, by designing a carbohydrate drink that was matched for caloric content, viscosity, and volume to the HF drink. To match the energy content of the HF drink (593 kcal/300 ml), the HC drink was designed to have 0% fat. HC drink was created with 52% (w/w) maltodextrin, with dextrose equivalent (DE) of 18 (Maltosweet 180, Azelis UK) dissolved in mineral water. Maltodextrin polymers were preferable to glucose monosaccharides they have lower osmolarities than the monosaccharide glucose solution (Brouns and Kovacs, 1997). The maltodextrin DE of 18 was chosen as it has moderate sweetness (Saldivar and Perez-Carrillo, 2016, Muhamad et al., 2018).

However, the generated control drink (HC) was found through sensory testing, by the research team, to be perceived as more viscous compared with the HF drink, and as explained in section 2.2.1.2, the physical characteristics of the meal/drink including viscosity play a significant role in satiety feeling/responses. Therefore, as described below, work was done to match the viscosity of the HF and HC drinks.

6.3.2 Viscosity measurement

In collaboration with colleagues from the food structure research group, viscosity measurements of the HC were carried out using a Physica MCR 301 Rheometer (Figure 6.1) at 25°C per second. Drinks samples of 50 ml were prepared and tested for stress viscosity using a concentric cylinder geometry (CC27) for 5 minutes, and viscosity measurements were acquired at 50 logarithmic ramp shear rates ranging from 0.1 to 100 s⁻¹ and 100 to 0.1 s⁻¹. Measurements were repeated 3 times, and an average of the replicates was calculated. This yielded a viscosity of 61.4 mPa·s at sheer rate of 50 s⁻¹. This shear rate of 50 s⁻¹ has been selected as the matching viscosity point for the tested drinks based on previous studies which suggested 50 s^{-1} represents the shear rate in mouth (Hollowood et al., 2008). To match the viscosity of the HF drink with the HC, five fat emulsion drinks were prepared with different concentrations of the hydroxypropyl methylcellulose (HPMC, BenecelTM, K15M, Ashland, IMCD UK Ltd) thickening agent (0.25%, 0.3%, 0.35%, 0.40% and 0.45%). Measurements were repeated 3 times, and an average of the replicates was calculated and plotted against the viscosity measurement of the HC sample. As shown in Figure 6.2, the HF drink with 0.45% HPMC concentration was found to be the most closely match with the HC drink at the 50 s⁻¹ shear rate. Hence this HMPC concentration was used in the HF drinks.



Figure 6.1. Anoton Parr, Benelux SC-MCR-301-Rheometer.



Figure 6.2. Shows rheology data for 5 high-fat (HF) drinks and the carbohydrate drink (HC drink, maltodextrin). The plots show the matched viscosity profile of the HF emulsion with the addition 0.45% HPMC (hydroxypropyl methylcellulose) and the carbohydrate drink (HC drink, maltodextrin).

However, another factor was taken into consideration in designing the fat emulsion which was the intragastric acid stability of fat emulsion. Previous research showed that the stability of fat emulsion may impact GE, hence satiety sensation (Marciani et al., 2008). Emulsions that remain stable in the stomach's acidic environment empty from the stomach more slowly than emulsions that break up and layer, possibly via enhancing the satiety perception and CCK hormonal response (Marciani et al., 2007, Marciani et al., 2008). Another factor may be that water and fat/cream will empty at different rates when separated. Therefore, fat emulsion stability in the gastric phase was assessed using the in vitro digestion model described below.

6.3.3 In vitro digestion assessment for the fat emulsion stability in the gastric phase

In collaboration with colleagues from the in vitro digestion research group, the stability of the fat emulsion in this study was assessed. An in vitro digestion experiment was undertaken to test the stability of the 22% fat emulsion drink in the gastric phase for a period of 24 hours using the fat emulsion used in Eldeghaidy et al. (2016) with the addition of 0.45% HPMC. In addition, we also wanted to assess the stability of HF using emulsifier and thickening agent used in Hussein et al (2015). Therefore, four samples were assessed using two fat emulsifiers (sucrose stearate and Tween 20) and two thickening agents [HPMC and locust bean gum (LBG)]. Table 6.1 shows the composition of 100 g total solution of the four samples used for in vitro digestion experiment. Samples were prepared as described in Appendix 10.2.4 and 10.2.5. The concentration of the LBG (0.5%) was based on the work of Hussein et al (2015), whereas the concentration of HPMC (0.45%) was based on the viscosity matching of the control drink (as described above).

Table 6.1. Composition of 100 g total solution of the four samples used for in vitro digestion experiment.

Fat emulsion sample	Rapeseed oil (g)	Thickening (g)	Water (g)	Emulsifier (g)
Sucrose + HPMC	22	0.45	76.5	1
Sucrose + LBG	22	0.3	76.7	1
Tween + LBG	22	0.3	76.7	1
Tween + HPMC	22	0.45	76.5	1

Preparation of gastric solutions

To simulate the gastric digestion, an in vitro digestion procedure adapted from INFOGEST static in vitro digestion model was used (Brodkorb et al., 2019). For the purposes of this study only the gastric phase was simulated. Firstly, the simulated gastric fluids (SGF) were prepared according to the compositional Table

6.2 by mixing a series of electrolytes solutions which mimics the composition of the physiological fluids.

			SSF (pH 7)		SGF (pH 3)	
Salt solution	Stock		Milliliters of	Final salt	Milliliters of	Final salt
added	concent	trations	stock added to	concentratio	stock added to	concentration in
			prepare 0.4 L	n in SSF	prepare 0.4 L	SGF
			(1.25×)		(1.25×)	
	(g/L)	(M)	(mL)	(mM)	(mL)	(mM)
KCl	37.3	0.5	15.1	15.1	6.9	6.9
KH2PO4	68	0.5	3.7	3.7	0.9	0.9
NaHCO3a	84	1	6.8	13.6	12.5	25
NaCl	117	2	_	_	11.8	47.2
MgCl2(H2O)6	30.5	0.15	0.5	0.15	0.4	0.12
(NH4)2CO3*	48	0.5	0.06	0.06	0.5	0.5
HC1		6	0.09	1.1	1.3	15.6
CaCl2(H2O)2b	44.1	0.3	0.025	1.5	0.005	0.15

Table 6.2. Volumes of electrolytes solution of the digestion fluids for a volume of 400 mL diluted with water

Gastric Phase: The emulsions were mixed with SGF at two ratios in a 50 ml Falcon tube. The first mixing ratio is 1:2 of emulsion to final digestion mixture (emulsion: SGF) where 10 ml of each emulsion is being mixed with 8 ml of SGF added to the rest of digestion components to achieve a total volume of 20 ml as in Table 6.3. The second is 1:4 where 5 ml of each emulsion is mixed with 13 ml of SGF added to the other digestion components to achieve a total volume of 20 ml as in Table 6.4. To each digestion tube, 1.0 mL of porcine pepsin (EC 3.4.23.1) and 5.0 μ L of CaCl2 to achieve 2000 U /mL and 0.075 mM, respectively in the final digestion mixture. The pH was then reduced to 3 using 1 M HCl by direct measurement and accordingly distilled water was added to complete the volume to 20 ml and samples were returned to the shaking incubator (150 rpm for 2 h at 37°C).

Components	1:2 ratio	1:4 ratio
Liquid food	10 ml	5 ml
Simulated gastric fluid (SGF)	8 ml	13 ml
Pepsin solution	1.00 ml	1.00 ml
0.3 M of CaCl ₂	5 µl	5 µl
HCl to reach pH at 3.0	0.5000 ml	0.5000 ml
Water	0.4950 ml	0.4950 ml
Total sample to be removed for sampling during gastric stage (ml)	0 ml	0 ml
Final gastric mixture volume	20 ml	20 ml

Table 6.3. Gastric phase 1:2 and 1:4 ratios: duration 120 minutes for each ratio.

Stability test

The digested emulsions in each tube at the end of gastric phase were then observed for top layer creaming appearance/separation every 30 minutes for 2 hours and then at 24 hours.

Results

The mixing ratio1:2

Table 6.4 displays the stability of the 4 samples at various time intervals: 30 minutes, 60 minutes, 90 minutes, 120 minutes, and 24 hours. Initially, at 30 minutes, all samples showed stability and there was no appearance of a creaming top layer. The first observation of cream top layer appeared in all samples started at 60 minutes. However, there were variations among the samples. Both Sucrose-LBG and Sucrose-HPMC showed an obvious cream top layer, while Tween-LBG and Tween-HPMC samples displayed partial separation with some appearance of a cream top layer (Table 6.4). The Tween-HPMC showed less creaming in comparison to Tween-LBG. These patterns of separation and creaming remained consistent at 0 minutes and 120 minutes, and even 24 hours.

Table 6.4. Observations of cream top layer separation at 30 minutes, 60 minutes, 90minutes, 120 minutes, and 24 hours in the 4 digested samples with 1:2 mixing ratio.

Thickenin g agent	LBG		НРМС	
	Tween	sucrose	Tween	sucrose
Time scale n	1:2 Twean LBG	12	35 30 25 20 15 10 5	25 26 26 27 20 10 5 10 5 10 5 10 5 10 5 10 5 10 1





The mixing ratio1:4

Table 6.5 shows the stability of the 4 samples at 30 minutes, 60 minutes, 90 minutes, 120 minutes, and 24 hours. The stability pattern for this ratio follows the same trend as the 1:2 ratio.

Based on the results obtained from the in-vitro testing, it is evident that the fat emulsion containing Tween-HPMC demonstrated the highest level of stability at all timepoints for both the 1:2 and 1:4 mixing ratios. As a result, this emulsion was selected for use in this study.

Thicke agent	ening	LBG		НРМС	
emuls	ifier	Tween	sucrose	Tween	sucrose
Time	30 min	1:5% Thean LBG	192 Suc 28G	1.5% ton	H. St Par
scale	60 min	1:52 Trean 1859	154 Suc LBG	1:4 Twenn HPMC	1.41 Suc HPuc

Table 6.5. Observations of cream top layer separation at 30 minutes, 60 minutes, 90minutes, 120 minutes, and 24 hours in the 4 digested samples with 1:4 mixing ratio.



6.4 Main "MRI" study data collection

6.4.1 <u>Study design and ethics</u>

Following the development work on the test drinks, a single blinded MRI study was conducted at SPMIC to investigate the effect of fat and carbohydrate macronutrients on satiety in NW and Obese participants. Ethical approval for the study was obtained from the research ethics committee of the Faculty of Medicine and Health Sciences (Sponsor ref: RGS 19047; IRAS ID: 261806) and health research authority (HRA) (IRAS project ID: 261806).

Prior to conducting the pilot MRI study, taste acceptability of the HF and HC drinks were evaluated. The presence of the Tween 20 emulsifier introduced a notable bitter flavour to the fat emulsion. After experimenting with various combinations of flavour compounds, coffee extract (Nielsen Massey Vanillas, Inc, USA) was found the most effective flavour for masking the bitterness associated with the Tween 20 (SIGMA, Aldrich, Gillingham, U.K.) emulsifier. Consequently, this was used to improve the overall taste of the HF emulsion drink. Tween 20 was also added into the HC drink, in order to match the composition of the HF and HC samples. In addition to the coffee extract, taste acceptability of both drinks was enhanced with the addition of 5 artificial sweeteners (Hermesetas, Switzerland). The energy and macronutrient composition of the drinks is shown in Table 6.6.

Nutrient	HF drink	HC drink
Energy (kcal)	593.3	586.5
Carbohydrate (g)	0.54	147.3
Of which sugar (g)	0	10.8
Of which fibre (g)	0.84	0
Fat (g)	65.7	0
Protein (g)	0	0

Table 6.6. Nutrient information per 300 ml of the high fat (HF) emulsion drink and high carbohydrate (HC) drink.

6.4.2 Eligibility criteria

Healthy participants, ranging in age from 18-45 years old with BMI of normal-body weight (BMI 18.5-24.9 kg/m²) or Obese (BMI 30-40 kg/m²), were recruited by

advertisement. Four participants with obesity were compared with 4 NW participants. All participants were healthy according to the inclusion criteria explained in section 5.3.3. However, this study had further exclusion criteria, as described below.

- Any reported history of neurological disorders
- Abnormal screening procedures including depression measured by Beck's Depression Inventory (Beck et al., 1961) and eating restriction measured by EAT (Garner and Garfinkel, 1979, Garner et al., 1982).
- Following a medically- or self-prescribed diet during the two weeks prior to the pre-study examination and until the end of the study
- Reported weight loss or gain ≥ 10 % of bodyweight during the six months period before the pre-study examination.
- Pregnancy declared by candidate (Pregnancy tests are also available in the female toilets)
- Left-handed assessed by handedness questionnaire. This is to control for brain's lateralisation effects (activation in one side of the brain) that may show variations between left and right-handed participants.

6.4.3 Sample size calculation

This is a pilot study; hence no power calculation was conducted. However, previous studies have demonstrated significant differences in GE due to meal characteristics using a sample size of 8–12 subjects (Hussein et al., 2015, Marciani et al., 2015), therefore we aimed to recruit 11 participants for this study. This sample size has also been found to be suitable for a pilot brain imaging study assessing the gut-brain axis in response to fat intake (Frank-Podlech et al., 2019). While the original plan was to recruit 11 participants in each group (11 NW and 11 Obese, a total of 22 participants), time constrains led to the inclusion of only 8 participants (4 NW and 4 Obese participants) in this thesis.

6.4.4 Study protocol

Participants were invited to attend a total of 3 visits at the SPMIC: a screening visit, and two MRI study visits (approximately one week apart). On the day prior to each

MRI scan day (visits 2 and 3), participants were instructed to consume a non-fatty meal between 19:00 and 20:00 and subsequently fast, consuming only water. Participants were also advised to refrain from alcohol and strenuous exercise for 24 hours before the MRI visits.

Screening visit

Using online questionnaires, participants were initially screened for MRI eligibility, depression, intake of medication, and general health condition. If participants met the eligibility criteria, they were asked to come to the SPMIC, fasted for at least 12 hours for a blood sample collection to measure their fasting lipid levels and hemoglobin A1c (HbA1c), as well as to measure their weight, height, and blood pressure. Participants were also asked to complete online questionnaires to assess their eating behaviours including TFEQ (Stunkard and Messick, 1985), BES (Gormally et al., 1982), PFS (Cappelleri et al., 2009), PrefQuest (Deglaire et al., 2012), and CoEQ (Dalton et al., 2017).

MRI visit 1

Figure 6.3 shows a schematic diagram of the MRI protocol. Participants were asked to come in the morning (7.30 am) to the SPMIC, fasted for at least 12 hours, to be scanned using the 3T wide-bore Philips Ingenia MRI scanner. Baseline/fasted scans (T0) were collected to measure brain (cerebral blood flow, resting-state fMRI, and visual-task fMRI) and gut (GCV, SBWC, and SMA blood flows) measurements. Participants were then given 300 ml of either the HF or HC drinks to consume outside the scanner within 15 minutes in an upright seated position. Following the consumption of the test drinks (T60), participants were scanned up to 120 mins postingestion. Gastrointestinal responses were collected at the following time points: GCV and SBWC at T0, T60, T100, T150 and T180 minutes and SMA blood flow at T0, T40, T70, T90, T110, T140 and T160 minutes.

Brain scans were also collected at T0, T60, T80, T100, and T130. Satiety questionnaires were measured using the VAS scale at T0, T60, T120, and T180 minutes, and blood samples were collected at timepoints T0, T60, T90, T110, T130, T160 and T180 minutes to measure satiety and appetite regulators.



Figure 6.3. MRI study protocol.

MRI visit 2

At least a week after the MRI visit 1, participants were invited to undergo a second MRI scan visit, that was identical to MRI visit 1, but with the different drink. Participants were blinded to the drinks.

6.4.5 <u>Gut imaging protocol</u>

GCV: was measured using a sequence of coronal HASTE acquiring 28 contiguous axial slices with a slice thickness of 7 mm with a 1 mm gap between slices and reconstructed in-plane resolution 0.78 mm \times 0.78 mm and compressed SENSE of 3. The parameters of the sequence were optimised by Dr Caroline Hoad using flip angle 90, TR=707 milliseconds, TE=40 milliseconds. The data were acquired during 1 breath-hold of 20 s.

SBWC: was acquired as previously described (Hoad et al., 2007, Marciani et al., 2013b) using a sequence of coronal single-shot fast spin echo acquiring two stacks of 14 contiguous coronal slices with slice thickness 7 mm with no gaps between slices, reconstructed in-plane resolution 0.78 mm \times 0.78 mm and SENSE of 2.2, flip angle 90, TE=400 ms, TR=1728 ms. The data were acquired during 2 breath holds of 24 s each.

SMA blood flow: was collected using a sequence of phase-contrast MRI angiography with 20 cardiac phases acquiring 14 contiguous coronal slices with slice thickness 6 mm with no gaps between slices, reconstructed in-plane resolution 1.17 mm \times 1.17 mm, and SENSE of 3, flip angle 25°, TE (shortest), TR (shortest),

and velocity encoding 140 cm/s for baseline and 200 cm/s for other time points. The data were acquired during a breath hold of under 20 s and varied in length due to differences in heart rate between participants as it depends on the cardiac cycle for each participant. An MRI-compatible PPU was used to allow cardiac gating for the SMA blood flow scan.

6.4.6 Data analysis

The MRI analysis methods for GCV, SBWC and SMA blood flow were carried out as described previously in section 5.3.7. However, as the GCV was collected for only four timepoints, T₅₀ and GE rate was not calculated in this study due to the small number of points.

6.4.7 Statistical analysis

Type of data presented, and the normality test was conducted as explained in section 5.3.8. To assess the effect of HF and HC macronutrient on GI responses regardless of body weight, data from both NW and Obese participants were combined. A two-way ANOVA for repeated measures was used to analyse the data, with the type of drink and time as the main effects of interactions. If the two-way ANOVA revealed significant differences between the drinks, further analysis was conducted using the Bonferroni post hoc test. The total AUC for postprandial responses following each drink was calculated using the trapezoid rule (Section 5.3.8). The AUC was calculated from 0-180 minutes by the GraphPad Prism for the area under the reading curve down to fasting baseline values. The peaks that were below the baseline value were also considered in the AUC calculation. An unpaired-t-test or Mann-Whitney test was used to test differences between drinks (HF vs. HC) in baseline (fasting) levels and total AUC.

Data analysis was also conducted for each participant group (NW vs. Obese) and for each type of drink (HF vs. HC) was also conducted. However, statistical analysis was deemed inappropriate and would not yield valid results due to the small number of participants included in each group (n=4). Therefore, instead, a simple descriptive analysis is presented (mean and SD, or median and IQR). The next section describes the results of this study. Data form the combined participant's groups, after each drink, is first presented, followed by descriptive results for each NW and Obese group following the HF and HC drinks.

6.5 Results

6.5.1.1 Participant's demographic characteristics

Ninety-five participants contacted the research team and were assessed for eligibility. Twenty-two participants were screened for the study, of which 4 NW and 4 Obese participants were included in this Chapter (Figure 6.4). Table 6.7 summarises the demographic characteristics of the participants.



Figure 6.4. Recruitment flow diagram

Table 6.7. Summary of demographic characteristics of participants. Data are presented
as the mean \pm SD.

	NW group	Obese group
Age (year)	29.5 ± 5.9	35.2 ± 4.9
Gender	All female	2 female & 2 males
BMI (kg/m ²)	22.3 ± 1.1	31.6 ± 0.96

6.5.1.2 Gastric content volume

Good quality images of the HF and HC drinks in the stomach were obtained. Figure 6.5 illustrates an example of gastric scans form a NW participant, whereas Figure 6.6 shows scans from an individual with obesity. Images from the HF drinks show a homogeneous emulsion drink, indicating the stability of fat emulsion in the stomach.

High-fat drink







High-carbohydrate drink



Figure 6.5. Examples of MRI scans of a participant with normal-weight showing the stomach at fasting/baseline and at different time points following the consumption of the high-fat and high-carbohydrate drinks.

High-fat drink





High-carbohydrate drink



Figure 6.6. Examples of MRI images of a participant with obesity showing the stomach at fasting/baseline and at different time points following the consumption of the high-fat and high-carbohydrate drinks.

Figure 6.7 shows the time courses of the median GCV of the HF and HC drinks across the combined NW and Obese groups. At baseline (fasted), GCV did not show significant different between the HF and HC drinks (P= 0.93, Table 6.8). However, as expected, GCV decreased with time for both drinks. While there is a suggestion of higher GCV values following the HF drink (Figure 6.7), this did not reach statistical significance (drink effect, P= 0.11; drink × time interactions, P=0.1). AUC for postprandial GCV did not differ between the HF and HC drinks (Table 6.9).

Table 6.8. Bassline (fasting) levels of gastric content volume (GCV) at the high-fat(HF) and high-carbohydrate (HC) drinks. Data are presented as the median and (IQR).

Fasting levels		HF drink	HC drink	P-value		
	Combined	l participant g	groups			
GCV	(ml)	40 (24)	33 (70)	0.93		
Separate participant groups						
GCV (ml)	NW group	40.5 (83.5)	40 (80.5)	Not applicable		
	Obese group	34 (24)	33 (70.3)			
			``´´			
		1	1	1		

6.5.2


Figure 6.7. The gastric content volume (GCV) over time following the consumption of the high-fat (HF) and high-carbohydrate (HC) drinks for combined participant groups. Values are presented as the median and IQR, n=8 for each drink.

Table 6.9. Total area under curve (AUC) following the consumption of the high-fat (HF) and high-carbohydrate (HC) drinks across the combined groups. Data are presented as the mean± SD.

	HF drink	HC drink	P-value		
MRI measurements					
GCV (ml) from 0-150	1254 ± 188.5	1096 ± 216.9	0.14		
min					
SMA blood flow (ml/s)	57.54 ± 24	40.98 ± 30.75	0.25		
from 0-160 min					
SBWC (ml) from 0-150	1089 ± 632.4	189.3 ± 293.3	0.0026		
min					
Satiety VAS scores from 0-180 min					
DTE (mm)	18.03 ± 58.32	27.27 ± 62.45	0.76		
Hunger (mm)	21.11 ± 60.2	18.4 ± 72.9	0.93		
PFI (mm)	11.5 ± 38.5	15.5 ± 67.4	0.88		
Fullness (mm)	24.2 ± 68.5	25.9 ± 65.9	0.95		
CSS (mm)	8.7 ± 43.2	17.8 ± 55.3	0.72		

CSS, combined satiety scores; DTE, desire to eat; GCV, gastric content volume; PFI, prospective food intake; SBWC, small bowel water content; SMA, superior mesenteric artery; VAS, visual analogue scale.

As a second step, we visually assessed the effect of each macronutrient on the GCV for the NW and Obese groups individually. Figure 6.8 show the time courses of the mean GCV of the HF and HC drinks in the NW and Obese groups. No differences were observed between the drinks or across the groups at baseline (fasted) (Table 6.8). However, as expected, GCV decreased with time for both drinks in the NW and Obese groups. At 120 minutes, both the NW and Obese groups had a similar GCV after the HF drink (379 ± 32 ml vs. 374 ± 44 ml respectively). In terms of the HC drink, GCV showed a lower level in the Obese group following the HC drink

compared to the NW group (Figure 6.8). Table 6.8 illustrates the CCV values for both the fasted state combined across the NW and Obese groups and separated by group.



Figure 6.8. The gastric content volume (GCV) over time following the consumption of the high-fat (HF) and high-carbohydrate (HC) drinks in normal-weight (NW) and Obese participants. Values are presented as the median and IQR, n=4 in each group.

1.1.1.1 Superior mesenteric artery blood flow

Figure 6.9 shows the time courses of the mean SMA blood flow following the HF and HC drinks. At baseline SMA blood flow, no significant difference between the HF and HC drinks (P = 0.69, Table 6.9) was found. However, the SMA flow rate increased immediately after both drinks, with the highest value at 30 minutes for the HF drink, whereas the HC drink showed a decrease in SMA flow levels (Figure 6.9). Although no significant differences were found using the two-way ANOVA (drink effect P = 0.18; drink × time interactions, P = 0.28), there was a higher pattern of SMA flow rate following the consumption of the HF drink compared to the HC drink (Figure 6.9). Interestingly, rates of blood flow at 70 minutes post-drink were similar for the HF and HC drinks (13.11 \pm 2.3 vs. 13.7 \pm 5.9 ml/s, respectively). However, while the blood flow increased following the HF drink at 100 and 120 minutes, it decreased following the HC drink (Figure 6.9).



Figure 6.9. Superior mesenteric artery (SMA) blood flow values over time following the consumption of high-fat (HF) and high-carbohydrate (HC) drinks for combined participant groups. Values are presented as the mean and SD, n=8 for each drink.

Table 6.10. Bassline (fasting) levels of superior mesenteric artery (SMA) blood flow at the high-fat (HF) and high-carbohydrate (HC) drinks.

Fasting levels	HF drink	HC drink	P-value	
Combined participant groups				
SMA blood flow rate (ml/s)	5.9 ±1.6	6.3 ± 1.4	0.69	

(mean ± SD)				
Separate participant groups				
SMA blood flow rate $(m^{1/s})$	NW group	6.3 (3.6)	6.4 (3.7)	Not applicable
[median (IQR)]	Obese group	5.5 (2.7)	5.8 (0.2)	

6.5.3

Figure 6.10 displays the time courses of the mean SMA blood flow following the HF and HC drinks in the NW and Obese groups. Interestingly, the SMA in the Obese group showed highest blood flow immediately after consumption of the HF drink, which may have contributed to the increase in SMA observed in Figure 6.10. Table 6.8 provides an overview of the SMA values for both the fasted state combined across NW and Obese groups and separated by group.



Figure 6.10. Superior mesenteric artery (SMA) blood flow values over time for normalweight (NW) and Obese participants following the consumption of high-fat (HF) and

high-carbohydrate (HC) drinks in the combined groups. Values are presented as the median and IQR, n=4 in each group.

6.5.3.1 Small bowel water content

Figure 6.11 displays the time courses of the mean SBWC following each drink. At baseline (fasted), there was a trend for lower SBWC levels for the HF drink compared to the HC drink (P=0.139, Table 6.11). However, SBWC increased immediately after the consumption of the HF drink and continued to rise until 120 minutes post-drink (Figure 6.11). In contrast, SBWC decreased immediately after the consumption of the HC drink and remained lower until 120 minutes post-drink (Figure 6.11).

Table 6.11. Bassline (fasting) levels of small bowel water content (SBWC) at the high-fat (HF) and high-carbohydrate (HC) drinks.

	HF drink	HC drink	P-value
~			
Combi	ned participant g	groups	
		r	r
ml)	83.8 ± 52.5	146.1 ± 97.4	0.139
SD)			
(
Sanar	to nortiginant g	roung	
Separa	ate participant gi	oups	
NW group	65.5 (110)	102.5 (198.8)	Not applicable
Obese group	67 (85.25)	245 (183)	
Sector Broup	07 (00.20)	(100)	
	Combin ml) SD) Separa NW group Obese group	HF drink Combined participant g ml) 83.8 ± 52.5 SD) Separate participant g NW group 65.5 (110) Obese group 67 (85.25)	HF drinkHC drinkCombined participant groupsml) 83.8 ± 52.5 146.1 ± 97.4 SD)Separate participant groupsNW group65.5 (110) 102.5 (198.8)Obese group 67 (85.25) 245 (183)

The two-way ANOVA test revealed a significant difference in the SBWC between the HF and HC drinks (drink effect, P<0.0001, Figure 6.11). In addition, there was also a significant drink × time interaction (P<0.0001). The AUC for postprandial SBWC was significantly higher for the HF drink compared to the HC drink (P = 0.0026, Table 6.9). Post-hoc analysis showed significantly higher SBWC levels at 0 minutes (P=0.01), 40 minutes (P= 0.002), 90 minutes (P=0.012), and 120 minutes (P=0.0005) following the consumption of the HF drink compared to the HC drink (Figure 6.11).



Figure 6.11. Small bowel water content (SBWC) volume over time following the consumption of the high-fat (HF) and high-carbohydrate (HC) drinks for combined participant groups. Values are presented as the mean and SD, n=8 for each drink.

Figure 6.12 displays the time courses of the mean SBWC following each drink in the NW and Obese groups. No differences were observed at baseline across study groups except for the Obese group at the HC drink visit which had a slightly higher levels than other groups (Figure 6.13). While the SBWC was shown to increase linearly in both groups following the consumption of the HF drink, Obese group seems to show higher SBWC volumes at 120 minutes post-HF drink. Examples of MIP images of the segmented SBWC of NW and Obese participants at baseline and after consumption of the drinks are shown in Figures 6.13 and 6.14. Table 6.11

shows the SBWC values for both the fasted state combined across the NW and Obese groups and separated by group.



Figure 6.12. Small bowel water content (SBWC) volume over time for normal-weight (NW) and Obese participants following the consumption of the high-fat (HF) and high-carbohydrate (HC) drinks. Values are presented as the median and IQR, n=4 in each group.

High-fat drink





High-carbohydrate drink



Figure 6.13. Examples of maximum intensity projections (MIP) images of small bowel water content of a normal-weight (NW) participant at fasting/baseline and at different time points following the consumption of the high-fat and high-carbohydrate drinks.

High-fat drink



High-carbohydrate drink



Figure 6.14. Examples of maximum intensity projections (MIP) images of small bowel water content of a participant with obesity at fasting/ baseline and at different time points following the consumption of the high-fat and high-carbohydrate drink.

6.5.3.2 Subjective satiety

No differences in VAS ratings (desire to eat, hunger, fullness, prospective food) or CSS were found between the HF and HC drinks at baseline/fasted (Table 6.12) or following the drinks, as illustrated in Figures 6.15a, 6.15b, 6.15c, 6.15e.

Table 6.12. Bassline (fasting) levels of the visual analogue scale (VAS) domains and composite satiety scores (CSS).

Fasting levels of VA	S domains	HF	НС	P-value
	Combin	ed participa	nt groups	
DTE (mm)		18.03 ±	27.27 ± 62.45	0.90
$(\text{mean} \pm \text{SD})$		58.32		
Hunger (mm)		21.11 ±	18.4 ± 72.9	0.39
$(\text{mean} \pm \text{SD})$		60.2		
PFI (mm)		11.5 ±	15.5 ± 67.4	0.31
$(\text{mean} \pm \text{SD})$		38.5		
Fullness (mm)		24.2 ±	25.9 ± 65.9	0.50
$(\text{mean} \pm \text{SD})$		68.5		
CSS (mm)		8.7 ± 43.2	17.8 ± 55.3	0.48
$(\text{mean} \pm \text{SD})$				
	Fasting levels	for each par	rticipant group	
DTE (mm)	NW group	48.5±	45.8 ± 15.6	Not applicable
$(\text{mean} \pm \text{SD})$		15.2		
	Obese group	$20.3{\pm}8.8$	31.8 ± 33.8	
Hunger (mm)	NW group	47.5	57 (17)	
[median (IQR)]		(37.25)		
	Obese group	24.5	24 (54.5)	
		(22.25)		
PFI (mm)	NW group	45 (19.5)	46.5 (28.75)	
[median (IQR)]	Obese group	37.5	48 (3.5)	
		(39.75)		
Fullness (mm)	NW group	31.3	38 ± 27.4	
$(mean \pm SD)$		±13.9		

Obese group	$36.3 \pm$	39.5 ± 14.6	
	29.8		
NW group	48.3 ±	47 ± 15.4	
	10.2		
Obese group	62.3 ±	57.8 ± 18.3	
	15.2		
	Obese group NW group Obese group	Obese group $36.3 \pm$ 29.8NW group $48.3 \pm$ 10.2Obese group $62.3 \pm$ 15.2	Obese group $36.3 \pm$ 29.8 39.5 ± 14.6 47 ± 15.4 10.2 NW group $48.3 \pm$ 10.2 47 ± 15.4 57.8 ± 18.3 15.2

CSS, combined satiety scores; DTE, desire to eat; GCV, gastric content volume; PFI, prospective food intake.







Figure 6.15. Subjective satiety ratings over time using Visual Analogue Scale (VAS) scores of (a) desire to eat, (b) hunger, (c) fullness, (d) prospective food intake, and (e) composite satiety scores at fasted and following the consumption of the high-fat (HF) and high-carbohydrate (HC) drinks for combined groups. Values are presented as the mean and SD for desire to eat and hunger, median and IQR for the others, n=8 for each drink.

In terms of each group, there was a pattern toward higher feelings reported by the NW group for desire to eat, hunger, fullness, and prospective food intake following the HC drink compared to the Obese group (Figure 6.16a, 6.16b, and 6.16c). Additionally, CSS was lower for the NW group compared with the Obese group after the HC drinks (Figure 6.16e). No differences were visually observed in fullness ratings between groups and drinks (Figure 6.16d).







Figure 6.16. Subjective satiety ratings over time using Visual Analogue Scale (VAS) scores of (a) desire to eat, (b) hunger, (c) fullness, (d) prospective food intake, and (e) composite satiety scores at fasted and following the consumption of the high-fat (HF) and high-carbohydrate (HC) drinks in normal-weight (NW) and Obese participants. Values are presented as the mean and SD, except for hunger, which is presented as the median and IQR, n=4 in each group.

6.5.4 Eating behaviour questionnaires

6.5.4.1 Three-eating factor questionnaire

Four domains were measured in the TFE questionnaire: restriction, disinhibition, and hunger. As expected, there is a trend for higher scores seen for the Obese group compared with the NW group in all domains and aggregate scores (Figure 6.17).



Figure 6.17. Three-factor eating (TFE) scores in normal-weight (NW) and Obese participants. Values are presented as the median and IQR, n=4 in each group.

6.5.4.2 Control of eating questionnaire

Four domains were measured in this questionnaire: craving control, craving for sweets, craving for savoury and positive mood. The data for this questionnaire was available for analysis from 3 NW and 3 Obese participants. As expected, scores for craving control were higher in the Obese group compared with the NW group (Figure 6.18). Unexpectedly, scores for higher positive mood were seen in the NW group compared with the Obese group. For other domains, no differences appear to find between groups.



Figure 6.18. Control of eating questionnaire (CoEQ) scores in normal-weight (NW) and 2 Obese participants. Values are presented as the mean and SD, n=3 for NW and n=2 for Obese.

6.5.4.3 Binge eating scale

The data for this questionnaire was available for analysis from 3 NW and 3 Obese participants. Surprisingly, the NW group had a higher score than the Obese group (Figure 6.19), though there were very large inter-individual variations.



Figure 6.19. Total scores for binge eating scale in normal-weight (NW) and Obese participants. Values are presented as the mean and SD, n=3 in each group.

6.5.4.4 Power of food scale

Three domains were measured in this questionnaire: food available, food present, food tasted. The data for this questionnaire was available for analysis from 3 NW and 3 Obese participants. There was a trend for higher scores in the NW group compared with the Obese group in all domains and total score (Figure 6.20).



Figure 6.20. Scores for power of food scale in normal-weight (NW) and Obese participants. Values are presented as the median and range, n=3 in each group.

1.1.1.1 Food preference questionnaire

Four domains of food preference were measured: sweet liking, sweet and fat liking, salt liking, and salt and fat liking. Interestingly and unexpectedly, the NW group showed to have higher ratings for sweet liking, sweet and fat liking, and salt liking than the Obese group. No differences appear to show between groups in preference for the salt and fat liking (Figure 6.21).



Figure 6.21. Food preference ratings for normal-weight (NW) and 3 Obese participants. Values are presented as the mean and SD, n=3 in each group.

6.6 Discussion

This study has developed HF and HC drinks that are isoenergic, isovolumic, and isoviscous, thus carefully controlling for key factors that influence appetite and satiety, facilitating the investigation of GI responses to fat and carbohydrate meals/drinks. Due to the low number of participants, the effects of these drinks on GCV, SBWC and SMA blood flow, as well as subjective satiety, were only compared statistically in combined data for NW and Obese participants.

Gastric volume measured immediately after consuming the two test drinks was almost identical. However, although not reaching statistically significance, the gastric volume following the HF drink was observed to be higher when compared with the HC drink. These differences may potentially become statistically significant with a larger sample size, as originally planned with a total of 22 participants. Since the drinks were carefully matched for energy content, the variation in GCV may be related to differing osmolality of the drinks. Previous MRI

study have reported differences in the GE curve for iso-caloric fat and glucose solutions, even when the calorie load was controlled (Goetze et al., 2007). In contrast to the findings of this study, Marciani et al. (2015) reported higher GCV and slower GE rate following the consumption of the HC rice pudding when compared to the HF rice pudding in NW participants. It's important to note that the design of that study was different from the current study, as the calories were not matched across the meals. Additionally, unlike the current study, which assessed liquid meals, the authors of that study evaluated GCV for solid meals.

SBWC revealed significant difference between HF and HC drink. This aligns with Hussein et al. (2015) findings, where the authors showed a significant increase in SBWC following the ingestion of a high fat emulsion (large droplet size with LBG). These observations could be explained by the secretions of pancreases and bile ducts, which are released following fat intake to aid fat digestion and absorption (Phan and Tso, 2001, Mu and Høy, 2004). In contrast to the HF drink, the results from the SBWC showed little changes following the HC drink in both NW and Obese groups. These findings are in agreement to Murray et al. (2014) who reported little changes in SBWC following a glucose drink (400 ml, 40 g glucose, 156 kcal).

SMA blood flow: Studies using ultrasound, have revealed distinct patterns in SMA blood flow response following the consumption of HF meals in comparison to HC meals, with a far more sustained response after fat meals (Sidery et al., 1994). In the current study, a similar difference in SMA blood flow pattern were observed for the HF and HC drinks. These results suggest that dietary fat may stimulates intestinal blood flow more effectively than carbohydrates. However, the results of Qamar and Read (1988) do not align with the findings mentioned above. In their study, isovolumic and isocaloric fat and carbohydrate liquid meals were compared and the SMA blood flow was found to be slower after the fat meal compared to the carbohydrate meal at 5-, 10- and 15-minutes post-ingestion. The current results indicate that the increase in SMA blood flow was more pronounced in the Obese group. However, a larger number of participants is needed to confirm this observation.

The results of VAS ratings between the meals and across the groups were inconclusive. However, previous studies showed conflicting results when comparing subjective satiety rating from fat and carbohydrate drinks. For example, Yang et al. (2009) compared the satiating effects between isoenergic HF (25% carbohydrate, 4% protein and 71% fat) and HC (88% carbohydrate, 4% fat and 8% protein) meals in 12 NW and 15 Obese participants. They found lower satiety VAS scores after HF meals compared to HC meals in both Obese and NW participants. In contrast, Gibbons et al. (2016) reported no differences in hunger and fullness VAS scores in Obese participants after isoenergic, iso-volume HF and HC meals. Further research is needed to determine the impact of carbohydrates and fat on satiety and weight management, considering factors like meal volume and energy density (as discussed in section 2.2.1.2), which will also affect GE rate.

Regarding results for the eating behaviour questionnaires, it was expected that the Obese group would have higher scores in the TEF questionnaires compared to the NW group. This was the finding of a previous study conducted by Jáuregui-Lobera et al. (2014) which showed that Obese participants have higher scores of cognitive restraints than NW participants. A study carried out by Vainik et al. (2019) demonstrated that uncontrolled eating and BMI are typically independent phenomena that interact with one another. Unexpectedly, the NW group had a trend for higher scores for BES, CoEQ (positive mood domain) and PFS than the Obese group. Additionally, NW group reported higher ratings for sweet liking, and sweet and fat liking than the Obese group. These results could be possibly linked to the under-reporting of the Obese group, as previous studies have found instances of misreporting dietary intake among individuals with obesity (Wehling and Lusher, 2019)

6.7 Conclusion

In this pilot study, a high fat emulsion drink was developed with an isoenergic, isovolumic and iso-viscous carbohydrate drink. No definitive conclusion could be made in the outcomes of this study because of a small number of participants

included. However, the present study suggests that a HF meal causes an increase in GCV, SBWC and SMA blood flow compared to HC meal, in both NW and Obese participants. This study is currently continuing, and a larger sample size will enable a more robust assessment of the effect fat and carbohydrate on gastrointestinal responses (GCV, SBWC and SMA blood flow) in NW and Obese participants. A further aim was to investigate the interactions between the brain and the GI tract in relation to food intake. Unfortunately, the brain data was not included in this work as there was not enough time to acquire the skills needed to analyse the data. However once this data is finalised, it will provide valuable understanding of the physiological mechanisms underlying appetite and satiety regulations in normal weight participants and alterations with obesity.

7 Assessing appetite and satiety responses to a high protein drink in adults with obesity and without obesity

This chapter studies the satiety responses to a high protein drink in individuals with and without obesity by using ad-libitum meal intake and subjective satiety rating measures. The study also explores satiety sensations in individuals who were formerly Obese and have lost weight (referred to as Ex-Obese in this chapter). The initial stage of this research involved the development of a high protein drink and a control carbohydrate drink with were matched in viscosity and of similar caloric content. Subsequently, an eating behaviour trial was conducted to evaluate satiety sensations in Obese, Ex-Obese and NW participants.

7.1 Introduction

Hight-protein (HP) meals are defined as 25-81% of a meal's energy from protein (Johnson and Vickers, 1993, Stubbs et al., 1996, Blom et al., 2006, Brennan et al., 2012), whereas high-carbohydrate (HC) meals are defined as 47-88% of a meal's energy from carbohydrate (Blom et al., 2006, Yang et al., 2009, Brennan et al., 2012). It has been suggested that an increase in protein intake may facilitate weight loss (Skov et al., 1999, Farnsworth et al., 2003, Weigle et al., 2005). However, while the intake of higher protein is suggested to reduce satiety and promote weight loss, it is often confounded by carbohydrate content. Several studies have attempted to untangle the effect of dietary protein from carbohydrate by assessing the satiety effects of HP intake and comparing the outcomes with high carbohydrate intake (Blom et al., 2006, Brennan et al., 2012, Ghazzawi and Mustafa, 2019). However, the satiety effects of HP meals versus HC meals remains an ongoing debate. For instance, a study conducted by Barkeling et al. (1990) involving 20 participants with NW found that participants consumed 12% less (P<0.05) of an ad libitum meal, 4hour after a HP lunch (meat casserole, 43 % of total energy from protein) compared to a HC lunch (vegetarian casserole, 69% of total energy from carbohydrate). However, Ghazzawi and Mustafa (2019) reported no differences in any of the VAS ratings at baseline, 30 minutes, and 60 minutes between the HP meal (400 kcal, 51% protein, 36% fat, and 13% carbohydrate) and the HC meal (403 kcal, 9.9% protein, 27% fat and 63% carbohydrate). However, they reported higher VAS ratings at 120 minutes in all VAS ratings following the HP meal when compared to the HC meal in a group of 30 NW participants. Similarly, Blom et al. (2006) showed no significance differences between a HP breakfast (394 kcal, 58.1% protein, and 14.1% carbohydrate) and HC breakfast (389 kcal, 19.3% protein, and 47.3% carbohydrate) meals on satiety VAS ratings and ad libitum energy intake in 15 NW participants.

Contradicting outcomes of the satiety effects of HP vs HC meals have also been reported in people living with obesity. For instance, Brennan et al. (2012) reported less VAS ratings for hunger and ad libitum energy intake after a HP lunch meal (212 kcal, 30% carbohydrate, 45% protein, and 25% fat) compared to a HC lunch meal (213 kcal, 60% carbohydrate, 30% fat, and 10% protein) in 16 Obese participants. However, Witjaksono et al. (2018) reported no differences among low protein/HC (12.4% of protein and 68.2% of carbohydrate), HP/moderate carbohydrate (40.6% protein & 40.2% carbohydrate) and moderate protein/moderate carbohydrate (23.5% protein and 56.9% carbohydrate) meals in a cohort of 22 Obese participants.

The discrepancies observed in previous findings could arise from various factors including the energy density, portion size and/or viscosity of the test meals, rather than solely the macronutrient compositions. The impact of energy density and portion size on satiety has been demonstrated in a study by Kral et al. (2004) involving 39 women with NW. In this study, lunch meals were provided at two energy density levels (5.23 or 7.32 kJ/g) and three portion sizes (500, 700, or 900 g). It was found that higher energy density meal reduced food consumed and increased energy intake at a subsequent meal (P<0.0001). Additionally, larger portion sizes increased food consumed (P<0.03) and energy intake (P<0.0001). The viscosity of the meal also plays an important factor in satiety. Mattes and Rothacker (2001) demonstrated that a low viscous drink led to a greater sensation of hunger (P<0.016) compared to a higher viscous drink with the same macronutrient

composition and energy density. Therefore, when using a HP diet to promote weight loss and maintain body weight, a thorough evaluation of these factors is essential.

Interestingly, no study to date has considered matching viscosity, portion size and energy content of HP and HC meals to assess their satiety effects in individuals with obesity compared to those with NW. In addition, no study has compared the appetite and satiety responses to HP and HC meals in individuals with obesity who have successfully lose weight (Ex-Obese) compared to those with NW adults, or individuals with obesity who have not achieved weight loss. Also, it remains unclear whether satiety responses to HP meal in Ex-Obese individuals would follow the same patterns as in people living with obesity or those with NW.

7.2 Aims and hypothesis

Primary aim

The purpose of this study was to compare the effects of HP vs. HC drinks, which have similar caloric content and are matched in viscosity and volume, on satiety feelings assessed by ad libitum meal intake in Obese, Ex-Obese and NW participants.

Secondary aims

- To compare the effects of HP vs. HC drinks on subjective satiety measured by VAS scale.
- To compare the effects of HP vs. HC drinks on energy intake for the rest of the study days between the study groups.
- To compare 3-day energy and macronutrient intake between the study groups.
- To assess eating behaviour between the study groups measured by TFE, BES and DEBQ.

Hypothesis

People with obesity will have lower satiety feelings compared to NW participants following the consumption of HP, as opposed to the HC drink. Similarly, Ex-Obese will have lower satiety feelings compared to Obese participants who haven't lost weight or NW participants following the consumption of HP, as opposed to the HC drink.

7.3 Methods

7.3.1 <u>Development work on designing the study drinks.</u>

To uncouple the satiety effects of protein from carbohydrate, the study was designed to measure the satiety effects of pure HP and HC drinks rather than mixed macronutrient drinks/meal. This approach provides a clearer insight into the impact of individual macronutrients on satiety sensations.

Designing the high-protein drink

Type of protein: The protein drink in this study was prepared from whey protein. We have chosen whey protein as it is considered a 'fast protein', and evidence suggests it is more satiating than 'slow protein', such as casein, in supressing hunger feelings (Veldhorst et al., 2009) and reducing subsequent food intake (Hall et al., 2003). The casein protein delays GE and slows the postprandial increase in amino acids by coagulating in the acidic environment of the stomach. Whey protein, in contrast, is categorised as a fast protein that causes a rapid and temporary rise in postprandial amino acid levels. Additionally, whey protein appears more efficacious in stimulating secretion of gut hormones, especially gastric inhibitory polypeptide (GIP), GLP-1 and CCK (Hall et al., 2003).

Amount/content of protein: Adults should consume 45% to 65% of their total calories from carbohydrates, 20% to 35% from fat, and 10% to 35% from protein according to the acceptable macronutrient distribution range (Trumbo et al., 2002). Consumption of a high-protein diet (defined as 30% or more from protein of total energy intake) has been shown to induce higher satiety compared with consumption of normal-protein diet (defined as 18% from protein of total energy intake) (Moran et al., 2005, Leidy et al., 2007). Within a single meal, previous studies showed a

significant increase in protein-induced satiety from a single meal containing 25% to 81% protein energy contents (Johnson and Vickers, 1993, Stubbs et al., 1996, Blom et al., 2006, Brennan et al., 2012). Moreover, evidence suggests having a dose of 50 grams per meal or more of protein is necessary to see a significant effect of protein on satiety (Poppitt et al., 1998, Anderson and Moore, 2004, Belza et al., 2013). In this study, the amount of the protein in the test drink was designed based on the upper safe limit of protein intake that can be tolerated by healthy adults without side effects. This was defined as 3.5 g/kg/d divided over 4 meals (Bilsborough and Mann, 2006). The protein amount in this study was calculated at 3.5 g/kg for a 70 kg man, which results in 245 g/day divided across 4 meals (61.2 g/meal), which accounts for 37% of the daily energy intake of 2625 kcal. Based on this, the protein amount was based on 61.6 from whey protein. Seventy grams of Diet Whey Protein Isolate 90 (Protein Works, Chocolate Silk, www.theproteinworks.com) were used to prepare the protein drink which included 255 kcal, 61.6 g of protein (96.6% of the meal's energy), 1.4 g of carbohydrate (2.1% of the meal's energy), and 1.12 g of fat (3.9% of the meal's energy). Two-hundred and thirty millilitres of water was mixed with the protein amount to make 300 ml of the HP drink.

Designing the high carbohydrate drink

To match the energy content of the protein drink, a carbohydrate solution was made of 63 g maltodextrin dissolved in 237 ml of water to make 300 ml of the HC drink. Maltodextrin DE of 18 was chosen as it has moderate sweetness (Saldivar and Perez-Carrillo, 2016, Muhamad et al., 2018). The nutrient composition of the drinks is shown below in Table 7.1.

Table 7.1. Nutrition information of 300 ml of the high-protein (HP) and highcarbohydrate (HC) drink.

Nutrition	НР	НС
Energy (kcal)	255	285
Carbohydrate (g)	1.4	62
Of which sugars (g)	0.56	4.4
Of which dietary fibre (g)	0	5.8

Protein (g)	61.6	1.2
Fat (g)	1.12	1.26
Of which saturated fat (g)	0.56	0.063

Viscosity measurement

As highlighted, one of the main aims of this study is to study the satiety effects of match viscosity HP and HC drink. Viscosity measurements were carried out for the two drinks, in collaboration with colleagues from the Food Science group, using Physica MCR 301 rheometer (Anoton Parr, Benelux) at 25°C per second. Samples of 50 ml were tested for stress viscosity measurement using a concentric cylinder geometry for 5 minutes. Viscosity measurements were acquired at 50 logarithmic ramp shear rates ranging from 0.1 to 100 s⁻¹ and 100 to 0.1 s⁻¹. The initial viscosity/ rheology measurements of the drinks showed that the HP drink is significantly more viscous compared to the HC drink at shear rate of the mouth 50 s⁻¹ (Figure 7.1). To match the viscosity of the two drinks, different concentrations of the HPMC thickening agent were added to the HC drink. As explained in section 6.3.1, HPMC was used as a thickening agent due its low taste and odour properties (Sothornvit, 2009). The concentrations of HPMC were 0.1%, 0.2 %, 0.3%, 0.4%, 0.5, 0.6%, 0.8%, 1.2%, 1.4%, 1.6%, 2.3, 2.7%, 3.2%, 3.6%, and 4%. The HP drink and the fifteen HC drinks with different concentrations of HPMC were tested for viscosity measurements. The composition of the drinks tested for viscosity is shown in Table 7.2. The results showed that the HC drink with a concentration of 1.2% HPMC was the most closely match viscosity for the HP drink at the shear rate of 50 s⁻¹, as shown in Figures 7.2.



Figure 7.1. Shows rheology data for the high-protein (HP) and the carbohydrate drink (HC drink, maltodextrin).

Sample	HPMC (g)	Water (g)	Maltodextrin (g)
HC drink with 0.1 % HPMC	0.1	78.9	21
HC drink with 0.2% HPMC	0.2	78.8	21
HC drink with 0.3% HPMC	0.3	78.7	21
HC drink with 0.4% HPMC	0.4	78.6	21
HC drink with 0.5% HPMC	0.5	78.5	21
HC drink with 0.6 % HPMC	0.6	78.4	21
HC drink with 0.8% HPMC	0.8	78.2	21
HC drink with 1.2% HPMC	1.2	77.8	21
HC drink with 1.4% HPMC	1.4	77.6	21
HC drink with 1.6% HPMC	1.6	77.4	21
HC drink with 2.3 % HPMC	2.3	76.7	21
HC drink with 2.7% HPMC	2.7	76.3	21
HC drink with 3.2 % HPMC	3.2	75.8	21
HC drink with 3.6% HPMC	3.6	75.4	21
HC drink with 4% HPMC	4	75	21

Table 7.2. Composition of 100 g high carbohydrate (HC) drinks tested for viscosity.



Figure 7.2. Shows rheology data for the high-protein (HP) and the carbohydrate drink (HC drink, maltodextrin). The plots show the matched viscosity profile of the HP drink with the addition 1.2% HPMC (hydroxypropyl methylcellulose) and the carbohydrate drink (HC drink, maltodextrin).

7.4 Intervention Study

7.4.1 Methods

7.4.1.1 Study design and ethics

This is a single-blinded, randomised crossover study, where the study participants were not aware of the nature of the test meals. Meals were randomised across participants using research randomizer software (https://www.randomizer.org/). Ethical approval for the study was obtained from the Medical School Research Ethics Committee at the University of Nottingham, reference FMHS 25-0520.

7.4.1.2 Recruitment

The study was advertised using poster. Interested volunteers were given participants' information to read. The local Slimming World Groups were used to advertise for the Ex-Obese group.

7.4.1.3 Eligibility criteria

Inclusion criteria

- Adults aged 18-65, male and female with body mass index (BMI) of:
 - \circ Normal-body weight (BMI 18.5-24.9 kg/m²).
 - Obese (BMI 30-40 kg/m²) who have not lost weight over the past 6 months.
 - Ex-Obese (BMI 25-40 kg/m²) who have lost 5% or more of their initial body weight and have been weight stable for at least the past three months. Successful weight loss is defined as a decrease in body weight of at least 5% of initial body weight (Ramage et al., 2014). This level of weight loss is thought to have beneficial clinical effects on comorbidities such as lipid profile, blood pressure, glycaemic control as well as osteoarthritis and gastroesophageal reflux disease (Lau et al., 2007).
- Able to understand the study requirements.
- Able to give voluntary written informed consent to participate in the study.
- Apparently healthy (judged by health questionnaire, and blood screening): no medical conditions which might affect study measurements.

Exclusion criteria

- Any reported history of metabolic, endocrine, renal disease or gastrointestinal disorders.
- Abnormal screening procedures (including depression and restrictive eating) and laboratory results that are clinically significant, including diabetes, dyslipidemia, impaired renal functions, liver diseases, pancreatitis, or untreated hypertension.
- Under medication (except aspirin/paracetamol), prescribed probiotic or antibiotic treatment in the past 12 weeks.
- Under medication that may have influenced appetite and sensory functioning.

- Following a self-prescribed or medical diet during the two weeks before the pre-study examination and until the end of the study.
- Pregnancy or breastfeeding declared by candidate.
- Smoking.
- No understanding of the written and/or spoken English language.

7.4.1.4 Sample size calculation

The sample size calculation was based on previous data by Blom et al. (2006) on 15 NW men who consumed a high protein breakfast dairy product meal (393 kcal) with an average ad libitum energy intake of 4697 ± 1784 kJ (1118 ± 426 kcal) (mean \pm SD) measured 3 hours after the meal. Employing this in the sample size calculation, 11 participants in each independent group would give 90% power (α = 0.05) to detect a 13% (627 kJ, 150 kcal) difference in the ad libitum energy intake in the Obese group. To allow for an estimated 10% dropout rate, the aim was to recruit 12 participants for each group.

7.4.1.5 Study protocol

Participants were invited to attend 2 study visits, in addition to the screening/consenting visit.

Screening visit

Interested participants were asked to sign an online consent form and to fill out online questionnaires. These questionnaires include a depression scale questionnaire, a health questionnaire, EAT, eating behaviour questionnaires (TEF, BES, and DEBQ) as well as a 3-day food intake diary, using the Intake24online dietary recall survey, designed and validated by Newcastle University (Intake24, 2023). Eligible participants were asked to come to the SPMIC for blood samples to measure full blood count, liver function tests, renal function tests, and HbA1c. Blood samples were collected by a research nurse from the SPMIC. Body weight, height, waist circumference and blood pressure were also measured. Before participants left the SPMIC, they were given a leaflet containing instructions to follow the day before the study visits. These instructions included limiting sporting activity, which may affect satiety.

Study visit 1

Eligible participants were invited to attend a study visit starting at 8:45 am, having fasted for at least 8 hours. They were required to eat their evening meal on the night before the study between 20:00 and 21:00 and have no further food or drink, other than water, after 10 pm. They were further instructed to consume nothing in the morning prior to the study visits. A further reminder of these instructions was provided immediately before the study day. Baseline measurements (T0) of VAS questionnaires were taken at 9:00 am. Following this, participants were instructed to have breakfast consisting of 300 ml of either the HP or HC drink within 15 minutes. These drinks were provided in random order and blinded to the participants. The VAS questionnaire was given to participants right after they had the test drink (T15). Thirty minutes after (T45), satiety and appetite feelings were measured using the VAS and every 30 minutes for the duration of the experiment (three hours until T195). Participants received an ad-libitum test meal at 12:00 pm (T200) and were asked to eat as much as they could in 30 minutes until they felt comfortably full. Participants were asked to note how much they had eaten for the rest of the day using the Intake24 online dietary recall survey (Intake24, 2023).

Study visit 2

At least one week after study visit 1, to allow for the washout period, participants conducted a second study visit that was identical to study visit 1, but in which the second test meal was given.

7.4.1.6 Test drinks

Each volunteer received a 300 ml serving of either a pure protein or pure carbohydrate drinks. The drink was served at room temperature in an opaque paper cup with a straw and a lid. The composition of each drink is detailed in Table 7.1. Cocoa powder was included in both drinks to ensure consistent flavour across the drinks. Appendix 10.2.6 and Appendix 10.2.7 gives a full description of the drinks' preparation. The palatability and flavour of the drinks were tested by the research team before commencing the study.

7.4.1.7 Ad Libitum lunch meal

The ad libitum lunch meal in this study was served 3 hours after the drinks intake to assess satiety.-The ad libitum meal was a standardised tomato-based pasta meal (Horner et al., 2014b, Alhussain et al., 2016). The composition of the meal was as the follow: 250 g of dried white pasta (Sainsbury's Supermarket, UK), 340 g of tomato sauce (Dolmio Bolognese Pasta Sauce, Freepost Mars Food UK, Dublin, UK), 30 g olive oil (Tesco Supermarket, UK), and 80 g of cheddar cheese (Sainsbury's Supermarket, UK). The preparation method of the meal is illustrated in Appendix 10.2.8. The macronutrient composition of the meal was 50% from the energy from carbohydrates, 12.8% from protein, and 36% from fat (Table 7.3).

The pasta was aliquot into ~1000 g portions and stored in the fridge overnight. On the study day, the pasta was heated in a microwave for 2 minutes, stirred hallway through, and then provided to participants straight away with 250 ml water . The pasta plate was always replenished when it was about three-quarters empty, ensuring that there was always enough hot food for participants and that they weren't prompted to stop eating by emptying their plate. Participants were asked to eat as much as they could within 30 minutes until they felt comfortably full. The energy intake was measured by the amount of food consumed.

Nutrition	Per 100 g of ad libitum meal
Energy (kcal)	156.4
Carbohydrate (g)	19.5
Of which sugars (g)	10
Of which dietary fibre	7.5
(g)	
Protein (g)	5
Fat (g)	6.28
Of which saturated fat	2.2
(g)	

Table 7.3. Nutrition information of 100 g of Ad Libitum lunch meal.

7.4.1.8 Subjective satiety

Satiety questionnaires were completed as previously described in section 2.6.1 using a 100 mm VAS scale at baseline, immediately after the drinks (T0), and then every 30 minutes (at 30, 60, 90, 120, 150, and 180 minutes) throughout the experiment (three hours) to assess feelings of DTE, hunger, fullness, and PFI.

7.4.1.9 Satiety quotient

SQ for each VAS domain was calculated to determine the satiety efficiency of the drinks relative to their energy content as discussed earlier in section 2.6.2 (Green et al., 1997).

7.4.1.10 Eating behaviour questionnaires

Eating behavior was assessed by TFEQ, DEBQ, and BES, with more details available in section 2.6.7.

7.4.1.11 Statistical analysis

Type of data presented, and the normality test was conducted as explained in section 5.3.8. A two-way ANOVA was used to test variations in baseline/fasting levels of VAS scales ratings and ad libitum meal intake across the different groups (NW vs. Obese vs. Ex-Obese) with the type of drink and BMI group as the main factors of interaction. A three-way analysis of variance (ANOVA) was used to identify differences in satiety VAS responses for repeated measurements with the type of drink, BMI group and time as the main effects of interactions. If there were significant differences between the groups detected by the two-way or three-way ANOVA analysis, further exploration was carried out using the Bonferroni post hoc test.

The total AUC for postprandial responses was calculated using the trapezoid rule. The postprandial AUC from 0-195 minutes was calculated by the GraphPad Prism for the area under the reading curve down to fasting baseline values. The peaks that were below the baseline value were also considered in the AUC calculation. A twoway ANOVA was used to test differences between groups (NW vs. Obese vs. Ex-Obese) in total AUC after the HP and HC drinks. Unpaired-t-test or Mann-Whitney
test were used to test differences between groups (NW vs. Obese vs. Ex-Obese) for eating behaviour questionnaires and dietary intake.

7.5 Results

7.5.1 Participant's descriptive results

One hundred and thirty participants contacted the research team and were assessed for eligibility. Forty-six participants were screened online for the study, 32 of whom were invited for a screening session. Six participants were excluded after screening (Figure 7.3). Twenty-three participants were included in the final analysis of the study, 12 NW and 11 Obese. Table 7.4 summarises the demographic characteristics of the included participants. Due to the consequences of COVID pandemic and difficulty of recruiting Ex-Obese participants, only two participants in this group were recruited, and were therefore not included in the statistical analysis of the study results. However, some of their individual data are presented in Figure 7.5 and Table 7.6.



Figure 7.3. Recruitment flow cha

Table 7.4. Demographic characteristics of the included participants.

	NW	Obese	Ex-	P-value
			Obese	(NW vs.
				Obese)
Age (mean \pm SD)	24.1 ± 4.3	29.1 ± 6.78	30.5 ± 2.1	0.05
Gender	10 female & 2	8 female &	All	
	males	3 males	female	
Height [cm, median	166 (13.75)	165 (20.1)	160.5 (15)	0.47
(IQR)]				
Weight [kg, median	64 (14.8)	87.3 (17)	69.2 (7.1)	< 0.0001
(IQR)]				
BMI (mean ± SD)	22.2 ± 1.8	33.5 ± 2.6	26.8 ± 1.5	< 0.0001
(kg/m^2)				

Waist	circumference	Not measured	101.5 ±	76 ± 3.2	Not
(mean \pm 5	SD) (cm)		11.48		applicable
Waist-hei	ght ratio (mean	Not measured	0.6 ± 0.06	0.74 ±	Not
± SD)				0.05	applicable

7.5.2 Ad libitum meal intake

The amount of food consumed and energy intake from the ad libitum meal did not differ between the NW and Obese groups for the HP and HC as shown in Table 7.5. No difference was found for the group \times treatment interactions. Individual data for ad libitum meal intake following the HP and HC drinks for the Ex-Obese group are shown in Table 7.6.

Table 7.5. Amount consumed and energy intake of the ad libitum lunch meal. Data are presented as the mean \pm SD.

Outcomes	Group	HP drink	HC drink	Group	Treatment	Group x
						Treatment
Amount	NW	399.5 ±	484.1 ±	0.25	0.59	0.35
consumed (g)		178	246.9			
	Obese	519.3 ±	496.9 ±			
		144.6	183.2			
Energy intake	NW	648.1 ±	757.1 ±	0.31	0.68	0.42
(kcal)		276.7	386.2			
	Obese	812.1±	777.2 ±			
		226.1	286.5			

Table 7.6. Individual data representation for amount consumed and energy intake of the ad libitum lunch meal of the Ex-Obese group.

Outcomes	Participant ID	HP drink	HC drink
Amount consumed (g)	XOB01	97	485
	XOB02	446	457

Energy intake (kcal)	XOB01	151.7	758.5
	XOB02	697.5	714.7

7.5.3 <u>Subjective Satiety</u>

Figure 7.4 shows the subjective satiety sensation measured by the VAS questionnaire. There were no differences in fasting levels in the DTE, PFI, hunger domains and CSS between the NW and Obese groups for the HP and HC drinks (Table 7.7). However, there was a trend for higher baseline fullness for the HC drink condition in the Obese group (P=0.07).

No significant differences were found in DTE, hunger, and PFI domains and CSS between the HP and HC drinks in the NW and Obese groups (Figure 7.4a, 7.4b, 7.4d, and 7.4e). There was a trend for group effect for fullness (P=0.09, Figure 7.4c) which could be explained for higher baseline for the HC drink in the Obese group. No differences were shown for the following interactions: treatment \times group, time \times treatment, group \times time, and group \times time \times treatment for all VAS domains. Total AUC did not differ between the drinks in the NW and Obese groups for all VAS domains and CSS (Table 7.8). Individual data for VAS ratings following the HP and HC drinks for the Ex-Obese group are shown below in Figure 7.6.

	Group	Treatment		P-value		
		HP drink	HC drink	Group	Treatment	Group x
						Treatment
DTE (mm)	NW	51.5 ± 33.7	46.6 ± 27.4	0.32	0.49	0.93
	Obese	60 ± 21.1	53.8 ± 20.9			
Hunger (mm)	NW	50.4 ± 29.9	44.8 ± 29.5	0.43	0.87	0.55
	Obese	51.9 ± 19.6	55.2 ± 17.9			
Fullness (mm)	NW	21.3 ± 14.9	19.1 ± 12.7	0.07	0.35	0.16
	Obese	23.5 ± 21.4	34.6±13.7]		

Table 7.7. Bassline (fasting) levels of the visual analogue scale (VAS) domains and composite satiety scores (CSS). Data are presented as the mean \pm SD.

PFI (mm)	NW	52.5 ± 17.3	51.5 ± 24.6	0.32	0.79	0.93
	Obese	59 ± 20.6	59.9 ± 15.7			
CSS (mm)	NW	42 ± 18.6	44.2±19.1	0.60	0.54	0.84
	Obese	38.3 ± 17.2	42.5 ± 14.7			

DTE, desire to eat; PFI, prospective food intake.



Time (min)



Figure 7.4. Subjective satiety ratings over time using Visual Analogue Scale (VAS) scores of (a) desire to eat, (b) hunger, (c) fullness, (d) prospective food intake, and (e) composite satiety scores at fasted and following the consumption of the high-protein (HP) and high-carbohydrate (HC) drinks in 12 normal-weight (NW) and 11 Obese participants. Values are presented as the mean and SD.

Table 7.8. Total area under curve (AUC) from 0-195 minutes of the visual analogue scale (VAS) domains and composite satiety scores (CSS). Data are presented as the mean \pm SD.

Outcomes	Group	Treatment		P-value		
		HP drink	HC drink	Group	Treatment	Group x
						Treatment
DTE (mm)	NW	76.12	61.05	0.90	0.63	0.74
		± 138.2	± 162.2			
	Obese	95.26	69.5			
		± 132.3	± 145.5			
Hunger	NW	72.49	69.81 ±	0.99	0.68	0.64
(mm)		± 135.8	155.5			
	Obese	53.19 ±	88.47 ±			
		116.5	136.9			
Fullness	NW	90.58 ±	117.20 ±	0.99	0.79	0.34
(mm)		107.2	148.4			
	Obese	128.4 ± 140	80.8 ±			
			121.3			
PFI (mm)	NW	51.2 ± 119.2	60.9 ±	0.72	0.86	0.94
			148.4			
	Obese	68.01 ±	71.69 ±			
		134.9	122.3			
CSS (mm)	NW	68.93	70.66 ±	0.74	0.93	0.89
		± 95.26	124.4			
	Obese	83.98	77.02 ±	1		
		± 115.2	111.2			

DTE, desire to eat; PFI, prospective food intake.





Figure 7.5. Individual values for subjective satiety ratings over time using Visual Analogue Scale (VAS) scores of (a) desire to eat, (b) hunger, (c) fullness, (d) prospective food intake, and (e) composite satiety scores at fasted and following the consumption of the high-protein (HP) and high-carbohydrate (HC) drinks in 2 participants who were obese and lost weight (Ex-Obese group).

7.5.4 Satiety quotient

The SQ for each VAS domain as well as the mean SQ following the HP and HC drinks were analysed. The mean SQ and SQ for each VAS domain did not differ between the NW and Obese groups for the HP and HC drinks (Figure 7.6). The mean SQ for the NW and Obese groups following the two drinks was considered to

have low satiety phenotypes as their mean SQ was lower than 8 mm/100 kcal, as shown in Figure 7.6e.



a Satiety quotient for desire to eat



Figure 7.6. Satiety quotient for (a) desire to eat, (b) hunger, (c) fullness, (d) prospective food intake and (e) mean satiety quotient of the high-protein (HP) and highcarbohydrate (HC) in 12 normal weight participants (NW group) and 11 participants with obesity (Obese group). Values are presented as the mean and SD.

7.5.5 Energy intake for the rest of the study day

This outcome was available for 11 NW and 10 Obese participants. One NW and one Obese participant did not fill out the survey for one study visit, and their data for the other study visit were not included in the analysis. There were no differences

between the NW and Obese groups in energy intake for the rest of the study days following the HP and HC drinks (Table 7.9). Individual data in energy intake for the rest of the study days following the HP and HC drinks for Ex-Obese participants are shown below in Table 7.10.

Table 7.9. Energy intake of the rest of study days following the high-protein (HP) and high-carbohydrate (HC) drinks of the NW and Obese groups. Data are presented as the median and IQR.

Outcomes	Group	HP drink	HC drink	P-value		
	type			Group	Treatment	Group x
						Treatme
						nt
Energy intake		940	1209	0.44	0.99	0.97
(kcal)	IN W	(1144)	(833.1)			
	Ohasa	893 (976)	1070			
	Obese		(648.5)			

Table 7.10. Individual data for energy intake of the rest of study days following thehigh-protein (HP) and high-carbohydrate (HC) drinks of the Ex-Obese group.

Outcomes	Participant ID	HP drink	HC drink
Energy intake (kcal)	XOB01	1071	1435
	XOB02	862	570

7.5.6 Three days average intake of energy and macronutrients

No differences between the NW and Obese groups in the average intake of energy and macronutrients was found, as shown below in Figures 7.7a and 7.7b.



Figure 7.7. Three days average of energy and macronutrient intake of 12 normalweight (NW) participants and 11 participants with obesity (Obese). Values are presented as the median and IQR.

7.5.7 Eating behaviour scores

No differences between the NW and Obese groups in scores of BES (Figure 7.8a) and DEBQ (Figure 7.8b). Similarly, no differences were observed between the groups in TFEQ-Restraint and TFEQ-hunger and TFEQ-Total scores, however,

there was a trend for higher scores in the TFEQ-Disinhibition factor for the Obese group compared with the NW group (P=0.06, Figure 7.8c).



b

Dutch Eating Behaviour Questionnaire





Figure 7.8. Scores for binge eating scale (BES, a), Dutch eating behaviour questionnaire (DEBQ, b) and Three-factor eating (TFE, c) in 12 normal-weight (NW) participants and 11 participants with obesity (Obese, DEBQ only 10). Values are presented as the median and IQR except for DEBQ where values presented as the mean and SD.

7.6 Discussion

This is the first study to compare satiety responses between HP and HC drinks in people with and without obesity, using drinks that were matched for volume and viscosity, and similar in energy content. Protein plays an important role in satiety sensation and could potentially have a significant impact on weight loss and managements. A significant proportion, ranging from 30% to 40% of individuals with obesity who have undergone weight loss treatment have experienced weight regain within one year (Foster et al., 2010), and approximately 51% have returned to their initial weight by the fifth year of their weight-loss treatment (Weintraub et al., 2023). Therefore, understanding satiety patterns in these individuals is crucial for maintaining body weight. While one of the objectives of this chapter was to

evaluate satiety responses to high protein intake in individuals who were Obese and have successfully lost weight (Ex-obese), and to assess whether their satiety responses follow NW or Obese responses, this was unfeasible within the constraints of the current study due to the limited number of participants recruited (only 2 participants). Consequently, the subsequent sections of the discussion primarily focus on the outcomes from the NW and Obese groups.

The results of this study did not reveal significant differences in ad libitum energy intake and VAS scores among the drinks/treatments or across the NW and Obese groups. These findings are consistent with previous study by Blom et al. (2006) where no differences in satiety VAS ratings and ad libitum energy intake were observed between HP and HC isovolumic (400 g), isoviscous, and isoenergic dairy breakfast products: HP (394 kcal, 14.1% carbohydrate, 58.1% protein, 27.8% fat) and HC [389 kcal, 47.3% carbohydrate, 19.3% protein (57.2 g), 33.3% fat] in a group of 15 males with NW. Similarly, another study by Fischer (2004) did not identify differences in energy intake during the ad libitum lunch meal among isoenergic (398 kcal) and isovolumic (400 ml) preloads, including protein (10.5 g dried chicken egg white powder and 94.7 g milk protein), carbohydrate (84.2 g maltodextrin + 10.5 g glucose+ 10.5 g starch rice) and lipid (31.5 g double cream+ 15 g palm oil+ 15 g soyabean oil) in 17 males with NW. Additionally, a study by Potier et al. (2010) did not find differences in satiety VAS ratings between isoenergic and isovolumic (250 ml) preloads consisting of pure macronutrients: protein (243 kcal, 45 g whey and 5 g casein), carbohydrate (245 kcal, 50 g maltodextrin), and lipid (242 kcal, 11.2 g palm oil and 11.2 g soya bean oil). It is important to note that, while the above studies did control for energy and volume contents between the liquid test meals, the viscosity of the meals/drinks was matched only in the study of Blom et al. (2006).

The energy content/density of a meal, as well as the portion size/volume have been demonstrated to independently reduce the amount of food consumed and the subsequent energy intake at a later meal (Kral et al., 2004). In addition to the energy content and volume, meal viscosity is also known to influence satiety. Previous

215

studies have found that increasing the viscosity of a meal can lead to delay GE (Camps et al., 2016) and slow nutrient digestion and absorption, ultimately promoting satiety (Zavoral et al., 1983, Blackburn et al., 1984, Haskell et al., 1992). Therefore, the fact that the viscosity, energy content, and volume were all matched between the test meals may explain the lack of significant differences in satiety sensation between the HP and HC drinks between and across the NW and Obese groups in the present study.

Here, there was an observable trend towards increased fullness over time in the Obese group for both drinks compared to the NW group. This trend might be attributed to the higher baseline fullness observed in the Obese group during the HC drink condition.

The results of the VAS ratings for our study contradict findings from a previous study by Ghazzawi and Mustafa (2019). In their study, they reported higher ratings for desire to eat and prospective food intake and lower fullness after participants consumed a HC breakfast meal (403 kcal, 63% carbohydrate, 9.9% (10 g) protein & 27% fat) when compared to a HP breakfast meal (400 kcal, 13% carbohydrate, 51% (51 g) protein & 36% fat). Another study by Brennan et al. (2012) also found lower VAS ratings for hunger and lower ad libitum energy intake following a HP lunch meal (212 kcal, 30% carbohydrate, 45% (23.3 g) protein and 25% fat) in comparison to a HC lunch meal (213 kcal, 60% carbohydrate, 30% fat and 10% (6.2 g) protein) in the NW and Obese participants. The discrepancies between these and the current study, could be attributed to differences in meal viscosity and the use of solid meal. These distinctions in meal texture might account for the variation in satiety responses, as evidenced by a previous systematic review involving 48 experiments, which revealed that physical texture of a food can significantly impact satiety, with as solid and semi-solid food being more satiating than liquid forms (Almiron-Roig et al., 2013).

Data from the eating behaviour questionnaires including the DEBQ, BES, and TFEQ did not find significant different between the NW and Obese groups. However, there was a noticeable trend towards higher scores in the disinhibition domain for the Obese group, as expected. Disinhibition is the overconsumption of food in response to various stimuli, such as emotional stress, which is accompanied by a lack of control over food intake (Stunkard and Messick, 1985). The increased disinhibition trend in the current study is consistent with a previous review by Vainik et al. (2019), which demonstrated that uncontrolled eating and BMI are typically independent phenomena that can interact with one another.

7.7 Strength and limitations of the study

This is the first study to compare satiety responses between HP and HC drinks in NW and Obese participants while matching, volume, energy content, and viscosity. Unlike other studies, the drinks used in this study were pure macronutrients, allowing better understanding of individual macronutrient effects on satiety.

However, it is important to acknowledge that this study has some limitations which should be considered when interpreting the results. One of the limitations is the absence of data for the Ex-Obese group in the main analysis due to a small number of individuals recruited (only 2 participants). Instead, their individual data for the ad libitum meal intake, VAS ratings and energy intake were presented for these participants. Several factors impacted the recruitment of the Ex-Obese group. Firstly, the study experienced interruptions due to the COVID pandemic and related lockdown measures, with recruitment activities halted from March 2020 until June 2021. Secondly, recruiting participants for this group proved challenging as, although, advertisements were sent to local Slimming World Groups, these groups had been adversely affected by the pandemic, resulting in no participants contacting the research team through this route. Only two participants were recruited through the study advertisement posters placed around the three University's Campuses.

The other limitation of this study is the inclusion of fibre in the HC drink, with methylcellulose added to achieve the required viscosity (1.9 g of fiber/100 ml). Fibre can potentially impact satiety by increasing viscosity, forming gels in the stomach, and fermenting in the GI tract (Slavin and Green, 2007). However, a study conducted by Berthold et al. (2008) measured the effects of six highly cross-linked

cellulose capsules compared to six placebo capsules, administrated 30 minutes before a standardised meal in 19 Obese participants. Their findings demonstrated that supplementation with cross-linked cellulose did not influence GE and satiety sensation.

The study included both male and female participants, which may have introduced variability into the results. Female sex hormones, particularly oestrogens, can influence peripheral and central signals from several hormones involved in the regulation of food intake , including CCK, ghrelin, insulin, and leptin (Asarian and Geary, 2006). These hormones may also play a role in the inhibitory effect of oestrogenic on food consumption during meals. It has been shown that higher energy intake was seen in the luteal phase when compared to the follicular period (Davidsen et al., 2007). It is worth noting that female participants in this study were not recruited during the early phase (follicular period) of menstrual cycle due to logistical challenges in coordinating study visits and availability of the participants. Also, we were unable to conduct separate analysis to assess the impact of gender on satiety responses due to the relatively small sample size in each gender group.

While the use of pure macronutrient drinks in this study provides a significant advantage in understanding the specific effects of individual macronutrients on satiety, it's important to acknowledge that these drinks may not entirely represent the complexity of human diets. In reality, foods people consume are typically combinations of various macronutrients in different proportions.

7.8 Conclusion

This study did not reveal significant differences in ad libitum energy intake, or satiety VAS ratings between the HC and HP drinks among adults with a normalweight and obesity. Similarly, no differences were found in eating behaviour traits between the two groups. However, participants with obesity tended to have higher TFEQ-Disinhibition scores. While subjective satiety rating provides a validated method of measure satiety response, further studies should consider including more objective measures of satiety, such as gut hormones and GE rate, allowing comprehensive understanding of the satiety effect between macronutrients. In addition, other factors that might affect satiety, such age and gender should be considered. Furthermore, investigating satiety responses in the Ex-Obese group holds significant promise, as it can provide valuable insights into effective dietary strategies for weight maintenance, thereby contributing to the prevention of weight regain and the management of the obesity epidemic.

8 General discussion

The work in this PhD thesis aimed to use MRI measurements combined with behavioural and biochemical analysis to investigate and understand the potential mechanisms of the regulation of food intake in normal-weight adults and alterations in people living with obesity. Specific hypotheses to be tested in this thesis were that people with obesity when compared with normal weight participants will show A) an alteration in neurohormonal gut-brain signalling, B) an increase in gastric emptying rate, SBWC, and SMA with associated appetite scores, and C) a change in eating behaviour. These hypotheses were addressed through a systematic review and functional MRI meta-analysis and three eating behaviour studies, two of which used MRI techniques.

Recent advances in MRI methods allowed measuring the physiological responses to food intake in the GI system non-invasively (Carbone et al., 2010, Fruehauf et al., 2011). MRI can capture dynamic changes during the process of GE, making it suitable for studying the process under various conditions. As described in Chapter 2, MRI has been validated against the gold standard methods (scintigraphy) of measuring GE in healthy participants after both liquid and solid meals (Kunz et al., 1999, Teramoto et al., 2012, Khalaf et al., 2020a) and also in patients with gastroparesis (Hayakawa et al., 2017), diabetes (Parkman et al., 2010), and dysphagia (Menys et al., 2017). In this thesis, MRI was employed extensively to investigate and understand the GI responses to food consumption in people with obesity. This is demonstrated in Chapter 5 and Chapter 6. Chapter 5 assessed the GI responses to a standard pasta meal, while Chapter 6 used advances in MRI to assess food intake to different macronutrients (fat vs carbohydrate) with the same energy content, volume, and viscosity. To date, no previous studies have used MRI techniques to compare GI responses following food intake between normal-weight participants and those with obesity.

In terms of GI responses to a pasta meal, no significant differences were found in GCV, GE rate, SBWC and SMA blood flow, appetite, and satiety hormones when

people with obesity were compared with normal-weight adults. This suggests that the immediate GI responses to the meal were similar between Obese and normalweight individuals. Despite the similarities in GI responses, the studies revealed that the Obese group had lower satiety feelings compared to the NW group. This difference in satiety feelings implies that individuals with obesity may experience less satisfaction or fullness after eating, which could potentially lead to overeating and weight gain. In addition to lower satiety feelings, the Obese group displayed higher insulin and triglyceride concentrations. These hormonal differences could be important factors contributing to weight gain in individuals with obesity.

In chapter 6, no definitive conclusion could be made from the outcomes of this study because of the relatively small number of participants included. However, the study does suggest that a HF meal causes an increase in GCV, SBWC, and SMA blood flow compared to a HC meal in both NW and Obese participants. The completion of this work with a larger sample size will enable a more robust assessment of the effect of fat and carbohydrate on GI responses (GCV, SBWC, and SMA blood flow) in NW and Obese participants. Additionally, this study aimed to investigate the interactions between the brain and the GI tract in relation to food intake. Unfortunately, the brain data was not included in this work; however, once this data is finalised, it will provide a valuable understanding of the physiological mechanisms underlying appetite and satiety regulations in normal weight participants and alterations with obesity. Understanding the physiological mechanisms that regulate appetite and satiety could lead to the development of food products that enhance satiety feelings, prevent weight gain, and/or enhance weight loss. It is important to note that despite the great benefits of MRI to assess brain and GI responses, the technique is expensive and may not be readily available in all clinical or research settings. Also, MRI might not be suitable for individuals who are claustrophobic, or have MRI contraindications (i.e., pacemaker, metallic implants).

The work in this thesis also conducted eating behaviour trial to evaluate satiety sensations to acute high-protein (HP) consumption compared to high-carbohydrate

(HC) in Obese, Ex-Obese and NW participants. The results did not reveal significant differences in ad libitum energy intake or subjective satiety ratings between the acute consumption of HC and HP drinks among NW and Obese participants. The study challenges, at least in the short term, the widely held belief that protein is significantly more satiating than carbohydrate. It suggests that the specific form in which macronutrients are presented within food may have a major impact on their satiating effect. In other words, factors beyond macronutrient composition alone, such as the texture and overall composition of the food, can influence satiety. It's important to note that long-term and real-world dietary patterns may yield different results, and further research is needed to explore the complexities of macronutrient satiety in various contexts. Unfortunately, the initial aim of comparing these satiety effects in individuals who had lost weight following obesity was not possible to achieve or complete in this study due to the impact of COVID pandemic on recruitment within this group. Further research is needed for this group.

8.1 Advantages and limitations

The advantages and limitations of each study was discussed in detail in relevant chapters. This section highlights the general advantages and limitations of this PhD work.

However, the main strength on this thesis is the use of a multidisciplinary approach, through utilising and integrating different approaches, to measuring the regulation of food intake, including behavioural assessments (subjective satiety ratings) and physiological measurements obtained through MRI to potentially unravel the mechanisms of alerted eating behaviour in individuals with obesity. This holistic approach facilitates a comprehensive understanding of the impact of macronutrient composition on satiety, hormonal influences, gastrointestinal responses, neural signalling, and eating behaviours in the regulation of food intake.

One of the limitations of this work is the absence of brain MRI measurements in response to meal intake (Chapter 5). The brain plays a major role in regulating food intake, and future studies should include brain measurements in response to food

intake to allow a better understanding of the interactions between the gut and brain to tackle obesity. Understanding these dynamics should contribute to a better comprehension of the metabolic and neural alterations that often accompany obesity.

Participants in Chapter 7 were mostly females, which could have affected satiety responses. Female sex hormones, particularly oestrogens, can influence peripheral and central signals from several hormones involved in the regulation of food intake, including CCK, ghrelin, insulin, and leptin (Asarian and Geary, 2006). These hormones may also play a role in the inhibitory effect of oestrogen on food consumption during meals. On the other hand, the work conducted in Chapter 5 was collected from male-only participants, which cannot be generalised on female participants.

Another limitation in this thesis are small sample sizes which could not allow for more comprehensive comparisons between groups. However, it has been challenging to recruit people with obesity for this thesis for the following reasons. Initially, posters advertising the study were placed around the three University's Campuses, and the majority of participants were typically normal-weight individuals. Second, people with obesity frequently have other medical conditions like diabetes or hypertension, making it difficult to include them in the study. Third, the COVID pandemic had a negative impact on recruiting people with obesity who lost weight since it prevented any participants from contacting the research team via advertisements sent to local Slimming World Groups. Hence, future studies should be designed to recruit larger sample sizes, to allow for more comprehensive comparisons between different groups with altered eating behaviour. This will strengthen the findings and enhance the generalisability of the research.

8.2 **Recommendations for future work**

The need of more a holistic approach to unravel the mechanisms of altered eating behaviour in people with obesity: The functional neuroimaging metaanalysis in this thesis identified the caudate nucleus and hypothalamus as key areas associated with satiety regulators in NW participants. These findings provide valuable information about the neural mechanisms that underlie appetite control in individuals with a normal weight and can be used as a valuable reference point for future studies exploring alterations in gut-brain interactions associated with obesity. Although the work aimed to generate ALE maps for people with obesity, the limited number of studies conducted in this area hindered the ability to draw conclusive findings about the brain regions associated with appetite and satiety regulation in individuals with obesity. This gap in research emphasises the need for further investigation to better understand the neural correlates of appetite control in this population.

Individuals with obesity who have lost weight: A possible extension of this thesis is to investigate satiety responses in individuals with obesity who have lost weight. This area holds significant promise. It can provide valuable insights into effective dietary strategies for weight maintenance, contributing to the prevention of weight regain and the management of obesity. Another aspect to consider is to compare satiety responses between people who have lost weight but are still obese and people who have lost weight and are no longer obese. Previous research revealed that higher basal and postprandial ghrelin concentrations and hunger scores in individuals who lost weight but are still obese compared to prior weight loss (Coutinho et al., 2018, Nymo et al., 2018). However, another study carried out by Hernández Morante et al. (2020) showed no effect of weight loss on ghrelin concentrations in people who remain obese after weight loss. DeBenedictis et al. (2020) assessed satiety responses in individuals who had lost weight but were no longer obese. They found that after losing weight, there was an increase in basal and postprandial ghrelin concentrations as well as hunger scores. These areas need

further exploration to understanding whether alterations in satiety scores that might arise during obesity are reversable following weight loss.

Comparing protein types: In this thesis, behaviour scores to satiety responses were measured following whey protein intake, as evidence suggests it is more satiating than casein in supressing hunger feelings (Veldhorst et al., 2009) and reducing subsequent food intake (Hall et al., 2003). However, it is still inconclusive if one protein is more satiating than others (Bendtsen et al., 2013). Hence, it will be valuable for future studies to compare appetite and satiety responses between whey and casein protein in people living with obesity. Measuring the satiety responses to meals that combine casein and whey with carbohydrates is also another extension.

Larger sample sizes: future studies should be designed to recruit larger sample sizes, to allow for more comprehensive comparisons between different groups with altered eating behaviour. This will strengthen the findings and enhance the generalisability of the research.

A longitudinal study: a future longitudinal MRI study is recommended to assess factors of impaired appetite and satiety mechanisms that contributes to obesity. This study should follow participants from birth to adulthood to evaluate the factors that regulate food intake regulation which include the following: gastric emptying rate, concentrations of appetite hormone (ghrelin) and satiety hormones (PYY, GLP-1, CCK, insulin, leptin) during hunger and fed states, brain responses in homeostatic and hedonic areas, and ratings of subjective satiety feelings. This study should follow body weight status of participants to know if they develop obesity during adulthood. To determine if the factors mentioned above are risk factors or consequences of obesity, they should be assessed both before and after weight gain in participants who will have obesity during adulthood. It is important to take into account additional risk factors for obesity, such as level of physical activity, alcohol consumption, emotional eating, social network, genetics, and medical problems and medications that may increase weight gain.

An aging population: Examining satiety responses in older individuals is another potential extension for this thesis. The "anorexia of ageing" is a common issue, and

understanding how macronutrient intake affects satiety in this population can have implications for nutrition and health in older people (Morley and Silver, 1988). Around 15% to 30% of the elderly have been affected by anorexia of ageing, with rates higher in women, residents of nursing homes, and hospitalised patients (Malafarina et al., 2013). A decrease in appetite can result in a decrease in the amount of food and nutrients consumed (Payette et al., 1995), which raises the possibility of malnutrition and weight loss (Wilson et al., 2005, Brownie, 2006). Elderly people have a higher need for protein intake (1.1 to 1.2 g/kg) to support recovery from illness and improve muscle function and health (Bauer et al., 2013), but they may not usually to get their protein requirements due to the satiating effects of protein. Most studies compare satiety effects of macronutrients focused on young adults, hence, it is unclear if older people have different satiety responses to intake (Dericioglu et al., 2023). Future work should study satiety feelings, GE rate, hormonal, and neural responses following macronutrient intake in older individuals.

8.3 Conclusion

Different studies were conducted in this PhD thesis, including a systematic review and functional MRI meta-analysis and three eating behaviours studies, two of which used MRI techniques. The utilisation and integration of different approaches to measuring the regulation of food intake offer a holistic understanding of food intake regulation. Although the work performed in this thesis did not find significant difference in most of the behavioural and physiological measurements related to regulation of food intake between people with obesity compared to those without obesity, it paved the way for further research opportunities to address the identified limitations from the current work and expand our knowledge in this critical area of nutrition and health.

9 References

- ACAR, F., SEURINCK, R., EICKHOFF, S. B. & MOERKERKE, B. 2018. Assessing robustness against potential publication bias in Activation Likelihood Estimation (ALE) meta-analyses for fMRI. *PLoS One*, 13, e0208177.
- ADAM, T. C. & WESTERTERP-PLANTENGA, M. S. 2005. Glucagon-like peptide-1 release and satiety after a nutrient challenge in normal-weight and obese subjects. *Br J Nutr*, 93, 845-51.
- ADKIN, D. A., GOWLAND, P., SPILLER, R. C., FREEMAN, A., HYKIN, J., ISSA, B., HUCKLE, P. D. & WILDING, I. R. 1995. Echo-planar magnetic resonance imaging to assess water volume in the distal small bowel. *Pharmaceutical research*, 12, 1134-1139.
- ADRIAN, T., BESTERMAN, H., MALLINSON, C., GREENBERG, G. & BLOOM, S. 1979. Inhibition of secretin stimulated pancreatic secretion by pancreatic polypeptide. *Gut*, 20, 37-40.
- ADRIAN, T., BLOOM, S., BRYANT, M., POLAK, J., HEITZ, P. & BARNES, A. 1976. Distribution and release of human pancreatic polypeptide. *Gut*, 17, 940-944.
- ADRIAN, T., SAVAGE, A., SAGOR, G., ALLEN, J., BACARESE-HAMILTON, A., TATEMOTO, K., POLAK, J. & BLOOM, S. 1985a. Effect of peptide YY on gastric, pancreatic, and biliary function in humans. *Gastroenterology*, 89, 494-499.
- ADRIAN, T. E., FERRI, G. L., BACARESE-HAMILTON, A. J., FUESSL, H. S., POLAK, J. M. & BLOOM, S. R. 1985b. Human distribution and release of a putative new gut hormone, peptide YY. *Gastroenterology*, 89, 1070-7.
- AHIMA, R. S. & ANTWI, D. A. 2008. Brain regulation of appetite and satiety. *Endocrinol Metab Clin North Am*, 37, 811-23.
- AHIMA, R. S., PRABAKARAN, D., MANTZOROS, C., QU, D., LOWELL, B., MARATOS-FLIER, E. & FLIER, J. S. 1996. Role of leptin in the neuroendocrine response to fasting. *Nature*, 382, 250-252.
- AL-ZUBAIDI, A., HELDMANN, M., MERTINS, A., BRABANT, G., NOLDE, J. M., JAUCH-CHARA, K. & MUNTE, T. F. 2019. Impact of Hunger, Satiety, and Oral Glucose on the Association Between Insulin and Resting-State Human Brain Activity. *Front Hum Neurosci*, 13, 162.
- ALBERTS, B. 2017. Molecular biology of the cell, Garland science.
- ALHUSSAIN, M. H., MACDONALD, I. A. & TAYLOR, M. A. 2016. Irregular meal-pattern effects on energy expenditure, metabolism, and appetite regulation: a randomized controlled trial in healthy normal-weight women. *Am J Clin Nutr*, 104, 21-32.
- ALLER, E. E. J. G., LARSEN, T. M., CLAUS, H., LINDROOS, A. K., KAFATOS, A., PFEIFFER, A., MARTINEZ, J. A., HANDJIEVA-DARLENSKA, T., KUNESOVA, M., STENDER, S., SARIS, W. H. M., ASTRUP, A. & VAN BAAK, M. A. 2014. Weight loss maintenance in overweight subjects on ad libitum diets with high or low protein content

and glycemic index: the DIOGENES trial 12-month results. *International Journal of Obesity*, 38, 1511-1517.

- ALMIRON-ROIG, E., PALLA, L., GUEST, K., RICCHIUTI, C., VINT, N., JEBB, S. A. & DREWNOWSKI, A. 2013. Factors that determine energy compensation: a systematic review of preload studies. *Nutrition Reviews*, 71, 458-473.
- AMERICAN PSYCHIATRIC ASSOCIATION 2013. *Diagnostic and statistical manual of mental disorders: DSM-5*, American psychiatric association Washington, DC.
- ANDERSON, G. H. & MOORE, S. E. 2004. Dietary proteins in the regulation of food intake and body weight in humans. *J Nutr*, 134, 974s-9s.
- ANDREEVA, E., NEUMANN, M., NÖHRE, M., BRÄHLER, E., HILBERT, A.
 & DE ZWAAN, M. 2019. Validation of the German version of the power of food scale in a general population sample. *Obesity facts*, 12, 416-426.
- ARHIRE, L. I., NIŢĂ, O., POPA, A. D., GAL, A.-M., DUMITRASCU, O. M., GHERASIM, A., MIHALACHE, L. & GRAUR, M. 2021. Validation of the Dutch Eating Behavior Questionnaire in a Romanian Adult Population. *Nutrients*, 13.
- ARMAND, M., PASQUIER, B., ANDRÉ, M., BOREL, P., SENFT, M., PEYROT, J., SALDUCCI, J., PORTUGAL, H., JAUSSAN, V. & LAIRON, D. 1999. Digestion and absorption of 2 fat emulsions with different droplet sizes in the human digestive tract. *Am J Clin Nutr*, 70, 1096-106.
- ARVANITI, K., RICHARD, D. & TREMBLAY, A. 2000. Reproducibility of energy and macronutrient intake and related substrate oxidation rates in a buffet-type meal. *British Journal of Nutrition*, 83, 489-495.
- ASAKAWA, A., INUI, A., FUJIMIYA, M., SAKAMAKI, R., SHINFUKU, N., UETA, Y., MEGUID, M. M. & KASUGA, M. 2005. Stomach regulates energy balance via acylated ghrelin and desacyl ghrelin. *Gut*, 54, 18-24.
- ASAKAWA, A., INUI, A., UENO, N., MAKINO, S., FUJINO, M. A. & KASUGA, M. 1999. Urocortin reduces food intake and gastric emptying in lean and ob/ob obese mice. *Gastroenterology*, 116, 1287-92.
- ASAKAWA, A., INUI, A., YUZURIHA, H., UENO, N., KATSUURA, G., FUJIMIYA, M., FUJINO, M. A., NIIJIMA, A., MEGUID, M. M. & KASUGA, M. 2003. Characterization of the effects of pancreatic polypeptide in the regulation of energy balance. *Gastroenterology*, 124, 1325-36.
- ASARIAN, L. & GEARY, N. 2006. Modulation of appetite by gonadal steroid hormones. *Philosophical Transactions of the Royal Society B: Biological Sciences*, 361, 1251-1263.
- ASTBURY, N. M., TAYLOR, M. A., FRENCH, S. J. & MACDONALD, I. A. 2014. Snacks containing whey protein and polydextrose induce a sustained reduction in daily energy intake over 2 wk under free-living conditions. *Am J Clin Nutr*, 99, 1131-40.
- ASTBURY, N. M., TAYLOR, M. A. & MACDONALD, I. A. 2011. Breakfast consumption affects appetite, energy intake, and the metabolic and

endocrine responses to foods consumed later in the day in male habitual breakfast eaters. *The Journal of nutrition*, 141, 1381-1389.

- BAGGIO, L. L., HUANG, Q., BROWN, T. J. & DRUCKER, D. J. 2004. Oxyntomodulin and glucagon-like peptide-1 differentially regulate murine food intake and energy expenditure. *Gastroenterology*, 127, 546-58.
- BALLEINE, B. W., DELGADO, M. R. & HIKOSAKA, O. 2007. The role of the dorsal striatum in reward and decision-making. *Journal of Neuroscience*, 27, 8161-8165.
- BANKS, W. A., COON, A. B., ROBINSON, S. M., MOINUDDIN, A., SHULTZ, J. M., NAKAOKE, R. & MORLEY, J. E. 2004. Triglycerides induce leptin resistance at the blood-brain barrier. *Diabetes*, 53, 1253-1260.
- BARCLAY, A. W. & BRAND-MILLER, J. 2011. The Australian paradox: a substantial decline in sugars intake over the same timeframe that overweight and obesity have increased. *Nutrients*, 3, 491-504.
- BARDONE-CONE, A. M. & BOYD, C. A. 2007. Psychometric properties of eating disorder instruments in Black and White young women: internal consistency, temporal stability, and validity. *Psychological assessment*, 19, 356.
- BARKELING, B., KING, N. A., NÄSLUND, E. & BLUNDELL, J. E. 2007. Characterization of obese individuals who claim to detect no relationship between their eating pattern and sensations of hunger or fullness. *International Journal of Obesity*, 31, 435-439.
- BARKELING, B., RÖSSNER, S. & BJÖRVELL, H. 1990. Effects of a highprotein meal (meat) and a high-carbohydrate meal (vegetarian) on satiety measured by automated computerized monitoring of subsequent food intake, motivation to eat and food preferences. *Int J Obes*, 14, 743-51.
- BATTERHAM, R. L., COHEN, M. A., ELLIS, S. M., LE ROUX, C. W.,
 WITHERS, D. J., FROST, G. S., GHATEI, M. A. & BLOOM, S. R. 2003.
 Inhibition of food intake in obese subjects by peptide YY3-36. *N Engl J Med*, 349, 941-8.
- BATTERHAM, R. L., COWLEY, M. A., CONE, R. D., BLOOM, S. R., SMALL, C. J., HERZOG, H., COHEN, M. A., DAKIN, C. L., WREN, A. M., BRYNES, A. E., LOW, M. J. & GHATEI, M. A. 2002. Gut hormone PYY3-36 physiologically inhibits food intake. *Nature (London)*, 418, 650-654.
- BATTERHAM, R. L., FFYTCHE, D. H., ROSENTHAL, J. M., ZELAYA, F. O., BARKER, G. J., WITHERS, D. J. & WILLIAMS, S. C. 2007. PYY modulation of cortical and hypothalamic brain areas predicts feeding behaviour in humans. *Nature*, 450, 106-9.
- BATTERHAM, R. L., HEFFRON, H., KAPOOR, S., CHIVERS, J. E., CHANDARANA, K., HERZOG, H., LE ROUX, C. W., THOMAS, E. L., BELL, J. D. & WITHERS, D. J. 2006. Critical role for peptide YY in protein-mediated satiation and body-weight regulation. *Cell Metab*, 4, 223-33.
- BAUER, J., BIOLO, G., CEDERHOLM, T., CESARI, M., CRUZ-JENTOFT, A. J., MORLEY, J. E., PHILLIPS, S., SIEBER, C., STEHLE, P. & TETA, D.

2013. Evidence-based recommendations for optimal dietary protein intake in older people: a position paper from the PROT-AGE Study Group. *Journal of the american Medical Directors association*, 14, 542-559.

- BAYLY, G. R. 2014. CHAPTER 37 Lipids and disorders of lipoprotein metabolism. *In:* MARSHALL, W. J., LAPSLEY, M., DAY, A. P. & AYLING, R. M. (eds.) *Clinical Biochemistry: Metabolic and Clinical Aspects (Third Edition).* Churchill Livingstone.
- BECK, A. T., WARD, C., MENDELSON, M., MOCK, J. & ERBAUGH, J. 1961. Beck depression inventory (BDI). *Arch gen psychiatry*, 4, 561-571.
- BECKER, K. L. 2001. *Principles and practice of endocrinology and metabolism*, Lippincott Williams & Wilkins.
- BEGLINGER, C., DEGEN, L., MATZINGER, D., D'AMATO, M. & DREWE, J. R. 2001. Loxiglumide, a CCK-A receptor antagonist, stimulates calorie intake and hunger feelings in humans. *American Journal of Physiology-Regulatory, Integrative and Comparative Physiology*, 280, R1149-R1154.
- BELLMANN, S., LELIEVELD, J., GORISSEN, T., MINEKUS, M. & HAVENAAR, R. 2016. Development of an advanced in vitro model of the stomach and its evaluation versus human gastric physiology. *Food Research International*, 88, 191-198.
- BELZA, A., RITZ, C., SØRENSEN, M. Q., HOLST, J. J., REHFELD, J. F. & ASTRUP, A. 2013. Contribution of gastroenteropancreatic appetite hormones to protein-induced satiety. *The American of Clinical Nutrition*, 97, 980-989.
- BENDTSEN, L. Q., LORENZEN, J. K., BENDSEN, N. T., RASMUSSEN, C. & ASTRUP, A. 2013. Effect of dairy proteins on appetite, energy expenditure, body weight, and composition: a review of the evidence from controlled clinical trials. *Advances in nutrition*, 4, 418-438.
- BENELAM, B. 2009. Satiation, satiety and their effects on eating behaviour. *Nutrition Bulletin*, 34, 126-173.
- BENOIT, S. C., CLEGG, D. J., SEELEY, R. J. & WOODS, S. C. 2004. Insulin and leptin as adiposity signals. *Recent progress in hormone research*, 59, 267-286.
- BERG, J. K., KIM, E. H., LI, B., JOELSSON, B. & YOUSSEF, N. N. 2014. A randomized, double-blind, placebo-controlled, multiple-dose, parallelgroup clinical trial to assess the effects of teduglutide on gastric emptying of liquids in healthy subjects. *BMC Gastroenterol*, 14, 25.
- BERMUDEZ, O. I. & GAO, X. 2011. Greater consumption of sweetened beverages and added sugars is associated with obesity among US young adults. *Annals of Nutrition and Metabolism*, 57, 211-218.
- BERRIDGE, K. C. 1991. Modulation of taste affect by hunger, caloric satiety, and sensory-specific satiety in the rat. *Appetite*, 16, 103-120.

BERTHOLD, H. K., UNVERDORBEN, S., DEGENHARDT, R., UNVERDORBEN, M. & GOUNI-BERTHOLD, I. 2008. Effect of a Cellulose-containing Weight-loss Supplement on Gastric Emptying and Sensory Functions. *Obesity*, 16, 2272-2280.

- BILOUKHA, O. O. & UTERMOHLEN, V. 2001. Healthy eating in Ukraine: attitudes, barriers and information sources. *Public Health Nutrition*, 4, 207-215.
- BILSBOROUGH, S. & MANN, N. 2006. A review of issues of dietary protein intake in humans. *Int J Sport Nutr Exerc Metab*, 16, 129-52.
- BJERKNES, M. & CHENG, H. 2001. Modulation of specific intestinal epithelial progenitors by enteric neurons. *Proceedings of the National Academy of Sciences*, 98, 12497-12502.
- BLACKBURN, N., REDFERN, J., JARJIS, H., HOLGATE, A., HANNING, I., SCARPELLO, J., JOHNSON, I. & READ, N. 1984. The mechanism of action of guar gum in improving glucose tolerance in man. *Clinical Science (London, England: 1979)*, 66, 329-336.
- BLOM, W. A., LLUCH, A., STAFLEU, A., VINOY, S., HOLST, J. J., SCHAAFSMA, G. & HENDRIKS, H. F. 2006. Effect of a high-protein breakfast on the postprandial ghrelin response. *The American journal of clinical nutrition*, 83, 211-220.
- BLOUET, C. & SCHWARTZ, G. J. 2010. Hypothalamic nutrient sensing in the control of energy homeostasis. *Behav Brain Res*, 209, 1-12.
- BLUDELL, J., LAWTON, C., COTTON, J. & MACDIARMID, J. I. 1996. Control of human appetite: implications for the intake of dietary fat. *Annual review of nutrition*, 16, 285-319.
- BLUNDELL, J., DE GRAAF, C., HULSHOF, T., JEBB, S., LIVINGSTONE, B., LLUCH, A., MELA, D., SALAH, S., SCHURING, E., VAN DER KNAAP, H. & WESTERTERP, M. 2010. Appetite control: methodological aspects of the evaluation of foods. *Obes Rev*, 11, 251-70.
- BLUNDELL, J. & HALFORD, J. 1994. Regulation of nutrient supply: the brain and appetite control. *Proceedings of the Nutrition Society*, 53, 407-418.
- BLUNDELL, J. E. & MACDIARMID, J. I. 1997. Fat as a risk factor for overconsumption: satiation, satiety, and patterns of eating. *J Am Diet Assoc*, 97, S63-9.
- BLUNDELL, J. E., ROGERS, P. J. & HILL, A. J. Evaluating the satiating power of foods: implications for acceptance and consumption. 1987.
- BONNER, J. J., VAJJAH, P., ABDULJALIL, K., JAMEI, M., ROSTAMI-HODJEGAN, A., TUCKER, G. T. & JOHNSON, T. N. 2015. Does age affect gastric emptying time? A model-based meta-analysis of data from premature neonates through to adults. *Biopharm Drug Dispos*, 36, 245-57.
- BORNET, F. R. J., JARDY-GENNETIER, A.-E., JACQUET, N. & STOWELL, J. 2007. Glycaemic response to foods: Impact on satiety and long-term weight regulation. *Appetite*, 49, 535-553.
- BOWEN, J., NOAKES, M. & CLIFTON, P. M. 2006. Appetite Regulatory Hormone Responses to Various Dietary Proteins Differ by Body Mass Index Status Despite Similar Reductions in ad Libitum Energy Intake. *The Journal of Clinical Endocrinology & Metabolism*, 91, 2913-2919.
- BRAY, G. A., PAERATAKUL, S. & POPKIN, B. M. 2004. Dietary fat and obesity: a review of animal, clinical and epidemiological studies. *Physiology & behavior*, 83, 549-555.

- BRENNAN, I. M., LUSCOMBE-MARSH, N. D., SEIMON, R. V., OTTO, B., HOROWITZ, M., WISHART, J. M. & FEINLE-BISSET, C. 2012. Effects of fat, protein, and carbohydrate and protein load on appetite, plasma cholecystokinin, peptide YY, and ghrelin, and energy intake in lean and obese men. Am J Physiol Gastrointest Liver Physiol, 303, G129-40.
- BROGLIO, F., GOTTERO, C., VAN KOETSVELD, P., PRODAM, F., DESTEFANIS, S., BENSO, A., GAUNA, C., HOFLAND, L., ARVAT, E., VAN DER LELY, A. J. & GHIGO, E. 2004. Acetylcholine Regulates Ghrelin Secretion in Humans. *The Journal of Clinical Endocrinology & Metabolism*, 89, 2429-2433.
- BROGNA, A., LORENO, M., CATALANO, F., BUCCERI, A. M.,
 MALAGUARNERA, M., MURATORE, L. A. & TRAVALI, S. 2006.
 Radioisotopic assessment of gastric emptying of solids in elderly subjects.
 Aging Clin Exp Res, 18, 493-6.
- BROUNS, F. & KOVACS, E. 1997. Functional drinks for athletes. *Trends in Food Science & Technology*, 8, 414-421.
- BROWNIE, S. 2006. Why are elderly individuals at risk of nutritional deficiency? *International journal of nursing practice*, 12, 110-118.
- BRUNAULT, P., RABEMAMPIANINA, I., APFELDORFER, G., BALLON, N., COUET, C., RÉVEILLÈRE, C., GAILLARD, P. & EL-HAGE, W. 2015. The Dutch Eating Behavior Questionnaire: Further psychometric validation and clinical implications of the French version in normal weight and obese persons. *La Presse Médicale*, 44, e363-e372.
- BUCHHOLZ, V., BERKENSTADT, H., GOITEIN, D., DICKMAN, R.,
 BERNSTINE, H. & RUBIN, M. 2013. Gastric emptying is not prolonged in obese patients. Surgery for Obesity and Related Diseases, 9, 714-717.
- CALIGARI CONTI, D. 2016. Magnetic Resonance Imaging.
- CALLAHAN, H. S., CUMMINGS, D. E., PEPE, M. S., BREEN, P. A., MATTHYS, C. C. & WEIGLE, D. S. 2004. Postprandial Suppression of Plasma Ghrelin Level Is Proportional to Ingested Caloric Load but Does Not Predict Intermeal Interval in Humans. *The Journal of Clinical Endocrinology & Metabolism*, 89, 1319-1324.
- CAMASTRA, S., BONORA, E., DEL PRATO, S., RETT, K., WECK, M., FERRANNINI, E. & ON BEHALF OF, E. 1999. Effect of obesity and insulin resistance on resting and glucose-induced thermogenesis in man. *International Journal of Obesity*, 23, 1307-1313.
- CAMPFIELD, L. A., SMITH, F. J., ROSENBAUM, M. & HIRSCH, J. 1996. Human eating: evidence for a physiological basis using a modified paradigm. *Neuroscience & Biobehavioral Reviews*, 20, 133-137.
- CAMPOS, A., PORT, J. D. & ACOSTA, A. 2022. Integrative hedonic and homeostatic food intake regulation by the central nervous system: insights from neuroimaging. *Brain Sciences*, 12, 431.
- CAMPS, G., MARS, M., DE GRAAF, C. & SMEETS, P. A. 2016. Empty calories and phantom fullness: a randomized trial studying the relative effects of energy density and viscosity on gastric emptying determined by MRI and satiety. *The American Journal of Clinical Nutrition*, 104, 73-80.

- CANTOR, P. & REHFELD, J. F. 1989. Cholecystokinin in pig plasma: release of components devoid of a bioactive COOH-terminus. *American Journal of Physiology-Gastrointestinal and Liver Physiology*, 256, G53-G61.
- CAPPELLERI, J. C., BUSHMAKIN, A. G., GERBER, R. A., LEIDY, N. K., SEXTON, C. C., KARLSSON, J. & LOWE, M. R. 2009. Evaluating the Power of Food Scale in obese subjects and a general sample of individuals: development and measurement properties. *Int J Obes (Lond)*, 33, 913-22.
- CARBONE, S. F., TANGANELLI, I., CAPODIVENTO, S., RICCI, V. & VOLTERRANI, L. 2010. Magnetic resonance imaging in the evaluation of the gastric emptying and antral motion: Feasibility and reproducibility of a fast not invasive technique. *European journal of radiology*, 75, 212-214.
- CARROLL, J. F., KAISER, K. A., FRANKS, S. F., DEERE, C. & CAFFREY, J. L. 2007. Influence of BMI and gender on postprandial hormone responses. *Obesity (Silver Spring)*, 15, 2974-83.
- CECIL, J., FRANCIS, J. & READ, N. 1998. Relative contributions of intestinal, gastric, oro-sensory influences and information to changes in appetite induced by the same liquid meal. *Appetite*, 31, 377-390.
- CHAUDHRI, O., SMALL, C. & BLOOM, S. 2006a. Gastrointestinal hormones regulating appetite. *Philos Trans R Soc Lond B Biol Sci*, 361, 1187-209.
- CHAUDHRI, O. B., PARKINSON, J. R., KUO, Y. T., DRUCE, M. R., HERLIHY, A. H., BELL, J. D., DHILLO, W. S., STANLEY, S. A., GHATEI, M. A. & BLOOM, S. R. 2006b. Differential hypothalamic neuronal activation following peripheral injection of GLP-1 and oxyntomodulin in mice detected by manganese-enhanced magnetic resonance imaging. *Biochem Biophys Res Commun*, 350, 298-306.
- CHAUDHRI, O. B., WYNNE, K. & BLOOM, S. R. 2008. Can gut hormones control appetite and prevent obesity? *Diabetes care*, 31, S284-S289.
- CHEN, E. Y. & ZEFFIRO, T. A. 2020. Hunger and BMI modulate neural responses to sweet stimuli: fMRI meta-analysis. *Int J Obes (Lond)*, 44, 1636-1652.
- CHEN, H., CHARLAT, O., TARTAGLIA, L. A., WOOLF, E. A., WENG, X., ELLIS, S. J., LAKEY, N. D., CULPEPPER, J., MOORE, K. J., BREITBART, R. E., DUYK, G. M., TEPPER, R. I. & MORGENSTERN, J. P. 1996. Evidence that the diabetes gene encodes the leptin receptor: identification of a mutation in the leptin receptor gene in db/db mice. *Cell*, 84, 491-5.
- CHEN, Y.-C., EDINBURGH, R. M., HENGIST, A., SMITH, H. A., WALHIN, J.-P., BETTS, J. A., THOMPSON, D. & GONZALEZ, J. T. 2018. Venous blood provides lower glucagon-like peptide-1 concentrations than arterialized blood in the postprandial but not the fasted state: Consequences of sampling methods. *Experimental Physiology*, 103, 1200-1205.
- CHOWDHURY, A. H., MURRAY, K., HOAD, C. L., COSTIGAN, C.,
 MARCIANI, L., MACDONALD, I. A., BOWLING, T. E. & LOBO, D. N.
 2016. Effects of Bolus and Continuous Nasogastric Feeding on Gastric
 Emptying, Small Bowel Water Content, Superior Mesenteric Artery Blood

Flow, and Plasma Hormone Concentrations in Healthy Adults: A Randomized Crossover Study. *Ann Surg*, 263, 450-7.

- CHOWDHURY, R. & FORSMARK, C. 2003. Pancreatic function testing. *Alimentary pharmacology & therapeutics*, 17, 733-750.
- CIFUENTES, L. & ACOSTA, A. 2022. Homeostatic regulation of food intake. *Clinics and research in hepatology and gastroenterology*, 46, 101794-101794.
- CLARK, J. T., KALRA, P. S., CROWLEY, W. R. & KALRA, S. P. 1984. Neuropeptide Y and human pancreatic polypeptide stimulate feeding behavior in rats. *Endocrinology*, 115, 427-429.
- COHEN, M. A., ELLIS, S. M., LE ROUX, C. W., BATTERHAM, R. L., PARK,
 A., PATTERSON, M., FROST, G. S., GHATEI, M. A. & BLOOM, S. R.
 2003. Oxyntomodulin suppresses appetite and reduces food intake in humans. *J Clin Endocrinol Metab*, 88, 4696-701.
- COLLIS, J. & NEAVERSON, M. 1967. ARTERIALIZED VENOUS BLOOD: A comparison of pH, Pco2, Po2 and oxygen saturation with that of arterial blood. *British Journal of Anaesthesia*, 39, 883-886.
- CONSIDINE, R. V., SINHA, M. K., HEIMAN, M. L., KRIAUCIUNAS, A., STEPHENS, T. W., NYCE, M. R., OHANNESIAN, J. P., MARCO, C. C., MCKEE, L. J. & BAUER, T. L. 1996. Serum immunoreactive-leptin concentrations in normal-weight and obese humans. *New England Journal* of Medicine, 334, 292-295.
- COSS-ADAME, E. & RAO, S. S. 2014. Brain and gut interactions in irritable bowel syndrome: new paradigms and new understandings. *Curr Gastroenterol Rep,* 16, 379.
- COUTINHO, S. R., REHFELD, J. F., HOLST, J. J., KULSENG, B. & MARTINS, C. 2018. Impact of weight loss achieved through a multidisciplinary intervention on appetite in patients with severe obesity. *American Journal of Physiology-Endocrinology and Metabolism*.
- CRAIG, A. D. 2002. How do you feel? Interoception: the sense of the physiological condition of the body. *Nat Rev Neurosci*, 3, 655-666.
- CUMMINGS, D. E. & OVERDUIN, J. 2007. Gastrointestinal regulation of food intake. *J Clin Invest*, 117, 13-23.
- DALTON, M., FINLAYSON, G., ESDAILE, E. & KING, N. 2013. Appetite, Satiety, and Food Reward in Obese Individuals: A Behavioral Phenotype Approach. *Current Nutrition Reports*, 2, 207-215.
- DALTON, M., FINLAYSON, G., HILL, A. & BLUNDELL, J. 2015. Preliminary validation and principal components analysis of the Control of Eating Questionnaire (CoEQ) for the experience of food craving. *European Journal of Clinical Nutrition*, 69, 1313-1317.
- DALTON, M., FINLAYSON, G., WALSH, B., HALSETH, A. E., DUARTE, C. & BLUNDELL, J. E. 2017. Early improvement in food cravings are associated with long-term weight loss success in a large clinical sample. *International Journal of Obesity*, 41, 1232-1236.

- DAVIDSEN, L., VISTISEN, B. & ASTRUP, A. 2007. Impact of the menstrual cycle on determinants of energy balance: a putative role in weight loss attempts. *Int J Obes (Lond)*, 31, 1777-85.
- DE ARAUJO, I. E., GEHA, P. & SMALL, D. M. 2012. Orosensory and Homeostatic Functions of the Insular Taste Cortex. *Chemosens Percept*, 5, 64-79.
- DE ARAUJO, I. E., GUTIERREZ, R., OLIVEIRA-MAIA, A. J., PEREIRA, A., JR., NICOLELIS, M. A. & SIMON, S. A. 2006. Neural ensemble coding of satiety states. *Neuron*, 51, 483-94.
- DE ARAUJO, I. E. & ROLLS, E. T. 2004. Representation in the human brain of food texture and oral fat. *Journal of Neuroscience*, 24, 3086-3093.
- DE CARVALHO, M. V., CARDOSO, A. G. D. A., FEUERSTEIN, S. C., SOUSA, R. R. D., COLLESE, T. S., TORRES-LEAL, F. L., NASCIMENTO-FERREIRA, M. V. & DE MORAES, A. C. F. 2023. Reliability and validity of the dutch eating behavior questionnaire in an online format for university students from low-income regions in a pandemic context: A 24 hour MESYN study. *Frontiers in Epidemiology*, 2.
- DE GRAAF, C., BLOM, W. A., SMEETS, P. A., STAFLEU, A. & HENDRIKS, H. F. 2004. Biomarkers of satiation and satiety. *The American journal of clinical nutrition*, 79, 946-961.
- DE NAVA, A. S. L. & RAJA, A. 2022. Physiology, Metabolism. *StatPearls* [Internet]. StatPearls Publishing.
- DE SILVA, A., SALEM, V., LONG, C. J., MAKWANA, A., NEWBOULD, R.
 D., RABINER, E. A., GHATEI, M. A., BLOOM, S. R., MATTHEWS, P.
 M. & BEAVER, J. D. 2011. The gut hormones PYY3-36 and GLP-17-36 amide reduce food intake and modulate brain activity in appetite centers in humans. *Cell metabolism*, 14, 700-706.
- DEBENEDICTIS, J. N., NYMO, S., OLLESTAD, K. H., BOYESEN, G. A., REHFELD, J. F., HOLST, J. J., TRUBY, H., KULSENG, B. & MARTINS, C. 2020. Changes in the Homeostatic Appetite System After Weight Loss Reflect a Normalization Toward a Lower Body Weight. J Clin Endocrinol Metab, 105, e2538-46.
- DEGEN, L., DREWE, J., PICCOLI, F., GRANI, K., OESCH, S., BUNEA, R., D'AMATO, M. & BEGLINGER, C. 2007. Effect of CCK-1 receptor blockade on ghrelin and PYY secretion in men. *American Journal of Physiology-Regulatory, Integrative and Comparative Physiology*, 292, R1391-R1399.
- DEGLAIRE, A., MÉJEAN, C., CASTETBON, K., KESSE-GUYOT, E., URBANO, C., HERCBERG, S. & SCHLICH, P. 2012. Development of a questionnaire to assay recalled liking for salt, sweet and fat. *Food Quality and Preference*, 23, 110-124.
- DELGADO-AROS, S., KIM, D. Y., BURTON, D. D., THOMFORDE, G. M., STEPHENS, D., BRINKMANN, B. H., VELLA, A. & CAMILLERI, M. 2002. Effect of GLP-1 on gastric volume, emptying, maximum volume
ingested, and postprandial symptoms in humans. *Am J Physiol Gastrointest Liver Physiol*, 282, G424-31.

- DELLSCHAFT, N., HOAD, C., MARCIANI, L., GOWLAND, P. & SPILLER, R. 2022. Small bowel water content assessed by MRI in health and disease: a collation of single-centre studies. *Aliment Pharmacol Ther*, 55, 327-338.
- DELZENNE, N., BLUNDELL, J., BROUNS, F., CUNNINGHAM, K., DE GRAAF, K., ERKNER, A., LLUCH, A., MARS, M., PETERS, H. P. & WESTERTERP-PLANTENGA, M. 2010. Gastrointestinal targets of appetite regulation in humans. *Obes Rev*, 11, 234-50.
- DERICIOGLU, D., OLDHAM, S., METHVEN, L., SHAFAT, A. & CLEGG, M. E. 2023. Macronutrients effects on satiety and food intake in older and younger adults: A randomised controlled trial. *Appetite*, 189, 106982.
- DHANVANTARI, S., SEIDAH, N. G. & BRUBAKER, P. L. 1996. Role of prohormone convertases in the tissue-specific processing of proglucagon. *Mol Endocrinol*, 10, 342-55.
- DING, H., HU, G. L., ZHENG, X. Y., CHEN, Q., THREAPLETON, D. E. & ZHOU, Z. H. 2015. The method quality of cross-over studies involved in Cochrane Systematic Reviews. *PLoS One*, 10, e0120519.
- DORTON, H. M., LUO, S., MONTEROSSO, J. R. & PAGE, K. A. 2017. Influences of dietary added sugar consumption on striatal food-cue reactivity and postprandial GLP-1 response. *Front Psychiatry*, 8, 297.
- DRAPEAU, V., BLUNDELL, J., GALLANT, A. R., ARGUIN, H., DESPRÉS, J. P., LAMARCHE, B. & TREMBLAY, A. 2013. Behavioural and metabolic characterisation of the low satiety phenotype. *Appetite*, 70, 67-72.
- DRAPEAU, V., KING, N., HETHERINGTON, M., DOUCET, E., BLUNDELL, J. & TREMBLAY, A. 2007. Appetite sensations and satiety quotient: predictors of energy intake and weight loss. *Appetite*, 48, 159-66.
- DRESSLER, H. & SMITH, C. 2013. Food choice, eating behavior, and food liking differs between lean/normal and overweight/obese, low-income women. *Appetite*, 65, 145-152.
- DREWNOWSKI, A., RENDERSON, S. A., DRISCOLL, A. & ROLLS, B. J. 1997. The Dietary Variety Score: Assessing Diet Quality in Healthy Young and Older Adults. *Journal of the American Dietetic Association*, 97, 266-271.
- DUARTE, C., PINTO-GOUVEIA, J. & FERREIRA, C. 2015. Expanding binge eating assessment: Validity and screening value of the Binge Eating Scale in women from the general population. *Eating behaviors*, 18, 41-47.
- DULIŃSKA-LITEWKA, J., ŁAZARCZYK, A., HAŁUBIEC, P., SZAFRAŃSKI, O., KARNAS, K. & KAREWICZ, A. 2019. Superparamagnetic Iron Oxide Nanoparticles—Current and Prospective Medical Applications. *Materials*, 12, 617.
- DUMONT, Y., FOURNIER, A., ST-PIERRE, S. & QUIRION, R. 1995. Characterization of neuropeptide Y binding sites in rat brain membrane preparations using [1251][Leu31,Pro34]peptide YY and [1251]peptide

YY3-36 as selective Y1 and Y2 radioligands. *J Pharmacol Exp Ther*, 272, 673-80.

- DUTTON, E. & DOVEY, T. M. 2016. Validation of the Dutch Eating Behaviour Questionnaire (DEBQ) among Maltese women. *Appetite*, 107, 9-14.
- EBERLEIN, G. A., EYSSELEIN, V. E., SCHAEFFER, M., LAYER, P., GRANDT, D., GOEBELL, H., NIEBEL, W., DAVIS, M., LEE, T. D. & SHIVELY, J. E. 1989. A new molecular form of PYY: structural characterization of human PYY (3–36) and PYY (1–36). *Peptides*, 10, 797-803.
- EICKHOFF, S. B., LAIRD, A. R., GREFKES, C., WANG, L. E., ZILLES, K. & FOX, P. T. 2009. Coordinate-based activation likelihood estimation metaanalysis of neuroimaging data: a random-effects approach based on empirical estimates of spatial uncertainty. *Hum Brain Mapp*, 30, 2907-26.
- EICKHOFF, S. B., NICHOLS, T. E., LAIRD, A. R., HOFFSTAEDTER, F., AMUNTS, K., FOX, P. T., BZDOK, D. & EICKHOFF, C. R. 2016. Behavior, sensitivity, and power of activation likelihood estimation characterized by massive empirical simulation. *Neuroimage*, 137, 70-85.
- ELDEGHAIDY, S., MARCIANI, L., HORT, J., HOLLOWOOD, T., SINGH, G., BUSH, D., FOSTER, T., TAYLOR, A. J., BUSCH, J. & SPILLER, R. C. 2016. Prior consumption of a fat meal in healthy adults modulates the brain's response to fat. *The Journal of nutrition*, 146, 2187-2198.
- ELDEGHAIDY, S., MARCIANI, L., MCGLONE, F., HOLLOWOOD, T., HORT, J., HEAD, K., TAYLOR, A. J., BUSCH, J., SPILLER, R. C., GOWLAND, P. A. & FRANCIS, S. T. 2011. The cortical response to the oral perception of fat emulsions and the effect of taster status. *J Neurophysiol*, 105, 2572-81.
- ELIAS, C. F., LEE, C., KELLY, J., ASCHKENASI, C., AHIMA, R. S., COUCEYRO, P. R., KUHAR, M. J., SAPER, C. B. & ELMQUIST, J. K. 1998. Leptin activates hypothalamic CART neurons projecting to the spinal cord. *Neuron*, 21, 1375-1385.
- ESCRIVÁ-MARTÍNEZ, T., GALIANA, L., RODRÍGUEZ-ARIAS, M. & BAÑOS, R. M. 2019. The binge eating scale: Structural equation competitive models, invariance measurement between sexes, and relationships with food addiction, impulsivity, binge drinking, and body mass index. *Frontiers in Psychology*, 10, 530.
- ESSAH, P. A., LEVY, J. R., SISTRUN, S. N., KELLY, S. M. & NESTLER, J. E. 2007. Effect of macronutrient composition on postprandial peptide YY levels. *J Clin Endocrinol Metab*, 92, 4052-5.
- EVANS, A. C., COLLINS, D. L., MILLS, S. R., BROWN, E. D., KELLY, R. L.
 & PETERS, T. M. 1993. 3D statistical neuroanatomical models from 305 MRI volumes. *IEEE Conference Record Nuclear Science Symposium and Medical Imaging Conference*.
- FARNSWORTH, E., LUSCOMBE, N. D., NOAKES, M., WITTERT, G., ARGYIOU, E. & CLIFTON, P. M. 2003. Effect of a high-protein, energyrestricted diet on body composition, glycemic control, and lipid

concentrations in overweight and obese hyperinsulinemic men and women. *Am J Clin Nutr*, 78, 31-9.

- FAROOQI, I. S., BULLMORE, E., KEOGH, J., GILLARD, J., O'RAHILLY, S. & FLETCHER, P. C. 2007. Leptin regulates striatal regions and human eating behavior. *Science*, 317, 1355.
- FEINLE, C., KUNZ, P., BOESIGER, P., FRIED, M. & SCHWIZER, W. 1999. Scintigraphic validation of a magnetic resonance imaging method to study gastric emptying of a solid meal in humans. *Gut*, 44, 106.
- FLINT, A., GREGERSEN, N. T., GLUUD, L. L., MØLLER, B. K., RABEN, A., TETENS, I., VERDICH, C. & ASTRUP, A. 2007. Associations between postprandial insulin and blood glucose responses, appetite sensations and energy intake in normal weight and overweight individuals: a metaanalysis of test meal studies. *British Journal of Nutrition*, 98, 17-25.
- FLINT, A., RABEN, A., ASTRUP, A. & HOLST, J. J. 1998. Glucagon-like peptide 1 promotes satiety and suppresses energy intake in humans. *J Clin Invest*, 101, 515-20.
- FLINT, A., RABEN, A., BLUNDELL, J. & ASTRUP, A. 2000. Reproducibility, power and validity of visual analogue scales in assessment of appetite sensations in single test meal studies. *International journal of obesity*, 24, 38-48.
- FRANCIS, S. T. & ELDEGHAIDY, S. 2015. Imaging methodologies and applications for nutrition research: what can functional MRI offer? *Proc Nutr Soc*, 74, 89-98.
- FRANK-PODLECH, S., HEINZE, J. M., MACHANN, J., SCHEFFLER, K., CAMPS, G., FRITSCHE, A., ROSENBERGER, M., HINRICHS, J., VEIT, R. & PREISSL, H. 2019. Functional Connectivity Within the Gustatory Network Is Altered by Fat Content and Oral Fat Sensitivity - A Pilot Study. *Front Neurosci*, 13, 725.
- FREITAS, S. R., LOPES, C. S., APPOLINARIO, J. C. & COUTINHO, W. 2006. The assessment of binge eating disorder in obese women: A comparison of the binge eating scale with the structured clinical interview for the DSM-IV. *Eating behaviors*, 7, 282-289.
- FRIED, M., SCHWIZER, W., BEGLINGER, C., KELLER, U., JANSEN, J. & LAMERS, C. 1991. Physiological role of cholecystokinin on postprandial insulin secretion and gastric meal emptying in man. Studies with the cholecystokinin receptor antagonist loxiglumide. *Diabetologia*, 34, 721-726.
- FROMMER, K. W., SCHÄFFLER, A., REHART, S., LEHR, A., MÜLLER-LADNER, U. & NEUMANN, E. 2015. Free fatty acids: potential proinflammatory mediators in rheumatic diseases. *Annals of the rheumatic diseases*, 74, 303-310.
- FRUEHAUF, H., MENNE, D., KWIATEK, M. A., FORRAS-KAUFMAN, Z., KAUFMAN, E., GOETZE, O., FRIED, M., SCHWIZER, W. & FOX, M. 2011. Inter-observer reproducibility and analysis of gastric volume measurements and gastric emptying assessed with magnetic resonance imaging. *Neurogastroenterology and motility*, 23, 854-861.

- FURST, T., CONNORS, M., BISOGNI, C. A., SOBAL, J. & FALK, L. W. 1996. Food Choice: A Conceptual Model of the Process. *Appetite*, 26, 247-266.
- GAO, Q. & HORVATH, T. L. 2008. Neuronal control of energy homeostasis. *FEBS Letters*, 582, 132-141.
- GARFINKEL, P. E. & NEWMAN, A. 2001. The eating attitudes test: twenty-five years later. *Eat Weight Disord*, 6, 1-24.
- GARNER, D. M. & GARFINKEL, P. E. 1979. The Eating Attitudes Test: an index of the symptoms of anorexia nervosa. *Psychol Med*, 9, 273-9.
- GARNER, D. M., OLMSTED, M. P., BOHR, Y. & GARFINKEL, P. E. 1982. The eating attitudes test: psychometric features and clinical correlates. *Psychol Med*, 12, 871-8.
- GAUTIER, J. F., KEWEI, C., SALBE, A. D., BANDY, D., PRATLEY, R. E., HEIMAN, M., RAVUSSIN, E., REIMAN, E. M. & TATARANNI, P. A. 2000. Differential brain responses to satiation in obese and lean men. *Diabetes (New York, N.Y.)*, 49, 838-846.
- GEARY, N. 2004. Endocrine controls of eating: CCK, leptin, and ghrelin. *Physiology & behavior*, 81, 719-733.
- GEARY, N., KISSILEFF, H. R., PI-SUNYER, F. X. & HINTON, V. 1992. Individual, but not simultaneous, glucagon and cholecystokinin infusions inhibit feeding in men. *Am J Physiol*, 262, R975-80.
- GHAZZAWI, H. A. & MUSTAFA, S. 2019. Effect of high-protein breakfast meal on within-day appetite hormones: Peptide YY, glucagon like peptide-1 in adults. *Clinical Nutrition Experimental*, 28, 111-122.
- GIBBONS, C., FINLAYSON, G., CAUDWELL, P., WEBB, D. L., HELLSTRÖM, P. M., NÄSLUND, E. & BLUNDELL, J. E. 2016. Postprandial profiles of CCK after high fat and high carbohydrate meals and the relationship to satiety in humans. *Peptides*, 77, 3-8.
- GIBSON, C. D., CARNELL, S., OCHNER, C. N. & GELIEBTER, A. 2010. Neuroimaging, gut peptides and obesity: novel studies of the neurobiology of appetite. *J Neuroendocrinol*, 22, 833-45.
- GINSBERG, H. N. 2000. Insulin resistance and cardiovascular disease. *The Journal of clinical investigation*, 106, 453-458.
- GIROUSSE, A., TAVERNIER, G., VALLE, C., MORO, C., MEJHERT, N., DINEL, A.-L., HOUSSIER, M., ROUSSEL, B., BESSE-PATIN, A. & COMBES, M. 2013. Partial inhibition of adipose tissue lipolysis improves glucose metabolism and insulin sensitivity without alteration of fat mass. *PLoS biology*, 11, e1001485.
- GLANZ, K., BASIL, M., MAIBACH, E., GOLDBERG, J. & SNYDER, D. A. N. 1998. Why Americans Eat What They Do: Taste, Nutrition, Cost, Convenience, and Weight Control Concerns as Influences on Food Consumption. *Journal of the American Dietetic Association*, 98, 1118-1126.
- GOETZE, O., STEINGOETTER, A., MENNE, D., VAN DER VOORT, I. R., KWIATEK, M. A., BOESIGER, P., WEISHAUPT, D., THUMSHIRN, M., FRIED, M. & SCHWIZER, W. 2007. The effect of macronutrients on gastric volume responses and gastric emptying in humans: A magnetic

resonance imaging study. *Am J Physiol Gastrointest Liver Physiol*, 292, G11-7.

- GOLAY, A. & BOBBIONI, E. 1997. The role of dietary fat in obesity. International journal of obesity and related metabolic disorders: journal of the International Association for the Study of Obesity, 21, S2-11.
- GOLDSTONE, A. P., PRECHTL, C. G., SCHOLTZ, S., MIRAS, A. D.,
 CHHINA, N., DURIGHEL, G., DELIRAN, S. S., BECKMANN, C.,
 GHATEI, M. A., ASHBY, D. R., WALDMAN, A. D., GAYLINN, B. D.,
 THORNER, M. O., FROST, G. S., BLOOM, S. R. & BELL, J. D. 2014.
 Ghrelin mimics fasting to enhance human hedonic, orbitofrontal cortex,
 and hippocampal responses to food. *Am J Clin Nutr*, 99, 1319-1330.
- GONZALEZ-IZUNDEGUI, D., CAMPOS, A., CALDERON, G., RICARDO-SILGADO, M. L., CIFUENTES, L., DECKER, P. A., VARGAS, E. J., TRAN, L., BURTON, D., ABU DAYYEH, B., CAMILLERI, M., ECKEL-PASSOW, J. E. & ACOSTA, A. 2021. Association of gastric emptying with postprandial appetite and satiety sensations in obesity. *Obesity (Silver Spring)*, 29, 1497-1507.
- GONZALEZ-IZUNDEGUI, D., CAMPOS, A., CALDERON, G., RICARDO-SILGADO, M. L., CIFUENTES, L., DECKER, P. A., VARGAS, E. J., TRAN, L., BURTON, D. & ABU DAYYEH, B. 2021. Association of gastric emptying with postprandial appetite and satiety sensations in obesity. *Obesity*, 29, 1497-1507.
- GORMALLY, J., BLACK, S., DASTON, S. & RARDIN, D. 1982. The assessment of binge eating severity among obese persons. *Addictive behaviors*, 7, 47-55.
- GOTTFRIED, J. A., O'DOHERTY, J. & DOLAN, R. J. 2003. Encoding predictive reward value in human amygdala and orbitofrontal cortex. *Science*, 301, 1104-7.
- GOYAL, R. K., GUO, Y. & MASHIMO, H. 2019. Advances in the physiology of gastric emptying. *Neurogastroenterology & Motility*, 31, e13546.
- GRABENHORST, F., ROLLS, E. T., PARRIS, B. A. & D'SOUZA, A. A. 2010. How the brain represents the reward value of fat in the mouth. *Cereb Cortex*, 20, 1082-91.
- GRANDT, D., SCHIMICZEK, M., STRUK, K., SHIVELY, J., EYSSELEIN, V. E., GOEBELL, H. & REEVE, J. R., JR. 1994. Characterization of two forms of peptide YY, PYY(1-36) and PYY(3-36), in the rabbit. *Peptides*, 15, 815-20.
- GRANGER, N. D., HOLM, L. & KVIETYS, P. 2011. The gastrointestinal circulation: physiology and pathophysiology. *Comprehensive Physiology*, 5, 1541-1583.
- GREEN, S., DELARGY, H., JOANES, D. & BLUNDELL, J. 1997. A satiety quotient: a formulation to assess the satiating effect of food. *Appetite*, 29, 291-304.
- GREGERSEN, N. T., FLINT, A., BITZ, C., BLUNDELL, J. E., RABEN, A. & ASTRUP, A. 2008. Reproducibility and power of ad libitum energy intake

assessed by repeated single meals. *The American journal of clinical nutrition*, 87, 1277-1281.

- GRIMES, C. A., RIDDELL, L. J., CAMPBELL, K. J. & NOWSON, C. A. 2013. Dietary salt intake, sugar-sweetened beverage consumption, and obesity risk. *Pediatrics*, 131, 14-21.
- GRUNDY, S. M. 1999. Hypertriglyceridemia, insulin resistance, and the metabolic syndrome. *The American journal of cardiology*, 83, 25-29.
- GRUPSKI, A. E., HOOD, M. M., HALL, B. J., AZARBAD, L., FITZPATRICK, S. L. & CORSICA, J. A. 2013. Examining the Binge Eating Scale in screening for binge eating disorder in bariatric surgery candidates. *Obesity surgery*, 23, 1-6.
- GUAN, X., KARPEN, H. E., STEPHENS, J., BUKOWSKI, J. T., NIU, S.,
 ZHANG, G., STOLL, B., FINEGOLD, M. J., HOLST, J. J. & HADSELL,
 D. L. 2006. GLP-2 receptor localizes to enteric neurons and endocrine
 cells expressing vasoactive peptides and mediates increased blood flow. *Gastroenterology*, 130, 150-164.
- GUIDUCCI, L., JÄRVISALO, M., KISS, J., NÅGREN, K., VILJANEN, A., NAUM, A. G., GASTALDELLI, A., SAVUNEN, T., KNUUTI, J. & SALVADORI, P. A. 2006. [11C] palmitate kinetics across the splanchnic bed in arterial, portal and hepatic venous plasma during fasting and euglycemic hyperinsulinemia. *Nuclear medicine and biology*, 33, 521-528.
- GUYTON, A. C. 2006. Text book of medical physiology, China.
- HALAWI, H., CAMILLERI, M., ACOSTA, A., VAZQUEZ-ROQUE, M., ODUYEBO, I., BURTON, D., BUSCIGLIO, I. & ZINSMEISTER, A. R. 2017. Relationship of gastric emptying or accommodation with satiation, satiety, and postprandial symptoms in health. *American Journal of Physiology-Gastrointestinal and Liver Physiology*, 313, G442-G447.
- HALL, W. L., MILLWARD, D. J., LONG, S. J. & MORGAN, L. M. 2003. Casein and whey exert different effects on plasma amino acid profiles, gastrointestinal hormone secretion and appetite. *British Journal of Nutrition*, 89, 239-248.
- HANSEN, T. T., SJÖDIN, A., RITZ, C., BONNET, S. & KORNDAL, S. K. 2018. Macronutrient manipulations of cheese resulted in lower energy content without compromising its satiating capacity. J Nutr Sci, 7, e7.
- HARA, T., KASHIHARA, D., ICHIMURA, A., KIMURA, I., TSUJIMOTO, G.
 & HIRASAWA, A. 2014. Role of free fatty acid receptors in the regulation of energy metabolism. *Biochimica et Biophysica Acta (BBA) Molecular and Cell Biology of Lipids*, 1841, 1292-1300.
- HASKELL, W. L., SPILLER, G. A., JENSEN, C. D., ELLIS, B. K. & GATES, J. E. 1992. Role of water-soluble dietary fiber in the management of elevated plasma cholesterol in healthy subjects. *The American journal of cardiology*, 69, 433-439.
- HÅVERSEN, L., DANIELSSON, K. N., FOGELSTRAND, L. & WIKLUND, O. 2009. Induction of proinflammatory cytokines by long-chain saturated fatty acids in human macrophages. *Atherosclerosis*, 202, 382-393.

- HAVRANKOVA, J., ROTH, J. & BROWNSTEIN, M. 1978. Insulin receptors are widely distributed in the central nervous system of the rat. *Nature*, 272, 827-829.
- HAYAKAWA, N., NAKAMOTO, Y., CHEN-YOSHIKAWA, T. F., KIDO, A., ISHIMORI, T., FUJIMOTO, K., YAMADA, T., SATO, M., AOYAMA, A., DATE, H. & TOGASHI, K. 2017. Gastric motility and emptying assessment by magnetic resonance imaging after lung transplantation: correlation with gastric emptying scintigraphy. *Abdom Radiol (NY)*, 42, 818-824.
- HEISLER, L. K. & LAM, D. D. 2017. An appetite for life: brain regulation of hunger and satiety. *Current opinion in pharmacology*, 37, 100-106.
- HENGIST, A., SCIARRILLO, C. M., GUO, J., WALTER, M. & HALL, K. D. 2023. Discordance between gut-derived appetite hormones and energy intake in humans. *medRxiv*.
- HENI, M., KULLMANN, S., GALLWITZ, B., HARING, H. U., PREISSL, H. & FRITSCHE, A. 2015. Dissociation of GLP-1 and insulin association with food processing in the brain: GLP-1 sensitivity despite insulin resistance in obese humans. *Mol Metab*, 4, 971-6.
- HERNÁNDEZ MORANTE, J. J., DÍAZ SOLER, I., MUÑOZ, J. S. G.,
 SÁNCHEZ, H. P., BARBERÁ ORTEGA, M. D. C., MARTÍNEZ, C. M.
 & MORILLAS RUIZ, J. M. 2020. Moderate Weight Loss Modifies Leptin and Ghrelin Synthesis Rhythms but Not the Subjective Sensations of Appetite in Obesity Patients. *Nutrients*, 12.
- HIGGINS, J. P., THOMAS, J., CHANDLER, J., CUMPSTON, M., LI, T., PAGE, M. J. & WELCH, V. A. 2019. Cochrane handbook for systematic reviews of interventions, John Wiley & Sons.
- HIGGINS, S. C., GUEORGUIEV, M. & KORBONITS, M. 2007. Ghrelin, the peripheral hunger hormone. *Ann Med*, 39, 116-36.
- HIGHAM, A., VAILLANT, C., YEGEN, B., THOMPSON, D. G. & DOCKRAY,G. J. 1997. Relation between cholecystokinin and antral innervation in the control of gastric emptying in the rat. *Gut*, 41, 24-32.
- HILL, A. J. & BLUNDELL, J. E. 1982. Nutrients and behaviour: research strategies for the investigation of taste characteristics, food preferences, hunger sensations and eating patterns in man. *J Psychiatr Res*, 17, 203-12.
- HILL, J. O., MELANSON, E. L. & WYATT, H. T. 2000. Dietary Fat Intake and Regulation of Energy Balance: Implications for Obesity12. *The Journal of Nutrition*, 130, 284S-288S.
- HINDER, R. A. & KELLY, K. A. 1977. Canine gastric emptying of solids and liquids. *American Journal of Physiology-Endocrinology and Metabolism*, 233, E335.
- HOAD, C., MARCIANI, L., FOLEY, S., TOTMAN, J., WRIGHT, J., BUSH, D., COX, E., CAMPBELL, E., SPILLER, R. & GOWLAND, P. 2007. Noninvasive quantification of small bowel water content by MRI: a validation study. *Physics in Medicine & Biology*, 52, 6909.
- HOLLOWOOD, T., BAYARRI, S., MARCIANI, L., BUSCH, J., FRANCIS, S., SPILLER, R., TAYLOR, A. & HORT, J. 2008. Modelling sweetness and

texture perception in model emulsion systems. *European Food Research and Technology*, 227, 537-545.

- HOLT, S. H., MILLER, J. C., PETOCZ, P. & FARMAKALIDIS, E. 1995. A satiety index of common foods. *Eur J Clin Nutr*, 49, 675-90.
- HONDA, K. L., LAMON-FAVA, S., MATTHAN, N. R., WU, D. & LICHTENSTEIN, A. H. 2015. EPA and DHA exposure alters the inflammatory response but not the surface expression of Toll-like receptor 4 in macrophages. *Lipids*, 50, 121-129.
- HORNER, K. M., BYRNE, N. M., CLEGHORN, G. J. & KING, N. A. 2014a. Reproducibility of gastric emptying in overweight and obese males. *Clin Nutr*, 33, 684-8.
- HORNER, K. M., BYRNE, N. M. & KING, N. A. 2014b. Reproducibility of subjective appetite ratings and ad libitum test meal energy intake in overweight and obese males. *Appetite*, 81, 116-22.
- HOUSE OF COMMONS LIBRARY. 2023. *Obesity statistics* [Online]. UK Parliament Available: <u>https://commonslibrary.parliament.uk/research-briefings/sn03336/</u> [Accessed 12 July 2023].
- HOUTEN, M. V., POSNER, B. I., KOPRIWA, B. M. & BRAWER, J. R. 1979. Insulin-binding sites in the rat brain: in vivo localization to the circumventricular organs by quantitative radioautography. *Endocrinology*, 105, 666-673.
- HUNG, A. M., BOOKER, C., ELLIS, C. D., SIEW, E. D., GRAVES, A. J., SHINTANI, A., ABUMRAD, N. N., HIMMELFARB, J. & IKIZLER, T. A. 2015. Omega-3 fatty acids inhibit the up-regulation of endothelial chemokines in maintenance hemodialysis patients. *Nephrology Dialysis Transplantation*, 30, 266-274.
- HUSSEIN, M. O., HOAD, C. L., WRIGHT, J., SINGH, G., STEPHENSON, M. C., COX, E. F., PLACIDI, E., PRITCHARD, S. E., COSTIGAN, C. & RIBEIRO, H. 2015. Fat emulsion intragastric stability and droplet size modulate gastrointestinal responses and subsequent food intake in young adults. *The Journal of nutrition*, 145, 1170-1177.
- HYKIN, J., FREEMAN, A., GOWLAND, P., BOWTELL, R., WORTHINGTON, B., SPILLER, R. & MANSFIELD, P. 1994. Measurement of GI water content using EPI at 0.5 tesla. *Magnetic Resonance Materials in Physics*, *Biology and Medicine*, 2, 471-473.
- HYLAND, M. E., IRVINE, S. H., THACKER, C., DANN, P. L. & DENNIS, I.
 1989. Psychometric analysis of the Stunkard-Messick Eating
 Questionnaire (SMEQ) and comparison with the Dutch Eating Behavior
 Questionnaire (DEBQ). *Current Psychology*, 8, 228-233.
- IMAIOS. 2023. *Magnetic field gradients* [Online]. Available: <u>https://www.imaios.com/en/e-mri/spatial-encoding-in-mri/magnetic-field-gradients</u> [Accessed 28 October 2023].
- INTAKE24. 2023. *Intake24* [Online]. Available: <u>https://intake24.co.uk/#content</u> [Accessed 12 May 2023].
- JAKOBSDOTTIR S, D. R. M., DEIJEN JB, VELTMAN DJ, DRENT ML 2012. Brain Activation by Visual Food-Related Stimuli and Correlations with

Metabolic and Hormonal Parameters: A fMRI Study. *The Open Neuroendocrinology Journal*, 2012, 5-12.

- JAKOBSDOTTIR, S., RUITER, M. D., DEIJEN, J. B., VELTMAN, D. J. & DRENT, M. L. 2012. Brain Activation by Visual Food-Related Stimuli and Correlations with Metabolic and Hormonal Parameters: A fMRI study. *Open Neuroendocrinol J*, 2012, 5-12.
- JÁUREGUI-LOBERA, I., GARCÍA-CRUZ, P., CARBONERO-CARREÑO, R., MAGALLARES, A. & RUIZ-PRIETO, I. 2014. Psychometric properties of Spanish version of the Three-Factor Eating Questionnaire-R18 (Tfeq-Sp) and its relationship with some eating- and body image-related variables. *Nutrients*, 6, 5619-35.
- JAY, A. G., JOHN, O. D. & RAYMOND, J. D. 2003. Encoding Predictive Reward Value in Human Amygdala and Orbitofrontal Cortex. *Science* 301, 1104-1107.
- JEAYS, A. D., LAWFORD, P. V., GILLOTT, R., SPENCER, P. A., BARDHAN, K. D. & HOSE, D. R. 2007. A framework for the modeling of gut blood flow regulation and postprandial hyperaemia. *World journal of gastroenterology*, 13, 1393-1398.
- JI, Z., HADAC, E. M., HENNE, R. M., PATEL, S. A., LYBRAND, T. P. & MILLER, L. J. 1997. Direct identification of a distinct site of interaction between the carboxyl-terminal residue of cholecystokinin and the type A cholecystokinin receptor using photoaffinity labeling. *Journal of Biological Chemistry*, 272, 24393-24401.
- JOBST, E. E., ENRIORI, P. J. & COWLEY, M. A. 2004. The electrophysiology of feeding circuits. *Trends Endocrinol Metab*, 15, 488-99.
- JOHNSON, J. & VICKERS, Z. 1993. Effects of flavor and macronutrient composition of food servings on liking, hunger and subsequent intake. *Appetite*, 21, 25-39.
- JOHNSTONE, A. 2013. 7 Protein and satiety. *In:* BLUNDELL, J. E. & BELLISLE, F. (eds.) *Satiation, Satiety and the Control of Food Intake*. Woodhead Publishing.
- JONES, B. J., TAN, T. & BLOOM, S. R. 2012a. Minireview: Glucagon in stress and energy homeostasis. *Endocrinology*, 153, 1049-54.
- JONES, R. B., MCKIE, S., ASTBURY, N., LITTLE, T. J., TIVEY, S., LASSMAN, D. J., MCLAUGHLIN, J., LUCKMAN, S., WILLIAMS, S. R., DOCKRAY, G. J. & THOMPSON, D. G. 2012b. Functional neuroimaging demonstrates that ghrelin inhibits the central nervous system response to ingested lipid. *Gut*, 61, 1543-51.
- KELLEY, A. E., BALDO, B. A., PRATT, W. E. & WILL, M. J. 2005. Corticostriatal-hypothalamic circuitry and food motivation: integration of energy, action and reward. *Physiol Behav*, 86, 773-95.
- KHALAF, A., HOAD, C. L., BLACKSHAW, E., ALYAMI, J., SPILLER, R. C., GOWLAND, P. A., VINAYAKA-MOORTHY, V., PERKINS, A. C., MORAN, G. W. & MARCIANI, L. 2020a. Simultaneous measurement of gastric emptying of a soup test meal using MRI and gamma scintigraphy. *Diagnostics*, 10, 170.

- KHALAF, A., HOAD, C. L., MENYS, A., NOWAK, A., RADFORD, S., TAYLOR, S. A., LATIEF, K., LINGAYA, M., FALCONE, Y., SINGH, G., SPILLER, R. C., GOWLAND, P. A., MARCIANI, L. & MORAN, G.
 W. 2020b. Gastrointestinal peptides and small-bowel hypomotility are possible causes for fasting and postprandial symptoms in active Crohn's disease. *The American Journal of Clinical Nutrition*, 111, 131-140.
- KIM, G., LIN, J., BLOMAIN, E. & WALDMAN, S. 2013. Anti-Obesity Pharmacotherapy: New Drugs and Emerging Targets. *Clinical pharmacology and therapeutics*, 95.
- KOUROUNIOTIS, S., KEAST, R., RIDDELL, L., LACY, K., THORPE, M. & CICERALE, S. 2016. The importance of taste on dietary choice, behaviour and intake in a group of young adults. *Appetite*, 103, 1-7.
- KRAL, T. V., ROE, L. S. & ROLLS, B. J. 2004. Combined effects of energy density and portion size on energy intake in women. *The American journal* of clinical nutrition, 79, 962-968.
- KROEMER, N. B., KREBS, L., KOBIELLA, A., GRIMM, O., PILHATSCH, M., BIDLINGMAIER, M., ZIMMERMANN, U. S. & SMOLKA, M. N. 2013. Fasting levels of ghrelin covary with the brain response to food pictures. *Addict Biol*, 18, 855-62.
- KRUDE, H. & GRÜTERS, A. 2000. Implications of proopiomelanocortin (POMC) mutations in humans: the POMC deficiency syndrome. *Trends in Endocrinology & Metabolism*, 11, 15-22.
- KULKARNI, R. N., WANG, Z.-L., WANG, R.-M., HURLEY, J. D., SMITH, D. M., GHATEI, M. A., WITHERS, D. J., GARDINER, J. V., BAILEY, C. J. & BLOOM, S. R. 1997. Leptin rapidly suppresses insulin release from insulinoma cells, rat and human islets and, in vivo, in mice. *The Journal of clinical investigation*, 100, 2729-2736.
- KULLMANN, S., FRANK, S., HENI, M., KETTERER, C., VEIT, R., HÄRING, H.-U., FRITSCHE, A. & PREISSL, H. 2013. Intranasal insulin modulates intrinsic reward and prefrontal circuitry of the human brain in lean women. *Neuroendocrinology*, 97, 176-182.
- KUMAR, V. M. 1999. Neural regulation of glucose homeostasis. *Indian J Physiol Pharmacol*, 43, 415-24.
- KUNZ, P., FEINLE, C., SCHWIZER, W., FRIED, M. & BOESIGER, P. 1999. Assessment of gastric motor function during the emptying of solid and liquid meals in humans by MRI. *J Magn Reson Imaging*, 9, 75-80.
- LAESSLE, R. G., TUSCHL, R. J., KOTTHAUS, B. C. & PIRKE, K. M. 1989. A comparison of the validity of three scales for the assessment of dietary restraint. *J Abnorm Psychol*, 98, 504-7.
- LAIRD, A. R., ROBINSON, J. L., MCMILLAN, K. M., TORDESILLAS-GUTIERREZ, D., MORAN, S. T., GONZALES, S. M., RAY, K. L., FRANKLIN, C., GLAHN, D. C., FOX, P. T. & LANCASTER, J. L. 2010. Comparison of the disparity between Talairach and MNI coordinates in functional neuroimaging data: validation of the Lancaster transform. *Neuroimage*, 51, 677-83.

- LARA, J., TAYLOR, M. A. & MACDONALD, I. A. 2010. Is ad libitum energy intake in overweight subjects reproducible in laboratory studies using the preload paradigm? *European Journal of Clinical Nutrition*, 64, 1028-1031.
- LARSSON, L.-I., SUNDLER, F. & HÅKANSON, R. 1975. Immunohistochemical localization of human pancreatic polypeptide (HPP) to a population of islet cells. *Cell and tissue research*, 156, 167-171.
- LASSMAN, D. J., MCKIE, S., GREGORY, L. J., LAL, S., D'AMATO, M., STEELE, I., VARRO, A., DOCKRAY, G. J., WILLIAMS, S. C. & THOMPSON, D. G. 2010. Defining the role of cholecystokinin in the lipid-induced human brain activation matrix. *Gastroenterology*, 138, 1514-24.
- LAU, D. C., DOUKETIS, J. D., MORRISON, K. M., HRAMIAK, I. M., SHARMA, A. M. & UR, E. 2007. 2006 Canadian clinical practice guidelines on the management and prevention of obesity in adults and children [summary]. *Cmaj*, 176, S1-13.
- LAYER, P., JUUL HOLST, J., GRANDT, D. & GOEBELL, H. 1995. Ileal release of glucagon-like peptide-1 (GLP-1). *Digestive Diseases and Sciences*, 40, 1074-1082.
- LE QUELLEC, A., KERVRAN, A., BLACHE, P., CIURANA, A. J. & BATAILLE, D. 1992. Oxyntomodulin-like immunoreactivity: diurnal profile of a new potential enterogastrone. *J Clin Endocrinol Metab*, 74, 1405-9.
- LE ROUX, C. W., BATTERHAM, R. L., AYLWIN, S. J., PATTERSON, M., BORG, C. M., WYNNE, K. J., KENT, A., VINCENT, R. P., GARDINER, J., GHATEI, M. A. & BLOOM, S. R. 2006. Attenuated peptide YY release in obese subjects is associated with reduced satiety. *Endocrinology*, 147, 3-8.
- LEBLANC, J. 2000. Nutritional implications of cephalic phase thermogenic responses. *Appetite*, 34, 214-216.
- LEBLANC, J. & CABANAC, M. 1989. Cephalic postprandial thermogenesis in human subjects. *Physiology & Behavior*, 46, 479-482.
- LEBRUN, B., BARIOHAY, B., MOYSE, E. & JEAN, A. 2006. Brain-derived neurotrophic factor (BDNF) and food intake regulation: a minireview. *Auton Neurosci*, 126-127, 30-8.
- LEE, S., KWOK, K., LIAU, C. & LEUNG, T. 2002. Screening Chinese patients with eating disorders using the Eating Attitudes Test in Hong Kong. *Int J Eat Disord*, 32, 91-7.
- LEIDY, H. J., CARNELL, N. S., MATTES, R. D. & CAMPBELL, W. W. 2007. Higher Protein Intake Preserves Lean Mass and Satiety with Weight Loss in Pre-obese and Obese Women. *Obesity*, 15, 421-429.
- LENNERNÄS, M., FJELLSTRÖM, C., BECKER, W., GIACHETTI, I., SCHMITT, A., DE WINTER, A. & KEARNEY, M. 1997. Influences on food choice perceived to be important by nationally-representative samples of adults in the European Union. *European Journal of Clinical Nutrition*, 51.

- LENNERZ, B. S., ALSOP, D. C., HOLSEN, L. M., STERN, E., ROJAS, R., EBBELING, C. B., GOLDSTEIN, J. M. & LUDWIG, D. S. 2013. Effects of dietary glycemic index on brain regions related to reward and craving in men. *The American journal of clinical nutrition*, 98, 641-647.
- LEVIN, F., EDHOLM, T., SCHMIDT, P. T., GRYBÄCK, P., JACOBSSON, H., DEGERBLAD, M., HÖYBYE, C., HOLST, J. J., REHFELD, J. F., HELLSTRÖM, P. M. & NÄSLUND, E. 2006. Ghrelin Stimulates Gastric Emptying and Hunger in Normal-Weight Humans. *The Journal of Clinical Endocrinology & Metabolism*, 91, 3296-3302.
- LI, J., AN, R., ZHANG, Y., LI, X. & WANG, S. 2012. Correlations of macronutrient-induced functional magnetic resonance imaging signal changes in human brain and gut hormone responses. *Am J Clin Nutr*, 96, 275-82.
- LI, Y., HAO, Y. & OWYANG, C. 1997. High-affinity CCK-A receptors on the vagus nerve mediate CCK-stimulated pancreatic secretion in rats. *American Journal of Physiology-Gastrointestinal and Liver Physiology*, 273, G679-G685.
- LIANG, C.-P. & TALL, A. R. 2001. Transcriptional profiling reveals global defects in energy metabolism, lipoprotein, and bile acid synthesis and transport with reversal by leptin treatment in ob/ob mouse liver. *Journal of Biological Chemistry*, 276, 49066-49076.
- LIBERATI, A., ALTMAN, D. G., TETZLAFF, J., MULROW, C., GØTZSCHE, P. C., IOANNIDIS, J. P. A., CLARKE, M., DEVEREAUX, P. J., KLEIJNEN, J. & MOHER, D. 2009. The PRISMA statement for reporting systematic reviews and meta-analyses of studies that evaluate health care interventions: explanation and elaboration. *PLoS Med*, 6, e1000100e1000100.
- LICHTENSTEIN, A. H. 2013. Fats and Oils. *In:* CABALLERO, B. (ed.) *Encyclopedia of Human Nutrition (Third Edition).* Waltham: Academic Press.
- LIDDLE, R. A., GOLDFINE, I. D., ROSEN, M. S., TAPLITZ, R. & WILLIAMS, J. 1985. Cholecystokinin bioactivity in human plasma. Molecular forms, responses to feeding, and relationship to gallbladder contraction. *The Journal of clinical investigation*, 75, 1144-1152.
- LIDDLE, R. A., MORITA, E. T., CONRAD, C. K. & WILLIAMS, J. A. 1986. Regulation of gastric emptying in humans by cholecystokinin. *J Clin Invest*, 77, 992-6.
- LITTLE, T. J., MCKIE, S., JONES, R. B., D'AMATO, M., SMITH, C., KISS, O., THOMPSON, D. G. & MCLAUGHLIN, J. T. 2014. Mapping glucosemediated gut-to-brain signalling pathways in humans. *Neuroimage*, 96, 1-11.
- LIU, W., JIN, Y., WILDE, P. J., HOU, Y., WANG, Y. & HAN, J. 2021. Mechanisms, physiology, and recent research progress of gastric emptying. *Crit Rev Food Sci Nutr*, 61, 2742-2755.

- LIU, Y., FOX, P. T., LIU, H.-L. & GAO, J.-H. 2000. The temporal response of the brain after eating revealed by functional MRI. *Nature (London)*, 405, 1058-1062.
- LIVINGSTONE, K., GIVENS, D., JACKSON, K. & LOVEGROVE, J. 2014. Comparative effect of dairy fatty acids on cell adhesion molecules, nitric oxide and relative gene expression in healthy and diabetic human aortic endothelial cells. *Atherosclerosis*, 234, 65-72.
- LOCKE, A. E., KAHALI, B., BERNDT, S. I., JUSTICE, A. E., PERS, T. H., DAY, F. R., POWELL, C., VEDANTAM, S., BUCHKOVICH, M. L. & YANG, J. 2015. Genetic studies of body mass index yield new insights for obesity biology. *Nature*, 518, 197-206.
- LÖFFLER, A., LUCK, T., THEN, F. S., SIKORSKI, C., KOVACS, P.,
 BÖTTCHER, Y., BREITFELD, J., TÖNJES, A., HORSTMANN, A.,
 LÖFFLER, M., ENGEL, C., THIERY, J., VILLRINGER, A.,
 STUMVOLL, M. & RIEDEL-HELLER, S. G. 2015. Eating Behaviour in
 the General Population: An Analysis of the Factor Structure of the German
 Version of the Three-Factor-Eating-Questionnaire (TFEQ) and Its
 Association with the Body Mass Index. *PLoS One*, 10, e0133977.
- LOWE, M., BUTRYN, M., DIDIE, E., ANNUNZIATO, R., THOMAS, J., CRERAND, C., OCHNER, C., COLETTA, M., BELLACE, D., WALLAERT, M. & HALFORD, J. 2009. The Power of Food Scale. A new measure of the psychological influence of the food environment. *Appetite*, 53, 114-118.
- LUO, J., HU, Y., KONG, W. & YANG, M. 2014. Evaluation and structureactivity relationship analysis of a new series of arylnaphthalene lignans as potential anti-tumor agents. *Plos one*, 9, e93516.
- LUTTER, M. & NESTLER, E. J. 2009. Homeostatic and hedonic signals interact in the regulation of food intake. *J Nutr*, 139, 629-32.
- LUTZ, T. A. 2006. Amylinergic control of food intake. *Physiol Behav*, 89, 465-71.
- LUTZ, T. A., GEARY, N., SZABADY, M. M., DEL PRETE, E. & SCHARRER, E. 1995. Amylin decreases meal size in rats. *Physiol Behav*, 58, 1197-202.
- MA, M., LIU, H., YU, J., HE, S., LI, P., MA, C., ZHANG, H., XU, L., PING, F., LI, W., SUN, Q. & LI, Y. 2020. Triglyceride is independently correlated with insulin resistance and islet beta cell function: a study in population with different glucose and lipid metabolism states. *Lipids in Health and Disease*, 19, 121.
- MACFARLANE, N. G. 2018. Digestion and absorption. *Anaesthesia & Intensive Care Medicine*, 19, 125-127.
- MAKARONIDIS, J. M. & BATTERHAM, R. L. 2018. Obesity, body weight regulation and the brain: insights from fMRI. *The British Journal of Radiology*, 91, 20170910.
- MAKI, K. C. & PHILLIPS, A. K. 2015. Dietary substitutions for refined carbohydrate that show promise for reducing risk of type 2 diabetes in men and women. *The Journal of nutrition*, 145, 159S-163S.

- MALAFARINA, V., URIZ-OTANO, F., GIL-GUERRERO, L. & INIESTA, R. 2013. The anorexia of ageing: physiopathology, prevalence, associated comorbidity and mortality. A systematic review. *Maturitas*, 74, 293-302.
- MALANDRINO, N. & SMITH, R. J. 2018. Synthesis, secretion, and transport of peptide hormones. *Principles of Endocrinology and Hormone Action*, 1, 29-42.
- MALDJIAN, J. A., LAURIENTI, P. J., KRAFT, R. A. & BURDETTE, J. H. 2003. An automated method for neuroanatomic and cytoarchitectonic atlas-based interrogation of fMRI data sets. *NeuroImage (Orlando, Fla.)*, 19, 1233-1239.
- MALIK, S., MCGLONE, F., BEDROSSIAN, D. & DAGHER, A. 2008. Ghrelin modulates brain activity in areas that control appetitive behavior. *Cell Metab*, 7, 400-9.
- MARCIANI, L., COX, E. F., HOAD, C. L., PRITCHARD, S., TOTMAN, J. J., FOLEY, S., MISTRY, A., EVANS, S., GOWLAND, P. A. & SPILLER, R. C. 2010. Postprandial changes in small bowel water content in healthy subjects and patients with irritable bowel syndrome. *Gastroenterology*, 138, 469-77, 477.e1.
- MARCIANI, L., COX, E. F., HOAD, C. L., TOTMAN, J. J., COSTIGAN, C., SINGH, G., SHEPHERD, V., CHALKLEY, L., ROBINSON, M., ISON, R., GOWLAND, P. A. & SPILLER, R. C. 2013a. Effects of various food ingredients on gall bladder emptying. *Eur J Clin Nutr*, 67, 1182-7.
- MARCIANI, L., COX, E. F., PRITCHARD, S. E., MAJOR, G., HOAD, C. L., MELLOWS, M., HUSSEIN, M. O., COSTIGAN, C., FOX, M., GOWLAND, P. A. & SPILLER, R. C. 2015. Additive effects of gastric volumes and macronutrient composition on the sensation of postprandial fullness in humans. *European Journal of Clinical Nutrition*, 69, 380-384.
- MARCIANI, L., FAULKS, R., WICKHAM, M. S., BUSH, D., PICK, B., WRIGHT, J., COX, E. F., FILLERY-TRAVIS, A., GOWLAND, P. A. & SPILLER, R. C. 2008. Effect of intragastric acid stability of fat emulsions on gastric emptying, plasma lipid profile and postprandial satiety. *British Journal of Nutrition*, 101, 919-928.
- MARCIANI, L., GOWLAND, P. A., SPILLER, R. C., MANOJ, P., MOORE, R. J., YOUNG, P. & FILLERY-TRAVIS, A. J. 2001. Effect of meal viscosity and nutrients on satiety, intragastric dilution, and emptying assessed by MRI. American Journal of Physiology-Gastrointestinal and Liver Physiology, 280, G1227-G1233.
- MARCIANI, L., HALL, N., PRITCHARD, S. E., COX, E. F., TOTMAN, J. J., LAD, M., HOAD, C. L., FOSTER, T. J., GOWLAND, P. A. & SPILLER, R. C. 2012. Preventing Gastric Sieving by Blending a Solid/Water Meal Enhances Satiation in Healthy Humans. *The Journal of Nutrition*, 142, 1253-1258.
- MARCIANI, L., PRITCHARD, S. E., HELLIER-WOODS, C., COSTIGAN, C., HOAD, C. L., GOWLAND, P. A. & SPILLER, R. C. 2013b. Delayed gastric emptying and reduced postprandial small bowel water content of

equicaloric whole meal bread versus rice meals in healthy subjects: novel MRI insights. *Eur J Clin Nutr*, 67, 754-8.

- MARCIANI, L., WICKHAM, M., SINGH, G., BUSH, D., PICK, B., COX, E., FILLERY-TRAVIS, A., FAULKS, R., MARSDEN, C. & GOWLAND, P. A. 2007. Enhancement of intragastric acid stability of a fat emulsion meal delays gastric emptying and increases cholecystokinin release and gallbladder contraction. *American Journal of Physiology-Gastrointestinal and Liver Physiology*.
- MARCUS, M. D., WING, R. R. & HOPKINS, J. 1988. Obese binge eaters: Affect, cognitions, and response to behavioral weight control. *Journal of consulting and clinical psychology*, 56, 433.
- MARSTON, N. A., GIUGLIANO, R. P., IM, K., SILVERMAN, M. G.,
 O'DONOGHUE, M. L., WIVIOTT, S. D., FERENCE, B. A. &
 SABATINE, M. S. 2019. Association Between Triglyceride Lowering and
 Reduction of Cardiovascular Risk Across Multiple Lipid-Lowering
 Therapeutic Classes. *Circulation*, 140, 1308-1317.
- MARZULLO, P., VERTI, B., SAVIA, G., WALKER, G. E., GUZZALONI, G., TAGLIAFERRI, M., DI BLASIO, A. & LIUZZI, A. 2004. The relationship between active ghrelin levels and human obesity involves alterations in resting energy expenditure. *J Clin Endocrinol Metab*, 89, 936-9.
- MATSUDA, M., LIU, Y., MAHANKALI, S., PU, Y., MAHANKALI, A., WANG, J., DEFRONZO, R. A., FOX, P. T. & GAO, J. H. 1999. Altered hypothalamic function in response to glucose ingestion in obese humans. *Diabetes*, 48, 1801-6.
- MATTES, R. D. 1997. Physiologic responses to sensory stimulation by food: nutritional implications. *Journal of the American Dietetic Association*, 97, 406-413.
- MATTES, R. D. 2005. Fat taste and lipid metabolism in humans. *Physiology & behavior*, 86, 691-697.

MATTES, R. D. & ROTHACKER, D. 2001. Beverage viscosity is inversely related to postprandial hunger in humans. *Physiol Behav*, 74, 551-7.

- MAYER, J. 1955. Regulation of energy intake and the body weight: the glucostatic theory and the lipostatic hypothesis. *Ann N Y Acad Sci*, 63, 15-43.
- MAZZAWI, T., BARTSCH, E., BENAMMI, S., FERRO, R. M. C., NIKITINA, E., NIMER, N., ORTEGA, L. J., PERROTTE, C., PITHON, J. V., ROSALINA, S., SHARP, A., STEVANO, R., HATLEBAKK, J. G. & HAUSKEN, T. 2019. Gastric Emptying of Low- and High-Caloric Liquid Meals Measured Using Ultrasonography in Healthy Volunteers. *Ultrasound Int Open*, 5, E27-e33.
- MCGREGOR, I. S. & LEE, A. M. 1998. Changes in respiratory quotient elicited in rats by a conditioned stimulus predicting food. *Physiology & behavior*, 63, 227-232.
- MCROBBIE, D. W. 2003. *MRI from Picture to Proton*, Cambridge University Press.

- MEHTA, S., MELHORN, S. J., SMERAGLIO, A., TYAGI, V., GRABOWSKI, T., SCHWARTZ, M. W. & SCHUR, E. A. 2012. Regional brain response to visual food cues is a marker of satiety that predicts food choice. *The American journal of clinical nutrition*, 96, 989-999.
- MEIER, J. J., NAUCK, M. A., POTT, A., HEINZE, K., GOETZE, O., BULUT, K., SCHMIDT, W. E., GALLWITZ, B. & HOLST, J. J. 2006. Glucagonlike peptide 2 stimulates glucagon secretion, enhances lipid absorption, and inhibits gastric acid secretion in humans. *Gastroenterology*, 130, 44-54.
- MELHORN, S. J., TYAGI, V., SMERAGLIO, A., ROTH, C. L. & SCHUR, E. A. 2014. Initial evidence that GLP-1 receptor blockade fails to suppress postprandial satiety or promote food intake in humans. *Appetite*, 82, 85-90.
- MENDOZA, J. A., DREWNOWSKI, A. & CHRISTAKIS, D. A. 2007. Dietary energy density is associated with obesity and the metabolic syndrome in U.S. adults. *Diabetes Care*, 30, 974-9.
- MENYS, A., KESZTHELYI, D., FITZKE, H., FIKREE, A., ATKINSON, D., AZIZ, Q. & TAYLOR, S. A. 2017. A magnetic resonance imaging study of gastric motor function in patients with dyspepsia associated with Ehlers-Danlos Syndrome-Hypermobility Type: A feasibility study. *Neurogastroenterol Motil*, 29.
- MEYER, J. H., THOMSON, J. B., COHEN, M. B., SHADCHEHR, A. & MANDIOLA, S. A. 1979. Sieving of Solid Food by the Canine Stomach and Sieving After Gastric Surgery. *Gastroenterology*, 76, 804-813.
- MEYER-GERSPACH, A. C., WÖLNERHANSSEN, B., BEGLINGER, B., NESSENIUS, F., NAPITUPULU, M., SCHULTE, F. H., STEINERT, R.
 E. & BEGLINGER, C. 2014. Gastric and intestinal satiation in obese and normal weight healthy people. *Physiol Behav*, 129, 265-71.
- MIETTINEN, S.-M., TUORILA, H., PIIRONEN, V., VEHKALAHTI, K. & HYVÖNEN, L. 2002. Effect of emulsion characteristics on the release of aroma as detected by sensory evaluation, static headspace gas chromatography, and electronic nose. *Journal of agricultural and food chemistry*, 50, 4232-4239.
- MILES, E. A., THIES, F., WALLACE, F. A., POWELL, J. R., HURST, T. L., NEWSHOLME, E. A. & CALDER, P. C. 2001. Influence of age and dietary fish oil on plasma soluble adhesion molecule concentrations. *Clinical Science*, 100, 91-100.
- MINOKOSHI, Y., KIM, Y.-B., PERONI, O. D., FRYER, L. G., MÜLLER, C., CARLING, D. & KAHN, B. B. 2002. Leptin stimulates fatty-acid oxidation by activating AMP-activated protein kinase. *Nature*, 415, 339-343.
- MINTZ, L. B. & O'HALLORAN, M. S. 2000. The Eating Attitudes Test: validation with DSM-IV eating disorder criteria. *J Pers Assess*, 74, 489-503.
- MOLLET, A., LUTZ, T. A., MEIER, S., RIEDIGER, T., RUSHING, P. A. & SCHARRER, E. 2001. Histamine H1 receptors mediate the anorectic action of the pancreatic hormone amylin. *Am J Physiol Regul Integr Comp Physiol*, 281, R1442-8.

- MONETA, G. L., TAYLOR, D. C., HELTON, W. S., MULHOLLAND, M. W. & STRANDNESS, D. E., JR. 1988. Duplex ultrasound measurement of postprandial intestinal blood flow: effect of meal composition. *Gastroenterology*, 95, 1294-301.
- MORA, M. E. V., SCARFONE, A., VALENZA, V., CALVANI, M., GRECO, A. V., GASBARRINI, G. & MINGRONE, G. 2005. Ghrelin Does Not Influence Gastric Emptying in Obese Subjects. *Obesity Research*, 13, 739-744.
- MORAN, L. J., LUSCOMBE-MARSH, N. D., NOAKES, M., WITTERT, G. A., KEOGH, J. B. & CLIFTON, P. M. 2005. The satiating effect of dietary protein is unrelated to postprandial ghrelin secretion. *The Journal of Clinical Endocrinology & Metabolism*, 90, 5205-5211.
- MORAN, T. H. & KINZIG, K. P. 2004. Gastrointestinal satiety signals II. Cholecystokinin. *American Journal of Physiology-Gastrointestinal and Liver Physiology*, 286, G183-G188.
- MORAN, T. H., KORNBLUH, R., MOORE, K. & SCHWARTZ, G. J. 1994. Cholecystokinin inhibits gastric emptying and contracts the pyloric sphincter in rats by interacting with low affinity CCK receptor sites. *Regulatory Peptides*, 52, 165-172.
- MORAN, T. H., ROBINSON, P. H., GOLDRICH, M. S. & MCHUGH, P. R. 1986. Two brain cholecystokinin receptors: implications for behavioral actions. *Brain research*, 362, 175-179.
- MORLEY, J. E. & SILVER, A. J. 1988. Anorexia in the elderly. *Neurobiol Aging*, 9, 9-16.
- MORYS, F., GARCÍA-GARCÍA, I. & DAGHER, A. 2020. Is obesity related to enhanced neural reactivity to visual food cues? A review and metaanalysis. *Social Cognitive and Affective Neuroscience*, 18.
- MU, H. & HØY, C.-E. 2004. The digestion of dietary triacylglycerols. *Progress in lipid research*, 43, 105-133.
- MUHAMAD, I. I., JUSOH, Y. M. M., NAWI, N. M., AZIZ, A. A., PADZIL, A.
 M. & LIAN, H. L. 2018. Chapter 15 Advanced Natural Food Colorant Encapsulation Methods: Anthocyanin Plant Pigment. *In:* GRUMEZESCU, A. M. & HOLBAN, A. M. (eds.) *Natural and Artificial Flavoring Agents and Food Dyes.* Academic Press.
- MULLER, V. I., CIESLIK, E. C., LAIRD, A. R., FOX, P. T., RADUA, J., MATAIX-COLS, D., TENCH, C. R., YARKONI, T., NICHOLS, T. E., TURKELTAUB, P. E., WAGER, T. D. & EICKHOFF, S. B. 2018. Ten simple rules for neuroimaging meta-analysis. *Neurosci Biobehav Rev*, 84, 151-161.
- MURPHY, K. G. & BLOOM, S. R. 2006. Gut hormones and the regulation of energy homeostasis. *Nature*, 444, 854-859.

MURRAY, C. D., MARTIN, N. M., PATTERSON, M., TAYLOR, S. A., GHATEI, M. A., KAMM, M. A., JOHNSTON, C., BLOOM, S. R. & EMMANUEL, A. V. 2005. Ghrelin enhances gastric emptying in diabetic gastroparesis: a double blind, placebo controlled, crossover study. *Gut*, 54, 1693-8.

- MURRAY, K., WILKINSON-SMITH, V., HOAD, C., COSTIGAN, C., COX, E., LAM, C., MARCIANI, L., GOWLAND, P. & SPILLER, R. C. 2014. Differential effects of FODMAPs (fermentable oligo-, di-, monosaccharides and polyols) on small and large intestinal contents in healthy subjects shown by MRI. *Am J Gastroenterol*, 109, 110-9.
- MUTT, V. & JORPES, J. 1968. Structure of Porcine Cholecystokinin-Pancreozymin: 1. Cleavage with Thrombin and with Trypsin. *European journal of biochemistry*, 6, 156-162.
- MYERS, M. G., MÜNZBERG, H., LEINNINGER, G. M. & LESHAN, R. L. 2009. The geometry of leptin action in the brain: more complicated than a simple ARC. *Cell metabolism*, 9, 117-123.
- NAGELL, C., WETTERGREN, A., PEDERSEN, J., MORTENSEN, D. & HOLST, J. 2004. Glucagon-like peptide-2 inhibits antral emptying in man, but is not as potent as glucagon-like peptide-1. *Scandinavian journal of gastroenterology*, 39, 353-358.
- NAIR, N. S., BRENNAN, I. M., LITTLE, T. J., GENTILCORE, D., HAUSKEN, T., JONES, K. L., WISHART, J. M., HOROWITZ, M. & FEINLE-BISSET, C. 2008. Reproducibility of energy intake, gastric emptying, blood glucose, plasma insulin and cholecystokinin responses in healthy young males. *British Journal of Nutrition*, 101, 1094-1102.
- NAKAZATO, M., MURAKAMI, N., DATE, Y., KOJIMA, M., MATSUO, H., KANGAWA, K. & MATSUKURA, S. 2001. A role for ghrelin in the central regulation of feeding. *Nature*, 409, 194-8.
- NAKRANI, M. N., WINELAND, R. H. & ANJUM, F. 2023. Physiology, Glucose Metabolism. *StatPearls*. Treasure Island (FL) ineligible companies. Disclosure: Robert Wineland declares no relevant financial relationships with ineligible companies. Disclosure: Fatima Anjum declares no relevant financial relationships with ineligible companies.: StatPearls Publishing

Copyright © 2023, StatPearls Publishing LLC.

- NÄSLUND, E., KING, N., MANSTEN, S., ADNER, N., HOLST, J. J., GUTNIAK, M. & HELLSTRÖM, P. M. 2004. Prandial subcutaneous injections of glucagon-like peptide-1 cause weight loss in obese human subjects. *Br J Nutr*, 91, 439-46.
- NÄSLUND, E., SKOGAR, S., EFENDIC, S. & HELLSTRÖM, P. M. 2002. Glucagon-like peptide-1 analogue LY315902: effect on intestinal motility and release of insulin and somatostatin. *Regulatory peptides*, 106, 89-95.
- NAUCK, M. A., KLEINE, N., ORSKOV, C., HOLST, J. J., WILLMS, B. & CREUTZFELDT, W. 1993. Normalization of fasting hyperglycaemia by exogenous glucagon-like peptide 1 (7-36 amide) in type 2 (non-insulin-dependent) diabetic patients. *Diabetologia*, 36, 741-4.
- NEDERKOORN, C., SMULDERS, F. & JANSEN, A. 2000. Cephalic phase responses, craving and food intake in normal subjects. *Appetite*, 35, 45-55.
- NISWENDER, K. D., BASKIN, D. G. & SCHWARTZ, M. W. 2004. Insulin and its evolving partnership with leptin in the hypothalamic control of energy homeostasis. *Trends in Endocrinology & Metabolism*, 15, 362-369.

- NOGUEIRAS, R., WIEDMER, P., PEREZ-TILVE, D., VEYRAT-DUREBEX, C., KEOGH, J. M., SUTTON, G. M., PFLUGER, P. T., CASTANEDA, T. R., NESCHEN, S. & HOFMANN, S. M. 2007. The central melanocortin system directly controls peripheral lipid metabolism. *The Journal of clinical investigation*, 117, 3475-3488.
- NYMO, S., COUTINHO, S. R., EKNES, P. H., VESTBOSTAD, I., REHFELD, J. F., TRUBY, H., KULSENG, B. & MARTINS, C. 2018. Investigation of the long-term sustainability of changes in appetite after weight loss. *International Journal of Obesity*, 42, 1489-1499.
- OLSEN, N. V., MENICHELLI, E., SØRHEIM, O. & NÆS, T. 2012. Likelihood of buying healthy convenience food: An at-home testing procedure for ready-to-heat meals. *Food Quality and Preference*, 24, 171-178.
- OLSON, B. R., HOFFMAN, G. E., SVED, A. F., STRICKER, E. M. & VERBALIS, J. G. 1992. Cholecystokinin induces c-fos expression in hypothalamic oxytocinergic neurons projecting to the dorsal vagal complex. *Brain Res*, 569, 238-48.
- OPSTAL, A. M. V., AKINTOLA, A. A., ELST, M. V. D., WESTENDORP, R. G., PIJL, H., HEEMST, D. V. & GROND, J. V. D. 2017. Effects of intranasal insulin application on the hypothalamic BOLD response to glucose ingestion. *Scientific Reports*, 7, 1-7.
- ØRSKOV, C., HARTMANN, B., POULSEN, S. S., THULESEN, J., HARE, K. J. & HOLST, J. J. 2005. GLP-2 stimulates colonic growth via KGF, released by subepithelial myofibroblasts with GLP-2 receptors. *Regulatory Peptides*, 124, 105-112.
- ORSKOV, C., HOLST, J. J., KNUHTSEN, S., BALDISSERA, F. G., POULSEN, S. S. & NIELSEN, O. V. 1986. Glucagon-like peptides GLP-1 and GLP-2, predicted products of the glucagon gene, are secreted separately from pig small intestine but not pancreas. *Endocrinology*, 119, 1467-75.
- PADOA-SCHIOPPA, C. & ASSAD, J. A. 2008. The representation of economic value in the orbitofrontal cortex is invariant for changes of menu. *Nat Neurosci*, 11, 95-102.
- PAGE, K. A., ARORA, J., QIU, M., RELWANI, R., CONSTABLE, R. T. & SHERWIN, R. S. 2009. Small decrements in systemic glucose provoke increases in hypothalamic blood flow prior to the release of counterregulatory hormones. *Diabetes*, 58, 448-52.
- PAGE, K. A., CHAN, O., ARORA, J., BELFORT-DEAGUIAR, R., DZUIRA, J., ROEHMHOLDT, B., CLINE, G. W., NAIK, S., SINHA, R., CONSTABLE, R. T. & SHERWIN, R. S. 2013. Effects of fructose vs glucose on regional cerebral blood flow in brain regions involved with appetite and reward pathways. *JAMA*, 309, 63-70.
- PAGE, K. A., SEO, D., BELFORT-DEAGUIAR, R., LACADIE, C., DZUIRA, J., NAIK, S., AMARNATH, S., CONSTABLE, R. T., SHERWIN, R. S. & SINHA, R. 2011. Circulating glucose levels modulate neural control of desire for high-calorie foods in humans. *J Clin Invest*, 121, 4161-4169.

- PALMER, J., COOPER, C. & SHIPLEY, R. 1978. Rate of release of hepatic triacylglycerol into serum in the starved rat. *Biochemical Journal*, 172, 219-226.
- PANNACCIULLI, N., LE, D. S., SALBE, A. D., CHEN, K., REIMAN, E. M., TATARANNI, P. A. & KRAKOFF, J. 2007. Postprandial glucagon-like peptide-1 (GLP-1) response is positively associated with changes in neuronal activity of brain areas implicated in satiety and food intake regulation in humans. *Neuroimage*, 35, 511-7.
- PARKER, D. R., CARLISLE, K., COWAN, F. J., CORRALL, R. J. & READ, A.
 E. 1995. Postprandial mesenteric blood flow in humans: relationship to endogenous gastrointestinal hormone secretion and energy content of food. *European journal of gastroenterology & hepatology*, 7, 435-440.
- PARKER, H., TUCKER, E., HOAD, C., PAL, A., COSTIGAN, C., HUDDERS, N., PERKINS, A., BLACKSHAW, E., GOWLAND, P. & MARCIANI, L. 2016. Development and validation of a large, modular test meal with liquid and solid components for assessment of gastric motor and sensory function by non-invasive imaging. *Neurogastroenterology & Motility*, 28, 554-568.
- PARKINSON, J. R., CHAUDHRI, O. B., KUO, Y. T., FIELD, B. C., HERLIHY, A. H., DHILLO, W. S., GHATEI, M. A., BLOOM, S. R. & BELL, J. D. 2009. Differential patterns of neuronal activation in the brainstem and hypothalamus following peripheral injection of GLP-1, oxyntomodulin and lithium chloride in mice detected by manganese-enhanced magnetic resonance imaging (MEMRI). *Neuroimage*, 44, 1022-31.
- PARKMAN, H. P., FASS, R. & FOXX-ORENSTEIN, A. E. 2010. Treatment of patients with diabetic gastroparesis. *Gastroenterol Hepatol (N Y)*, 6, 1-16.
- PATRICIA, J. J. & DHAMOON, A. S. 2019. Physiology, Digestion.
- PAYETTE, H., GRAY-DONALD, K., CYR, R. & BOUTIER, V. 1995. Predictors of dietary intake in a functionally dependent elderly population in the community. *American Journal of Public Health*, 85, 677-683.
- PERKO, M. J. 2001. Duplex Ultrasound for Assessment of Superior Mesenteric Artery Blood Flow. *European Journal of Vascular and Endovascular Surgery*, 21, 106-117.
- PHAN, C. T. & TSO, P. 2001. Intestinal lipid absorption and transport. *Front Biosci*, 6, D299-D319.
- PIRONI, L., STANGHELLINI, V., MIGLIOLI, M., CORINALDESI, R., DE GIORGIO, R., RUGGERI, E., TOSETTI, C., POGGIOLI, G., MORSELLI LABATE, A. M., MONETTI, N. & ET AL. 1993. Fat-induced ileal brake in humans: a dose-dependent phenomenon correlated to the plasma levels of peptide YY. *Gastroenterology*, 105, 733-9.
- POCAI, A., MORGAN, K., BUETTNER, C., GUTIERREZ-JUAREZ, R., OBICI, S. & ROSSETTI, L. 2005. Central leptin acutely reverses diet-induced hepatic insulin resistance. *Diabetes*, 54, 3182-3189.
- POPPITT, S. D., MCCORMACK, D. & BUFFENSTEIN, R. 1998. Short-term effects of macronutrient preloads on appetite and energy intake in lean women. *Physiology & behavior*, 64, 279-285.

- POTIER, M., FROMENTIN, G., LESDEMA, A., BENAMOUZIG, R., TOMÉ, D. & MARSSET-BAGLIERI, A. 2010. The satiety effect of disguised liquid preloads administered acutely and differing only in their nutrient content tended to be weaker for lipids but did not differ between proteins and carbohydrates in human subjects. *Br J Nutr*, 104, 1406-14.
- POWLEY, T. & BERTHOUD, H.-R. 1985. Diet and cephalic phase insulin responses. *The American Journal of Clinical Nutrition*, 42, 991-1002.
- POWLEY, T. L. 1977. The ventromedial hypothalamic syndrome, satiety, and a cephalic phase hypothesis. *Psychological review*, 84, 89.
- PUNJABI, M., ARNOLD, M., GEARY, N., LANGHANS, W. & PACHECO-LÓPEZ, G. 2011. Peripheral glucagon-like peptide-1 (GLP-1) and satiation. *Physiol Behav*, 105, 71-6.
- QAMAR, M. & READ, A. 1988. Effects of ingestion of carbohydrate, fat, protein, and water on the mesenteric blood flow in man. *Scandinavian journal of gastroenterology*, 23, 26-30.
- RAMAGE, S., FARMER, A., ECCLES, K. A. & MCCARGAR, L. 2014. Healthy strategies for successful weight loss and weight maintenance: a systematic review. *Appl Physiol Nutr Metab*, 39, 1-20.
- RAPER, N., PERLOFF, B., INGWERSEN, L., STEINFELDT, L. & ANAND, J. 2004. An overview of USDA's Dietary Intake Data System. *Journal of Food Composition and Analysis*, 17, 545-555.
- READ, N. 1992. Role of gastrointestinal factors in hunger and satiety in man. *Proceedings of the Nutrition Society*, 51, 7-11.
- RICHARDSON, C. T. & FELDMAN, M. 1986. Salivary response to food in humans and its effect on gastric acid secretion. *American Journal of Physiology-Gastrointestinal and Liver Physiology*, 250, G85-G91.
- RIDGWAY, J. 2010. Cardiovascular magnetic resonance physics for clinicians: Part I. Journal of cardiovascular magnetic resonance : official journal of the Society for Cardiovascular Magnetic Resonance, 12, 71.
- RIGAUD, D., TROSTLER, N., ROZEN, R., VALLOT, T. & APFELBAUM, M. 1995. Gastric distension, hunger and energy intake after balloon implantation in severe obesity. *International Journal of Obesity and Related Metabolic Disorders: Journal of the International Association for the Study of Obesity*, 19, 489-495.
- RITENOUR, E. R. 1990. Doppler Ultrasound: Physics, Instrumentation and Clinical Applications. *Radiology*, 177, 346-346.
- RITTER, R. C. 2010. Gastrointestinal Peptides and the Control of Food Intake. *In:* KOOB, G. F., MOAL, M. L. & THOMPSON, R. F. (eds.) *Encyclopedia of Behavioral Neuroscience*. Oxford: Academic Press.
- RITTER, R. C. & LADENHEIM, E. E. 1985. Capsaicin pretreatment attenuates suppression of food intake by cholecystokinin. *American Journal of Physiology-Regulatory, Integrative and Comparative Physiology,* 248, R501-R504.
- RITTER, R. C., SLUSSER, P. G. & STONE, S. 1981. Glucoreceptors controlling feeding and blood glucose: location in the hindbrain. *Science*, 213, 451-2.

- ROE, M., PINCHEN, H., CHURCH, S. & FINGLAS, P. 2015. McCance and Widdowson's The Composition of Foods Seventh Summary Edition and updated Composition of Foods Integrated Dataset. *Nutrition Bulletin*, 40, 36-39.
- ROLLS, E. T. & MCCABE, C. 2007. Enhanced affective brain representations of chocolate in cravers vs. non-cravers. *Eur J Neurosci*, 26, 1067-76.
- RORDEN, C., KARNATH, H.-O. & BONILHA, L. 2007. Improving Lesion-Symptom Mapping. *Journal of cognitive neuroscience*, 19, 1081-1088.
- ROTHEMUND, Y., PREUSCHHOF, C., BOHNER, G., BAUKNECHT, H. C., KLINGEBIEL, R., FLOR, H. & KLAPP, B. F. 2007. Differential activation of the dorsal striatum by high-calorie visual food stimuli in obese individuals. *Neuroimage*, 37, 410-21.
- ROWLAND, M. K., ADAMSON, A. J., POLIAKOV, I., BRADLEY, J., SIMPSON, E., OLIVIER, P. & FOSTER, E. 2018. Field Testing of the Use of Intake24-An Online 24-Hour Dietary Recall System. *Nutrients*, 10.
- SALDIVAR, S. O. S. & PEREZ-CARRILLO, E. 2016. Maize. In: CABALLERO, B., FINGLAS, P. M. & TOLDRÁ, F. (eds.) Encyclopedia of Food and Health. Oxford: Academic Press.
- SAM, A. H., TROKE, R. C., TAN, T. M. & BEWICK, G. A. 2012. The role of the gut/brain axis in modulating food intake. *Neuropharmacology*, 63, 46-56.
- SANTANGELO, A., PERACCHI, M., CONTE, D., FRAQUELLI, M. & PORRINI, M. 1998. Physical state of meal affects gastric emptying, cholecystokinin release and satiety. *British Journal of Nutrition*, 80, 521-527.
- SANTOS, M., REZENDE, A., DOS SANTOS FILHO, P., GONÇALVES, J., BERALDO, F. & SAMPAIO, A. 2017. Intrapancreatic accessory spleen. *Einstein (São Paulo)*, 15.
- SAVAGE, A. P., ADRIAN, T. E., CAROLAN, G., CHATTERJEE, V. K. & BLOOM, S. R. 1987. Effects of peptide YY (PYY) on mouth to caecum intestinal transit time and on the rate of gastric emptying in healthy volunteers. *Gut*, 28, 166-70.
- SCHAEFFLER, A., GROSS, P., BUETTNER, R., BOLLHEIMER, C., BUECHLER, C., NEUMEIER, M., KOPP, A., SCHOELMERICH, J. & FALK, W. 2009. Fatty acid-induced induction of Toll-like receptor-4/nuclear factor-κB pathway in adipocytes links nutritional signalling with innate immunity. *Immunology*, 126, 233-245.
- SCHILLER, C., FRÖHLICH, C. P., GIESSMANN, T., SIEGMUND, W., MÖNNIKES, H., HOSTEN, N. & WEITSCHIES, W. 2005. Intestinal fluid volumes and transit of dosage forms as assessed by magnetic resonance imaging. *Alimentary pharmacology & therapeutics*, 22, 971-979.
- SCHILLING, T. M., FERREIRA DE SA, D. S., WESTERHAUSEN, R.,
 STRELZYK, F., LARRA, M. F., HALLSCHMID, M., SAVASKAN, E.,
 OITZL, M. S., BUSCH, H. P., NAUMANN, E. & SCHACHINGER, H.
 2014. Intranasal insulin increases regional cerebral blood flow in the

insular cortex in men independently of cortisol manipulation. *Hum Brain Mapp*, 35, 1944-56.

- SCHIRRA, J., NICOLAUS, M., ROGGEL, R., KATSCHINSKI, M., STORR, M., WOERLE, H. J. & GÖKE, B. 2006. Endogenous glucagon-like peptide 1 controls endocrine pancreatic secretion and antro-pyloro-duodenal motility in humans. *Gut*, 55, 243-251.
- SCHIRRA, J., NICOLAUS, M., WOERLE, H., STRUCKMEIER, C., KATSCHINSKI, M. & GÖKE, B. 2009. GLP-1 regulates gastroduodenal motility involving cholinergic pathways. *Neurogastroenterology & Motility*, 21, 609-e22.
- SCHJOLDAGER, B., MORTENSEN, P. E., MYHRE, J., CHRISTIANSEN, J. & HOLST, J. J. 1989. Oxyntomodulin from distal gut. Role in regulation of gastric and pancreatic functions. *Dig Dis Sci*, 34, 1411-9.
- SCHMIDT, P. T., NÄSLUND, E., GRYBÄCK, P., JACOBSSON, H., HARTMANN, B., HOLST, J. J. & HELLSTRÖM, P. M. 2003. Peripheral administration of GLP-2 to humans has no effect on gastric emptying or satiety. *Regul Pept*, 116, 21-5.
- SCHVARCZ, E., PALMÉR, M., ÅMAN, J. & BERNE, C. 1995. Hypoglycemm Increases the Gastric Emptying Rate in Healthy Subjects. *Diabetes Care*, 18, 674-676.
- SCHWARTZ, M. W., WOODS, S. C., PORTE, D., SEELEY, R. J. & BASKIN, D. G. 2000. Central nervous system control of food intake. *Nature*, 404, 661-671.
- SCHWARTZ, T. W., HOLST, J. J., FAHRENKRUG, J., JENSEN, S. L.,
 NIELSEN, O. V., REHFELD, J. F., DE MUCKADELL, O. B. & STADIL,
 F. 1978. Vagal, cholinergic regulation of pancreatic polypeptide secretion. *J Clin Invest*, 61, 781-9.
- SCHWIZER, W., FOX, M. & STEINGÖTTER, A. 2003. Non-invasive investigation of gastrointestinal functions with magnetic resonance imaging: towards an "ideal" investigation of gastrointestinal function. *Gut*, 52, iv34.
- SCOTT, T. R. & PLATA-SALAMÁN, C. R. 1999. Taste in the monkey cortex. *Physiol Behav*, 67, 489-511.
- SEIMON, R. V., BRENNAN, I. M., RUSSO, A., LITTLE, T. J., JONES, K. L., STANDFIELD, S., WISHART, J. M., HOROWITZ, M. & FEINLE-BISSET, C. 2013. Gastric emptying, mouth-to-cecum transit, and glycemic, insulin, incretin, and energy intake responses to a mixed-nutrient liquid in lean, overweight, and obese males. *Am J Physiol Endocrinol Metab*, 304, E294-300.
- SERHAN, C. N., CHIANG, N. & VAN DYKE, T. E. 2008. Resolving inflammation: dual anti-inflammatory and pro-resolution lipid mediators. *Nature Reviews Immunology*, 8, 349-361.
- SHA, E. 2000. Review article: control of gallbladder motor fuction. *Aliment Pharmacol Ther*, 14, 2-8.

- SHIANG, K. 2004. The SAS® calculations of areas under the curve (AUC) for multiple metabolic readings. Western users of SAS Software presentation, Pasadena 15.10. 2004.
- SIDERY, M., MACDONALD, I. & BLACKSHAW, P. 1994. Superior mesenteric artery blood flow and gastric emptying in humans and the differential effects of high fat and high carbohydrate meals. *Gut*, 35, 186-190.
- SIDERY, M. B. & MACDONALD, I. A. 1994. The effect of meal size on the cardiovascular responses to food ingestion. *British Journal of Nutrition*, 71, 835-848.
- SIEBER, C., BEGLINGER, C., JAEGER, K., HILDEBRAND, P. & STALDER, G. 1991. Regulation of postprandial mesenteric blood flow in humans: evidence for a cholinergic nervous reflex. *Gut*, 32, 361-366.
- SIEBER, C., BEGLINGER, C., JAGER, K. & STALDER, G. 1989. Regulatory phases of food-induced superior mesenteric artery blood flow (SMABF) in man. *Gastroenterology*, 96, A471.
- SKOV, A. R., TOUBRO, S., RØNN, B., HOLM, L. & ASTRUP, A. 1999. Randomized trial on protein vs carbohydrate in ad libitum fat reduced diet for the treatment of obesity. *Int J Obes Relat Metab Disord*, 23, 528-36.
- SLAVIN, J. & GREEN, H. 2007. Dietary fibre and satiety. *Nutrition Bulletin*, 32, 32-42.
- SMEETS, P. A. M., DE GRAAF, C., STAFLEU, A., VAN OSCH, M. J. P. & VAN DER GROND, J. 2005. Functional MRI of human hypothalamic responses following glucose ingestion. *NeuroImage (Orlando, Fla.)*, 24, 363-368.
- SOMEYA, N. 2007. Endo MY, Fukuba Y, Hayashi N. Limited effect of breathing frequency on blood velocity measurements in renal and superior mesenteric arteries. Physiol Meas, 28, 1369-1374.
- SOTHORNVIT, R. 2009. Effect of hydroxypropyl methylcellulose and lipid on mechanical properties and water vapor permeability of coated paper. *Food Research International*, 42, 307-311.
- SOUCY, J. & LEBLANC, J. 1998. Protein meals and postprandial thermogenesis. *Physiology & behavior*, 65, 705-709.
- SOYBEL, D. I. 2005. Anatomy and physiology of the stomach. *Surgical Clinics*, 85, 875-894.
- SPETTER, M. S., DE GRAAF, C., MARS, M., VIERGEVER, M. A. & SMEETS, P. A. M. 2014. The sum of its parts--effects of gastric distention, nutrient content and sensory stimulation on brain activation. *PloS One*, 9, e90872e90872.
- STICE, E., BURGER, K. S. & YOKUM, S. 2013. Relative ability of fat and sugar tastes to activate reward, gustatory, and somatosensory regions. *Am J Clin Nutr*, 98, 1377-84.

STOECKEL, L. E., WELLER, R. E., COOK, E. W., 3RD, TWIEG, D. B., KNOWLTON, R. C. & COX, J. E. 2008. Widespread reward-system activation in obese women in response to pictures of high-calorie foods. *Neuroimage*, 41, 636-47.

- STRUBBE, J. H. & WOODS, S. C. 2004. The timing of meals. *Psychological review*, 111, 128.
- STUBBS, R. J., HUGHES, D. A., JOHNSTONE, A. M., ROWLEY, E., REID, C., ELIA, M., STRATTON, R., DELARGY, H., KING, N. & BLUNDELL, J. E. 2000. The use of visual analogue scales to assess motivation to eat in human subjects: a review of their reliability and validity with an evaluation of new hand-held computerized systems for temporal tracking of appetite ratings. *British Journal of Nutrition*, 84, 405-415.
- STUBBS, R. J., VAN WYK, M. C., JOHNSTONE, A. M. & HARBRON, C. G. 1996. Breakfasts high in protein, fat or carbohydrate: effect on within-day appetite and energy balance. *Eur J Clin Nutr*, 50, 409-17.
- STUNKARD, A. J. & MESSICK, S. 1985. The three-factor eating questionnaire to measure dietary restraint, disinhibition and hunger. J Psychosom Res, 29, 71-83.
- SUGANUMA, Y., SHIMIZU, T., SATO, T., MORII, T., FUJITA, H., HARADA SASSA, M. & YAMADA, Y. 2020. Magnitude of slowing gastric emptying by glucagon-like peptide-1 receptor agonists determines the amelioration of postprandial glucose excursion in Japanese patients with type 2 diabetes. *J Diabetes Investig*, 11, 389-399.
- SUKKRIANG, N., CHANPRASERTPINYO, W., WATTANAPISIT, A., PUNSAWAD, C., THAMRONGRAT, N. & SANGPOOM, S. 2021. Correlation of body visceral fat rating with serum lipid profile and fasting blood sugar in obese adults using a noninvasive machine. *Heliyon*, 7, e06264.
- SUMITHRAN, P., PRENDERGAST, L. A., DELBRIDGE, E., PURCELL, K., SHULKES, A., KRIKETOS, A. & PROIETTO, J. 2011. Long-Term Persistence of Hormonal Adaptations to Weight Loss. *New England Journal of Medicine*, 365, 1597-1604.
- SUN, X., VELDHUIZEN, M. G., WRAY, A. E., DE ARAUJO, I. E., SHERWIN, R. S., SINHA, R. & SMALL, D. M. 2014. The neural signature of satiation is associated with ghrelin response and triglyceride metabolism. *Physiol Behav*, 136, 63-73.
- SVOBODA, M., TASTENOY, M., VERTONGEN, P. & ROBBERECHT, P. 1994. Relative quantitative analysis of glucagon receptor mRNA in rat tissues. *Mol Cell Endocrinol*, 105, 131-7.
- SZALAY, C., ARADI, M., SCHWARCZ, A., ORSI, G., PERLAKI, G., NEMETH, L., HANNA, S., TAKACS, G., SZABO, I., BAJNOK, L., VERECZKEI, A., DOCZI, T., JANSZKY, J., KOMOLY, S., ORS HORVATH, P., LENARD, L. & KARADI, Z. 2012. Gustatory perception alterations in obesity: an fMRI study. *Brain Res*, 1473, 131-40.
- TALAIRACH, J. & TOURNOUX, P. 1988. Co-Planar Stereotaxic Atlas of the Human Brain: 3-Dimensional Proportional System: An Approach to Cerebral Imaging. Georg Thieme Verlag.
- TATARANNI, P. A., JEAN-FRANÇOIS, G., KEWEI, C., ANNE, U., DANIEL, B., ARLINE, D. S., RICHARD, E. P., MICHAEL, L., ERIC, M. R. & ERIC, R. 1999. Neuroanatomical Correlates of Hunger and Satiation in

Humans Using Positron Emission Tomography. *Proc Natl Acad Sci*, 96, 4569-4574.

- TEFF, K. 2000. Nutritional implications of the cephalic-phase reflexes: endocrine responses. *Appetite*, 34, 206-213.
- TENCH, C., TANASESCU, R., CONSTANTINESCU, C. S., AUER, D. P. & COTTAM, W. 2021. Easy to interpret Coordinate Based Meta-Analysis of neuroimaging studies: Analysis of Brain Coordinates (ABC). *BioRxiv*, 2020.12. 03.409953.
- TERAMOTO, H., SHIMIZU, T., YOGO, H., NISHIMIYA, Y., HORI, S., KOSUGI, T. & NAKAYAMA, S. 2012. Assessment of gastric emptying and duodenal motility upon ingestion of a liquid meal using rapid magnetic resonance imaging. *Exp Physiol*, 97, 516-24.
- TERLOUW, L. G., VAN DIJK, L. J. D., VAN NOORD, D., VOOGD, T., BAKKER, B. J., NIKKESSEN, S., BRUNO, M. J. & MOELKER, A. 2022. MRI-based pre- and postprandial flow in the mesenteric vasculature of patients with suspected chronic mesenteric ischemia. *Eur J Radiol*, 151, 110316.
- THANARAJAH, S. E., IGLESIAS, S., KUZMANOVIC, B., RIGOUX, L., STEPHAN, K. E., BRÜNING, J. C. & TITTGEMEYER, M. 2019. Modulation of midbrain neurocircuitry by intranasal insulin. *Neuroimage*, 194, 120-127.
- THANARAJAH, S. E. & TITTGEMEYER, M. 2020. Food reward and gut-brain signalling. *Neuroforum*, 26, 1-9.
- THORNLEY, S., RUSSELL, B. & KYDD, R. 2011. Carbohydrate reward and psychosis: an explanation for neuroleptic induced weight gain and path to improved mental health? *Curr Neuropharmacol*, 9, 370-5.
- TIEDEMANN, L. J., SCHMID, S. M., HETTEL, J., GIESEN, K., FRANCKE, P., BÜCHEL, C. & BRASSEN, S. 2017. Central insulin modulates food valuation via mesolimbic pathways. *Nature Communications*, 8, 16052.
- TOLESSA, T., GUTNIAK, M., HOLST, J. J., EFENDIC, S. & HELLSTRÖM, P. M. 1998. Inhibitory effect of glucagon-like peptide-1 on small bowel motility. Fasting but not fed motility inhibited via nitric oxide independently of insulin and somatostatin. *The Journal of clinical investigation*, 102, 764-774.
- TOLESSA, T., NÄSLUND, E. & HELLSTRÖM, P. M. 2001. The inhibitory mechanism of GLP-1, but not glucagon, on fasted gut motility is dependent on the L-arginine/nitric oxide pathway. *Regulatory peptides*, 98, 33-40.
- TORELLI, A., TOBIA, V., ERZEGOVESI, S., GAMBARINI, A. & OGLIARI, A. L. 2022. Validation of the Italian version of the Power of Food Scale in the adult population. *Eating and Weight Disorders-Studies on Anorexia*, *Bulimia and Obesity*, 1-7.
- TOTMAN, J. J., MARCIANI, L., FOLEY, S., CAMPBELL, E., HOAD, C. L., MACDONALD, I. A., SPILLER, R. C. & GOWLAND, P. A. 2009. Characterization of the time course of the superior mesenteric, abdominal aorta, internal carotid and vertebral arteries blood flow response to the oral

glucose challenge test using magnetic resonance imaging. *Physiological measurement*, 30, 1117-1136.

- TREMBLAY, A. & BELLISLE, F. 2015. Nutrients, satiety, and control of energy intake. *Applied Physiology, Nutrition, and Metabolism,* 40, 971-979.
- TRUMBO, P., SCHLICKER, S., YATES, A. A. & POOS, M. 2002. Dietary reference intakes for energy, carbohydrate, fiber, fat, fatty acids, cholesterol, protein and amino acids. *J Am Diet Assoc*, 102, 1621-30.
- TURKELTAUB, P. E., EDEN, G. F., JONES, K. M. & ZEFFIRO, T. A. 2002. Meta-analysis of the functional neuroanatomy of single-word reading: method and validation. *Neuroimage*, 16, 765-80.
- TURKELTAUB, P. E., EICKHOFF, S. B., LAIRD, A. R., FOX, M., WIENER, M. & FOX, P. 2012. Minimizing within-experiment and within-group effects in Activation Likelihood Estimation meta-analyses. *Hum Brain Mapp*, 33, 1-13.
- VAINIK, U., GARCÍA-GARCÍA, I. & DAGHER, A. 2019. Uncontrolled eating: a unifying heritable trait linked with obesity, overeating, personality and the brain. *Eur J Neurosci*, 50, 2430-2445.
- VAN BLOEMENDAAL, L., IJZERMAN, R. G., TEN KULVE, J. S., BARKHOF, F., KONRAD, R. J., DRENT, M. L., VELTMAN, D. J. & DIAMANT, M. 2014. GLP-1 receptor activation modulates appetite-and reward-related brain areas in humans. *Diabetes*, 63, 4186-4196.
- VAN CAN, J., SLOTH, B., JENSEN, C., FLINT, A., BLAAK, E. & SARIS, W. 2014. Effects of the once-daily GLP-1 analog liraglutide on gastric emptying, glycemic parameters, appetite and energy metabolism in obese, non-diabetic adults. *International journal of obesity*, 38, 784-793.
- VAN DEN ABEELE, J., RUBBENS, J., BROUWERS, J. & AUGUSTIJNS, P. 2017. The dynamic gastric environment and its impact on drug and formulation behaviour. *European Journal of Pharmaceutical Sciences*, 96, 207-231.
- VAN KLEEF, E., VAN TRIJP, J., VAN DEN BORNE, J. & ZONDERVAN, C. 2012. Successful development of satiety enhancing food products: towards a multidisciplinary agenda of research challenges. *Critical reviews in food science and nutrition*, 52, 611-628.
- VAN STRIEN, T., FRIJTERS, J. E., BERGERS, G. P. & DEFARES, P. B. 1986. The Dutch Eating Behavior Questionnaire (DEBQ) for assessment of restrained, emotional, and external eating behavior. *International journal of eating disorders*, 5, 295-315.
- VAN VLIET, S., KOH, H. E., PATTERSON, B. W., YOSHINO, M., LAFOREST, R., GROPLER, R. J., KLEIN, S. & MITTENDORFER, B. 2020. Obesity Is Associated With Increased Basal and Postprandial β-Cell Insulin Secretion Even in the Absence of Insulin Resistance. *Diabetes*, 69, 2112-2119.
- VANIS, L., GENTILCORE, D., HAUSKEN, T., PILICHIEWICZ, A. N., LANGE, K., RAYNER, C. K., FEINLE-BISSET, C., MEYER, J. H., HOROWITZ, M. & JONES, K. L. 2010. Effects of gastric distension on blood pressure and superior mesenteric artery blood flow responses to

intraduodenal glucose in healthy older subjects. *Am J Physiol Regul Integr Comp Physiol*, 299, R960-7.

- VANIS, L., GENTILCORE, D., LANGE, K., GILJA, O. H., RIGDA, R. S., TRAHAIR, L. G., FEINLE-BISSET, C., RAYNER, C. K., HOROWITZ, M. & JONES, K. L. 2012. Effects of variations in intragastric volume on blood pressure and splanchnic blood flow during intraduodenal glucose infusion in healthy older subjects. *Am J Physiol Regul Integr Comp Physiol*, 302, R391-9.
- VELDHORST, M., SMEETS, A., SOENEN, S., HOCHSTENBACH-WAELEN, A., HURSEL, R., DIEPVENS, K., LEJEUNE, M., LUSCOMBE-MARSH, N. & WESTERTERP-PLANTENGA, M. 2008. Protein-induced satiety: effects and mechanisms of different proteins. *Physiology & behavior*, 94, 300-307.
- VELDHORST, M. A., NIEUWENHUIZEN, A. G., HOCHSTENBACH-WAELEN, A., WESTERTERP, K. R., ENGELEN, M. P., BRUMMER, R.-J. M., DEUTZ, N. E. & WESTERTERP-PLANTENGA, M. S. 2009. Effects of complete whey-protein breakfasts versus whey without GMPbreakfasts on energy intake and satiety. *Appetite*, 52, 388-395.
- VERDICH, C., FLINT, A., GUTZWILLER, J. P., NÄSLUND, E., BEGLINGER, C., HELLSTRÖM, P. M., LONG, S. J., MORGAN, L. M., HOLST, J. J. & ASTRUP, A. 2001a. A meta-analysis of the effect of glucagon-like peptide-1 (7-36) amide on ad libitum energy intake in humans. J Clin Endocrinol Metab, 86, 4382-9.
- VERDICH, C., TOUBRO, S., BUEMANN, B., LYSGÅRD MADSEN, J., JUUL HOLST, J. & ASTRUP, A. 2001b. The role of postprandial releases of insulin and incretin hormones in meal-induced satiety—effect of obesity and weight reduction. *International Journal of Obesity*, 25, 1206-1214.
- VIECILI, P. R. N., DA SILVA, B., HIRSCH, G. E., PORTO, F. G., PARISI, M. M., CASTANHO, A. R., WENDER, M. & KLAFKE, J. Z. 2017. Chapter One - Triglycerides Revisited to the Serial. *In:* MAKOWSKI, G. S. (ed.) *Advances in Clinical Chemistry*. Elsevier.
- VOLK, N. & LACY, B. 2017. Anatomy and physiology of the small bowel. *Gastrointestinal Endoscopy Clinics*, 27, 1-13.
- WANG, K. S., SMITH, D. V. & DELGADO, M. R. 2016. Using fMRI to study reward processing in humans: past, present, and future. *J Neurophysiol*, 115, 1664-78.
- WANG, Y. T., MOHAMMED, S. D., FARMER, A. D., WANG, D., ZARATE, N., HOBSON, A. R., HELLSTRÖM, P. M., SEMLER, J. R., KUO, B., RAO, S. S., HASLER, W. L., CAMILLERI, M. & SCOTT, S. M. 2015. Regional gastrointestinal transit and pH studied in 215 healthy volunteers using the wireless motility capsule: influence of age, gender, study country and testing protocol. *Aliment Pharmacol Ther*, 42, 761-72.
- WEHLING, H. & LUSHER, J. 2019. People with a body mass index ≥30 underreport their dietary intake: A systematic review. *Journal of Health Psychology*, 24, 2042-2059.

- WEIGLE, D. S., BREEN, P. A., MATTHYS, C. C., CALLAHAN, H. S., MEEUWS, K. E., BURDEN, V. R. & PURNELL, J. Q. 2005. A highprotein diet induces sustained reductions in appetite, ad libitum caloric intake, and body weight despite compensatory changes in diurnal plasma leptin and ghrelin concentrations. *Am J Clin Nutr*, 82, 41-8.
- WESTERTERP-PLANTENGA, M., LUSCOMBE-MARSH, N., LEJEUNE, M., DIEPVENS, K., NIEUWENHUIZEN, A., ENGELEN, M., DEUTZ, N., AZZOUT-MARNICHE, D., TOME, D. & WESTERTERP, K. 2006. Dietary protein, metabolism, and body-weight regulation: dose–response effects. *International Journal of Obesity*, 30, S16-S23.
- WETTERGREN, A., WØJDEMANN, M., MEISNER, S., STADIL, F. & HOLST, J. 1997. The inhibitory effect of glucagon-like peptide-1 (GLP-1) 7-36 amide on gastric acid secretion in humans depends on an intact vagal innervation. *Gut*, 40, 597-601.
- WHO. 2021. *Obesity and overweight* [Online]. WHO. Available: <u>https://www.who.int/news-room/fact-sheets/detail/obesity-and-</u> <u>overweight#:~:text=For%20adults%2C%20WHO%20defines%20overwei</u> <u>ght,than%20or%20equal%20to%2030</u>. [Accessed 21 June 2023].
- WIKAREK, T., KOCEŁAK, P., OWCZAREK, A. J., CHUDEK, J. & OLSZANECKA-GLINIANOWICZ, M. 2020. Effect of Dietary Macronutrients on Postprandial Glucagon and Insulin Release in Obese and Normal-Weight Women. *International Journal of Endocrinology*, 2020, 4603682.
- WILCOX, G. 2005. Insulin and insulin resistance. Clin Biochem Rev, 26, 19-39.
- WILLMS, B., WERNER, J., HOLST, J. J., ORSKOV, C., CREUTZFELDT, W. & NAUCK, M. A. 1996. Gastric emptying, glucose responses, and insulin secretion after a liquid test meal: effects of exogenous glucagon-like peptide-1 (GLP-1)-(7-36) amide in type 2 (noninsulin-dependent) diabetic patients. J Clin Endocrinol Metab, 81, 327-32.
- WILSON, M.-M. G., THOMAS, D. R., RUBENSTEIN, L. Z., CHIBNALL, J. T., ANDERSON, S., BAXI, A., DIEBOLD, M. R. & MORLEY, J. E. 2005. Appetite assessment: simple appetite questionnaire predicts weight loss in community-dwelling adults and nursing home residents. *The American journal of clinical nutrition*, 82, 1074-1081.
- WITJAKSONO, F., LUKITO, W., WIJAYA, A., ANNISA, N. G., JUTAMULIA, J., NURWIDYA, F. & SIMADIBRATA, M. 2018. The effect of breakfast with different macronutrient composition on PYY, ghrelin, and ad libitum intake 4 h after breakfast in Indonesian obese women. *BMC Res Notes*, 11, 787.
- WOLNERHANSSEN, B. K., MEYER-GERSPACH, A. C., SCHMIDT, A.,
 ZIMAK, N., PETERLI, R., BEGLINGER, C. & BORGWARDT, S. 2015.
 Dissociable Behavioral, Physiological and Neural Effects of Acute
 Glucose and Fructose Ingestion: A Pilot Study. *PLoS One*, 10, e0130280.
- WREN, A. M., SEAL, L. J., COHEN, M. A., BRYNES, A. E., FROST, G. S., MURPHY, K. G., DHILLO, W. S., GHATEI, M. A. & BLOOM, S. R.

2001. Ghrelin enhances appetite and increases food intake in humans. *J Clin Endocrinol Metab*, 86, 5992.

- WYATT, P., BERRY, S. E., FINLAYSON, G., O'DRISCOLL, R.,
 HADJIGEORGIOU, G., DREW, D. A., KHATIB, H. A., NGUYEN, L.
 H., LINENBERG, I., CHAN, A. T., SPECTOR, T. D., FRANKS, P. W.,
 WOLF, J., BLUNDELL, J. & VALDES, A. M. 2021. Postprandial
 glycaemic dips predict appetite and energy intake in healthy individuals.
 Nat Metab, 3, 523-529.
- WYNNE, K., PARK, A. J., SMALL, C. J., MEERAN, K., GHATEI, M. A., FROST, G. S. & BLOOM, S. R. 2006. Oxyntomodulin increases energy expenditure in addition to decreasing energy intake in overweight and obese humans: a randomised controlled trial. *Int J Obes (Lond)*, 30, 1729-36.
- YANG, N., LIU, X., DING, E. L., XU, M., WU, S., LIU, L., SUN, X. & HU, F. B. 2009. Impaired ghrelin response after high-fat meals is associated with decreased satiety in obese and lean Chinese young adults. *J Nutr*, 139, 1286-91.
- YEH, S.-T. 2002. Using trapezoidal rule for the area under a curve calculation. *Proceedings of the 27th Annual SAS*® *User Group International (SUGI'02)*, 1-5.
- YEOMANS, M., RE, R., WICKHAM, M., LUNDHOLM, H. & CHAMBERS, L. 2016. Beyond expectations: the physiological basis of sensory enhancement of satiety. *International Journal of Obesity*, 40, 1693-1698.
- YI, J., WARUNEK, D. & CRAFT, D. 2015. Degradation and stabilization of peptide hormones in human blood specimens. *PLoS One*, 10, e0134427.
- YUSTA, B., HUANG, L., MUNROE, D., WOLFF, G., FANTASKE, R., SHARMA, S., DEMCHYSHYN, L., ASA, S. L. & DRUCKER, D. J. 2000. Enteroendocrine localization of GLP-2 receptor expression in humans and rodents. *Gastroenterology*, 119, 744-755.
- ZANCHI, D., DEPOORTER, A., EGLOFF, L., HALLER, S., MAHLMANN, L., LANG, U. E., DREWE, J., BEGLINGER, C., SCHMIDT, A. & BORGWARDT, S. 2017. The impact of gut hormones on the neural circuit of appetite and satiety: A systematic review. *Neurosci Biobehav Rev*, 80, 457-475.
- ZAVORAL, J. H., HANNAN, P., FIELDS, D. J., HANSON, M. N., FRANTZ, I. D., KUBA, K., ELMER, P. & JACOBS JR, D. R. 1983. The hypolipidemic effect of locust bean gum food products in familial hypercholesterolemic adults and children. *The American journal of clinical nutrition*, 38, 285-294.
- ZHANG, B., TIAN, D., YU, C., LI, M., ZANG, Y., LIU, Y. & WALTER, M. 2015. Altered baseline brain activity differentiates regional mechanisms subserving biological and psychological alterations in obese men. *Scientific Reports*, 5, 11563.
- ZYOUD, S. E. H., SHAKHSHIR, M., ABUSHANAB, A. S., KONI, A., SHAHWAN, M., JAIROUN, A. A. & AL-JABI, S. W. 2022. Global

research trends on the links between insulin resistance and obesity: a visualization analysis. *Translational Medicine Communications*, 7, 18.

10 Appendices

10.1 Visual analogue scale (VAS)

Participant ID number:_____ Date:_____ Time-point:_____

Please rate the degree of desire to eat/hunger/fullness you would typically feel? Please put a slash (/) mark somewhere on the lines below

1.	How strong is your desire to eat?	
	Very weak	Very strong
2.	How thirsty do you feel?	
	Not at all	Extremely
3.	How hungry do you feel?	
	Not at all	Extremely
4.	How full do you feel?	
	Not at all	Very full
5.	How much food do you think you could eat?	
	Nothing	A large amount
	at all	

10.2 Preparation methods of drinks and meals used in the thesis

10.2.1 <u>Materials and equipment used in preparation of drinks and meals in</u> <u>the thesis</u>

- Mineral water (Evian, French Alps, France)
- Tween 20 (SIGMA, Aldrich, Gillingham, U.K)

- O Hydroxypropyl methylcellulose (HPMC, Benecel[™], K15M, Ashland, IMCD UK Ltd)
- o Coffee flavour (Nielsen Massey Vanillas, Inc, USA)
- Artificial sweeteners (Hermesetas, Switzerland)
- Rapeseed oil (Sainsbury's Supermarket, UK)
- Dolmio Bolognese Original Pasta sauce (Freepost Mars Food UK, Dublin, UK)
- Dried white Pasta (Sainsbury's Supermarket, UK)
- Olive oil (Sainsbury's Supermarket, UK)
- Cheddar cheese (Sainsbury's Supermarket, UK)
- Whey protein (Protein Works, Chocolate Silk, www.theproteinworks.com)
- Maltodextrin DE 18 (Maltosweet 180, Azelis UK)
- Cocoa powder (Tesco Supermarket, UK)
- Kitchen scales
- Measuring cups
- Heidolph stirrer (model RZR 2021, shown in Figure 10.1)
- Silverson (model L5R, shown in Figure 10.2)
- o Metallic spoon
- o Food thermometer
- Food blender
- Large saucepan
- Kitchen scales
- \circ Colander
- Mixing bowl
- Measuring bow
- Glassware (beakers, cylinders, and bottles)

10.2.2 Area and equipment cleaning

• Hands were thoroughly washed with soap and water (warm or cold) and dried before handling food.

- Utensils were washed before and after preparing food.
- Worktops were cleaned and sanitised before and after preparing food.

10.2.3 Preparation method of the standard pasta meal in Chapter 5

Ninety-five grams of the pasta was cooked in boiling water for eleven minutes. The pasta was drained well and rinsed in cold water until it was no longer warm. One hundred and ninety-five grams of Dolmio Bolognese Original Pasta sauce, 40 g of cheddar cheese, and 4.2 g of olive oil were added to the pasta and stirred until it became well combined. The pasta was aliquoted into ~500 g portions and stored in the refrigerator overnight. On the study day, the pasta was heated in a microwave for 4 minutes, stirred hallway through, and then provided to participants straight away.

10.2.4 Preparation of high-fat emulsion (High-fat drink) in Chapter 6

To prepare 500 ml of the fat emulsion, 148 g of water was added in a beaker with a 5 g Tween 20 emulsifier. The water and emulsifier were mixed in the Silverson (Figure 10.1) at full speed for two to three minutes. Then, 110 g of rapeseed oil was added to the mixture and mixing continued for 10 minutes. A Silverson homogenizer was used to achieve a highly uniform particle size distribution. This process involves three stages of mixing and homogenising, employing a high-speed rotor, centrifugal force, and hydraulic shear. A fat emulsion droplet size of approximately 6 m was produced. The fat emulsion was transferred to a 500-ml bottle, cooled to a temperature below 5°C, and then kept in the fridge until used.



Figure 10.1. Silverson homogenizer (model L5R). The picture taken from/source: the manufacture's website (<u>https://www.silverson.co.uk)</u>.

To prepare the HPMC mixture, a Heidolph magnetic stirrer (Figure 10.2) was used to prepare the HPMC solution. Water was divided into two equal quantities (150 g) and placed in two beakers. One beaker was covered with cling film and kept in the fridge, while the other beaker was placed on the heater, which was adjusted to a temperature of 100–150 °C. The paddle on the stirrer was set to a speed of 2, with the power level at 1, and the beaker was also covered with cling film to reserve heat. HPMC (4.5 g) was gradually introduced into the stirring water when the water reached a temperature of 70 °C. HPMC was added while stirring when the water reached 70 °C. Subsequently, the cold water was added while continuing to stir. The resulting mixture was covered with cling film and transferred to a 500-ml bottle, cooled to a temperature below 5°C, and kept in a fridge until used. Before being served to participants on the study days, drinks were removed from the fridge and allowed to warm up to room temperature for thirty to forty minutes. The drink was served in an opaque paper cup with a straw and a lid.



Figure 10.2. Heidolph magnetic stirrer (model RZR 2021).

To prepare the drink for a study visit day, 210 g of the fat emulsion was mixed with 90 g of the HPMC mixture. Next, 5 artificial sweeteners and 18 ml of the coffer flavour were added to the drink. Following the mixing of these components, the drink—roughly 280 ml—was put into a cylinder. Water was subsequently added to make 300 millilitres.

10.2.5 Preparation of high-carbohydrate drink in Chapter 6

To prepare 500 ml of the high-carbohydrate drink, two hundred and sixty grams of maltodextrin were mixed with 240 g of water by a metallic spoon until the mixture became smooth and no lumps were visible.

To prepare the drink for a study visit day, 300 g of the mixture was weighted, and then 5 artificial sweeteners and 18 ml of the coffer flavour were added to the mixture. Following the mixing of these components, the drink was put into a cylinder. Water was subsequently added to make 300 millilitres. To match the components of the HF and HC drinks, 3.3 g of Tween 20 was added to 300 ml of carbohydrate drink. Before being served to participants on the study days, drinks were removed from the fridge and allowed to warm up to room temperature for thirty to forty minutes. The drink was served in an opaque paper cup with a straw and a lid.
10.2.6 Preparation method of high-protein drink in Chapter 7

To prepare 500 ml of the high-protein drink, 116.6 g of whey protein was mixed with 383 g of water in the food blender at full speed for two minutes. The drink was transferred to a 500-ml bottle and kept in the fridge overnight at a temperature below 5 °C. To prepare the drink for a study visit day, 300 g of the mixture was weighted, and then the drink was put into a cylinder. Water was subsequently added to make 300 milliliters. Before being served to participants on the study days, drinks were removed from the fridge and allowed to warm up to room temperature for thirty to forty minutes. The drink was served in an opaque paper cup with a straw and a lid.

10.2.7 Preparation method of high-carbohydrate drink in Chapter 7

The HPMC mixture for this drink was prepared by mixing 7.2 g of HPMC with 300 g of water. The preparation method was explained previously in section 10.2.4.

To prepare 300 ml of the drink, using a food blender, mix 150 g of the maltodextrin mixture with 150 g of the HPMC mixture, 12 g of cocoa powder, and 3 artificial sweeteners. Following the mixing of these components, the drink was put into a cylinder. To prepare the drink for a study visit day, 300 g of the mixture was weighted, and then the drink was put into a cylinder. Water was subsequently added to make 300 milliliters. The drink was transferred to a 500-ml bottle and kept in the fridge overnight at a temperature below 5 °C. Before being served to participants on the study days, drinks were removed from the fridge and allowed to warm up to room temperature for thirty to forty minutes. The drink was served in an opaque paper cup with a straw and a lid.

10.2.8 Preparation of ad libitum pasta meal in Chapter 7

Two hundred and fifty grams of the pasta was cooked in boiling water for eleven minutes. The pasta was drained well and rinsed in a cold water until it was no longer warm. Three-hundred and forty grams of Dolmio Bolognese Original Pasta sauce, 80 grams of cheddar cheese, and 30 g of olive oil were added to the pasta and stirred until it became well combined. The pasta was aliquot into ~1000 g portions and stored in the fridge overnight. On the study day, the pasta was heated in a microwave for 2 minutes, stirred hallway through, and then provided to participants straight

away with 250 ml water. The pasta plate was always replenished when it was about three-quarters empty, ensuring that there was always enough hot food for participants and that they weren't prompted to stop eating by emptying their plate.